# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# FORM 8-K

**CURRENT REPORT** 

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of report (date of earliest event reported): October 9, 2020

# **BULLFROG GOLD CORP.**

(Exact name of registrant as specified in its charter)

<u>Delaware</u>

(State or other jurisdiction of incorporation)

000-54653 (Commission File Number)

(IRS Employer Identification No.)

41-2252162

897 Quail Run Drive, Grand Junction, Colorado (Address of principal executive offices) <u>81505</u> (Zip Code)

Registrant's telephone number, including area code: (970) 628-1670

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

[] Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

[] Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

[] Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

[] Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act: None

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§ 230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§ 240.12b-2 of this chapter).

Emerging growth company [ ]

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. []

#### Item 1.01. Entry into a Material Definitive Agreement.

#### Acquisition Transaction

On October 9, 2020, Bullfrog Gold Corp. (the "Company") entered into a membership interest purchase agreement (the "MIPA") with Homestake Mining Company of California ("Homestake"), and Lac Minerals (USA) LLC ("Lac Minerals" and together with Homestake, the "Barrick Parties"). Pursuant to the MIPA, the Company agreed to purchase from the Barrick Parties, and the Barrick Parties agreed to sell to the Company, all of the equity interests (the "Equity Interests") in Bullfrog Mines LLC ("BMLLC"), the successor by conversion of Barrick Bullfrog Inc. (the "Acquisition Transaction").

Under the Acquisition Transaction, in consideration for the purchase of the Equity Interests in BMLLC, the Company will issue or provide to the Barrick Parties and/or to Barrick Gold Corporation ("Barrick"), (i) 54,600,000 units of the Company, each unit consisting of one share of common stock and one four-year warrant (the "Warrant") to purchase one share of common stock at an exercise price of C\$0.30 ("Units"), (ii) a 2% net smelter returns royalty granted on all minerals produced from the Patented Claims and the Unpatented Claims (each as defined in the MIPA), pursuant to a royalty deed among BMLLC and the Barrick Parties to be entered into in connection with and upon closing of the Acquisition Transaction, and (iii) certain investor rights, including anti-dilution rights, pursuant to an investor rights agreement to be entered into in connection, among the Company, Augusta Investments Inc. ("Augusta"), and Barrick. Pursuant to the investor rights agreement, the Company will grant to Barrick the right to designate one member of the Board of Directors of the Company, subject to the terms and conditions set forth therein.

The closing of the Acquisition Transaction is subject to customary closing conditions and the closing of the Financing Transaction (defined below). Upon closing of the Acquisition Transaction, the Company is obligated to pay a fee of C\$1.2 million to Fort Capital Partners.

Each of the Company, on the one hand, and the Barrick Parties, on the other hand, made customary representations and warranties to the other in the MIPA for transactions of the type contemplated in the Acquisition Transaction, including, among other things, representations and warranties relating to their respective due organization, capitalization, title to assets, regulatory compliance, financial statements, litigation and absence of undisclosed liabilities.

#### Financing Transaction

Also on October 9, 2020, the Company entered into subscription agreements (the "Subscription Agreements") with Augusta and certain other individuals identified by Augusta (together, the "Augusta Group"), pursuant to which the Company agreed to sell to the Augusta Group, and the Augusta Group agreed to purchase from the Company, an aggregate of 110 million Units for an aggregate purchase price of C\$22 million (the "Financing Transaction," and together with the Acquisition Transaction, the "Transactions").

The closing of the Financing Transaction is subject to customary closing conditions and the closing of the Acquisition Transaction.

Upon and as a condition to the closing of the Transactions, the Board of Directors and management of the Company will be reconstituted as follows: (i) all directors of the Company (other than David Beling) will resign, and each of Daniel Earle, Maryse Bélanger and Donald R. Taylor will be appointed as a director, and (ii) all senior management of the Company will resign, and the following persons will be appointed officers of the Company: Maryse Bélanger will be appointed as President and Chief Executive Officer, Michael McClelland will be appointed as Chief Financial Officer, and Johnny Pappas will be appointed as Vice President, Environmental and Planning. Ms. Bélanger, Mr. Earle and Mr. Taylor are subscribers under the Financing Transaction in the amounts of C\$250,000, C\$500,000 and C\$200,000, respectively. Barrick's nominee, if identified by Barrick in advance of closing of the Transactions, will also be appointed as a director of the Company upon closing of the Transactions.

In connection with the foregoing, the Company relied upon the exemption from registration provided by Section 4(a)(2) under the Securities Act of 1933, as amended, for transactions not involving a public offering.

The foregoing summary of the MIPA, Warrant, and Subscription Agreement is qualified in its entirety by the full text of such documents which are filed as exhibits to this report, and incorporated herein by reference.

#### Item 3.02 Unregistered Sales of Equity Securities.

The information under Item 1.01 is incorporated by reference into this Item 3.02.

# Item 8.01 Other Information.

On October 13, 2020, the Company issued a press release announcing the execution of the MIPA and Subscription Agreements. A copy of this press release is filed as Exhibit 99.1 hereto, and incorporated herein by reference.

# Item 9.01 Financial Statements and Exhibits.

(c) Exhibits

Exhibit Number	Description
2.1	Membership Interest Purchase Agreement, dated as of October 9, 2020, by and among Bullfrog Gold Corp., Homestake Mining Company of California and Lac Minerals (USA) LLC
10.1	Form of Subscription Agreement, dated as of October 9, 2020
10.2	Form of Warrant
99.1	Press Release, issued by Bullfrog Gold Corp. on October 13, 2020
	3

# SIGNATURE

Pursuant to the requirement of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

# BULLFROG GOLD CORP.

Date: October 15, 2020

By: /s/ David Beling

David Beling Chief Executive Officer

## Exhibit 2.1

#### **Execution Version**

## MEMBERSHIP INTEREST PURCHASE AGREEMENT

THIS MEMBERSHIP INTEREST PURCHASE AGREEMENT (the "Agreement") is made and entered into as of this 9<sup>th</sup> day of October 2020 (the "<u>Execution Date</u>"), by and among Homestake Mining Company of California, a California corporation ("<u>Homestake</u>") and Lac Minerals (USA) LLC, a Delaware limited liability company ("<u>Lac Minerals</u>" and together with Homestake, the "<u>Sellers</u>"), and Bullfrog Gold Corp., a Delaware corporation (the "<u>Buyer</u>"). Homestake shall act as the "<u>Administrative Agent</u>" for Sellers under this Agreement. Homestake, Lac Minerals and Buyer sometimes may be referred to in this Agreement individually as a "<u>Party</u>," and collectively as the "<u>Parties</u>."

#### RECITALS

A. The Parties entered into the binding term sheet dated September 7, 2020, a first amendment thereto on October 7, 2020 and a second amendment thereto on October 8, 2020 (as amended, the "<u>Term Sheet</u>") with Augusta Investments Inc. ("<u>Augusta</u>") that sets forth the principal understanding with respect to an integrated transaction involving (i) the purchase by Buyer from Sellers of all of the equity interests in Bullfrog Mines LLC (the "<u>Company</u>"), a Delaware limited liability company and the successor by conversion of BBI (as defined below), and (ii) the concurrent financing pursuant to which Augusta and/or its Affiliates (as defined below) and certain other individuals identified by Augusta (together, the "<u>Augusta Group</u>"), as subscribers, will acquire 110,000,000 units of Buyer on the terms described in the Term Sheet (the "<u>Augusta Group Units</u>") and set forth in the subscription agreements dated as of the date hereof between the Augusta Group members and Buyer (the "<u>Subscription Agreements</u>").

B. Upon completion of the Contribution and Conversion (as defined below), (i) Homestake will own all of the Voting Membership Units (the "<u>Voting Units</u>") in the Company, and (ii) Lac Minerals will own all of the Non-Voting Membership Units (the "<u>Non-Voting Units</u>") in the Company. The Voting Units and the Non-Voting Units (collectively, the "<u>Membership Units</u>") constitute all of the issued and outstanding equity interests in the Company.

C. Buyer wishes to purchase the Membership Units from Sellers and each of Sellers wishes to sell its Membership Units to Buyer, all upon the terms and subject to the conditions set forth in this Agreement.

D. The concurrent closing of the transactions contemplated by the Subscription Agreements is a condition of Closing (as defined below).

#### AGREEMENT

NOW, THEREFORE, for and in consideration of the mutual covenants and agreements hereinafter set forth and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

#### ARTICLE 1 DEFINITIONS

1.1 <u>Certain Defined Terms</u>. As used in this Agreement and unless otherwise defined, the following capitalized terms shall have the following meanings:

"Administrative Agent" has the meaning set forth in the Preamble.

"<u>Affiliate</u>" of a Person means any other Person that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such Person. For purposes of this definition, the term "<u>control</u>" (including the terms "<u>controlled by</u>" and "<u>under common control with</u>") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise.

"Agreement" has the meaning set forth in the Preamble.

"Announcement" has the meaning set forth in Section 5.4(a).

"Anti-Money Laundering Laws" has the meaning set forth in Section 4.24(d).

"Augusta" has the meaning set forth in Recital A.

"Augusta Group" has the meaning set forth in Recital A.

"Augusta Group Units" has the meaning set forth in Recital A.

"Barrick Nevada Holding" has the meaning set forth in Section 2.5(b).

"BBI" means Barrick Bullfrog Inc., a Delaware corporation and the predecessor to the Company.

"Board Designee" has the meaning set forth in the Investor Rights Agreement.

"Books and Records" means books and records, including books of account, ledgers and general, financial and accounting records, machinery and equipment maintenance files, supplier lists, production data, quality control records and procedures, assay reports, drill logs, exploration reports, environmental studies, mine plans, operational documents, nonproprietary mining and reserve models, sales records, strategic plans, material and research, including all technical records, files papers, surveys and plans or specifications, in each case related to BBI or the Company and the Company Assets, but excluding any Privileged Documents or any documents related to the Contribution and Conversion.

"Bullfrog Permits" has the meaning set forth in Section 4.27(h).

"Bullfrog Project" has the meaning set forth in Section 4.25.

"Business Day" means from the hours of 9:00 a.m. (Eastern Time) through 5:00 p.m. (Eastern Time) of any day except Saturday, Sunday or any other day on which commercial banks located in Vancouver, British Columbia, Toronto, Ontario or Denver, Colorado are closed for business.

"Buyer" has the meaning set forth in the Preamble.

"Buyer Indemnified Parties" means, at the time of the applicable Claim, Buyer and its Affiliates, and its and their respective Representatives.

"<u>Canadian Securities Authorities</u>" means any of the securities commissions or similar securities regulatory authorities in each of the provinces and territories of Canada in which Buyer is a reporting issuer (or has analogous status).

"<u>Canadian Securities Laws</u>" means all applicable Canadian securities laws, the respective regulations, rules and orders made thereunder, and all applicable policies and notices issued by the Canadian Securities Authorities in the applicable jurisdictions in Canada.

"CDN\$" means "Canadian Dollars", the official currency of Canada.

"<u>Claim</u>" means any action, arbitration, cause of action, claim, counterclaim, demand, dispute, grievance, mediation, injunction, investigation, obligation, stay, suit or other proceeding, including a claim for indemnification made pursuant to Article 7.

"Closing" has the meaning set forth in Section 2.3.

"Closing Date" has the meaning set forth in Section 2.3.

"Closing Time" has the meaning set forth in Section 2.3.

"Common Share" means a share of common stock of Buyer, par value US\$0.0001 per share.

"<u>Company</u>" has the meaning set forth in <u>Recital A</u>, which is the successor by conversion to BBI pursuant to the Contribution and Conversion.

"<u>Company Assets</u>" means the Patented Claims, the Unpatented Claims, the Rights-of-Way, the Water Rights and the Operating Permits.

"Company Historical Financial Statements" has the meaning set forth in Section 5.13(a).

"Consideration" has the meaning set forth in Section 2.2.

"Consideration Shares" has the meaning set forth in Section 2.2(a).

"Consideration Units" has the meaning set forth in Section 2.2(a).

"Consideration Warrants" has the meaning set forth in Section 2.2(a).

"<u>Contribution and Conversion</u>" means the transactions pursuant to which (a) the Voting Stock and Non-Voting Stock held by Affiliates of Homestake (other than Lac Minerals) are contributed to Homestake, (b) BBI is converted into the Company, (c) the Voting Stock and Non-Voting Stock held by Homestake is converted into Voting Units, and the Non-Voting Stock held by Lac Minerals is converted into Non-Voting Units, in each case in accordance with the LLC Agreement, and (d) all assets (including any assets or ownership interests in Barrick Nevada Holding and NGM) other than the Company Assets are transferred or distributed by the Company to Homestake or Affiliates of Homestake.

"<u>CSE</u>" means the Canadian Securities Exchange or any successor stock exchange on which the Common Shares are listed.

"Deductible" has the meaning set forth in Section 7.5(a)(ii).

"De Minimis Amount" has the meaning set forth in Section 7.5(a)(i).

"Direct Claim" has the meaning set forth in Section 7.6(a).

"Disclosure Schedule" means the Disclosure Schedule delivered by the Administrative Agent to Buyer concurrently with the execution and delivery of this Agreement, as the same may be supplemented from time to time pursuant to this Agreement.

"Drop Dead Date" has the meaning set forth in Section 8.1(b)(i).

"D&O Insurance Policy" has the meaning set forth in Section 5.14.

"EDGAR" means the Electronic Data Gathering, Analysis and Retrieval system of the SEC.

"<u>Encumbrance</u>" means any lien, pledge, mortgage, deed of trust, security interest, charge, claim, easement, right-of-way, restriction, surface use agreement, imperfection of title, encroachment or other similar encumbrance.

"Environmental Laws" has the meaning set forth in Section 4.23(a).

"Environmental Permits" has the meaning set forth in Section 4.23(a).

"Exchange Act" means the United States Securities Exchange Act of 1934.

"Execution Date" has the meaning set forth in the Preamble.

"Financial Statement Exemption" has the meaning set forth in Section 5.13(a).

"Financial Statement Exemption Review Period" has the meaning set forth in Section 5.13(a).

"Financial Statements" has the meaning set forth in Section 4.20(a).

"Financing Transaction" has the meaning set forth in Section 4.13.

"GAAP" has the meaning set forth in Section 4.20(a).

"<u>Governance Changes</u>" means the changes to the officers and directors of Buyer contemplated in the Subscription Agreements or, subject to <u>Section 6.1(d)</u>, as otherwise agreed in writing among Buyer, Sellers and Augusta, each acting reasonably, prior to Closing.

"<u>Governmental Authority</u>" means: (a) any domestic or foreign government, whether national, federal, provincial, state, territorial, municipal or local (whether administrative, legislative, executive or otherwise); (b) any agency, authority, ministry, department, regulatory body, court, central bank, bureau, board or other instrumentality having legislative, judicial, taxing, regulatory, prosecutorial or administrative powers or functions of, or pertaining to, government; (c) any court, commission, individual arbitrator, arbitration panel or other body having adjudicative, regulatory, judicial, quasi-judicial, administrative or similar functions; or (d) any other body or entity created under the authority of or otherwise subject to the jurisdiction of any of the foregoing, including any stock or other securities exchange or professional association.

"Governmental Order" means any order, writ, judgment, injunction, decree, stipulation, determination or award entered by or with any Governmental Authority.

"Homestake" has the meaning set forth in the Preamble.

"Indemnified Party" means, as applicable, any one or more of the Seller Indemnified Parties or the Buyer Indemnified Parties.

"Indemnifying Party" means, as applicable, the Party or Parties against which Claims are asserted by the Indemnified Party.

"Indemnity Deed" means the form of Indemnity Deed attached to this Agreement as Exhibit A, including the form of Trust Deed attached thereto.

"Investment Company Act" means the United States Investment Company Act of 1940.

"Investor Rights Agreement" means the form of Investor Rights Agreement attached to this Agreement as Exhibit B.

"Knowledge of Buyer" means the actual knowledge of each of Dave Beling and Ty Minnick after due inquiry.

"Knowledge of Sellers" means the actual knowledge of each of Michael McCarthy and Patrick Malone after due inquiry.

"Lac Minerals" has the meaning set forth in the Preamble.

"Law" means any statute, law, ordinance, regulation, rule, code, order, constitution, treaty, common law, judgment, decree, formal interpretation, or other requirement or rule of law of any Governmental Authority.

"Lease and Option" has the meaning set forth in Section 2.7.

"Liability Limits" has the meaning set forth in Section 7.5.

"LLC Agreement" means the LLC agreement of the Company executed by Sellers upon formation thereof.

"Loss" means, in respect of any matter, all claims, demands, proceedings, losses, damages, liabilities, deficiencies, fines, costs and expenses (including reasonable legal and other professional fees and disbursements, interest, penalties and amounts paid in settlement but excluding punitive, exemplary, aggravated damages, lost opportunity damages and loss of profits), injuries and judgments arising directly or indirectly as a consequence of such matter.

"<u>Material Adverse Effect</u>" means any event, occurrence, fact, condition or change that is, or would reasonably be expected to be, materially adverse to the Patented Claims, the Unpatented Claims or Operating Permits taken as a whole; *provided, however,* that none of the following shall be taken into account in making a determination as to whether a Material Adverse Effect has occurred: (a) general economic or political conditions; (b) conditions generally affecting the mining industry; (c) any changes in financial, banking or securities markets in general, including any disruption thereof and any decline in the price of any security or any market index or any change in prevailing interest rates or capital costs or commodity markets; (d) acts of war (whether or not declared), armed hostilities or terrorism, or the escalation or worsening thereof; (e) any action required or permitted by the Transaction Documents or any action taken (or omitted to be taken) with the written consent of or at the written request of Buyer; (f) any change in the price of gold or other metals or any change in currency exchange rates; (g) any changes in applicable Laws or accounting rules; (h) any changes resulting from the announcement, pendency or completion of the Transactions; or (i) any natural or man-made disaster or acts of God (including Covid-19), except, in the case of clauses (d), (g) or (i), where such event, occurrence, fact, condition or change has a materially disproportionate effect on the Patented Claims, the Unpatented Claims or Operating Permits taken as a whole, relative to other comparable operations.

"Material Contracts" has the meaning set forth in Section 4.19.

"Membership Units" has the meaning set forth in Recital B.

"NGM" has the meaning set forth in Section 2.5(b).

"<u>NI 43-101</u>" has the meaning set forth in <u>Section 4.26</u>.

"Non-Voting Stock" means non-voting common stock of BBI.

"Non-Voting Units" has the meaning set forth in Recital B.

"<u>Notice of Claim</u>" has the meaning set forth in <u>Section 7.6(a)</u>.

"OFAC" has the meaning set forth in Section 4.24(e).

"Operating Permits" means the Permits set forth in Schedule 1.1(e) of the Disclosure Schedule.

"Parties" has the meaning set forth in the Preamble.

"Party" has the meaning set forth in the Preamble.

"Patented Claims" means patented mining claims set forth in Schedule 1.1(a) of the Disclosure Schedule.

"PCAOB" means the U.S. Public Company Accounting Oversight Board.

"<u>Permit</u>" means any permit, license, approval, consent, ruling, authorization, certification, concession, exemption, variance, notification, waiver, clearance or registration by or with any Governmental Authority or other third parties.

"<u>Permitted Encumbrances</u>" means: (a) Encumbrances for Taxes, assessments or other governmental charges not yet delinquent or the amount or validity of which is being contested in good faith by appropriate proceedings; (b) Encumbrances of mechanics, carriers, workers, repairers, warehousemen and similar Persons arising or incurred in the ordinary course of business; (c) matters of public record; (d) any conditions that reasonably would be expected to be shown by a current survey; (e) Environmental Laws; (f) Encumbrances that arise under contract, or due to zoning, subdivision, entitlement, and other land use Laws or regulations; (g) royalty interests encumbering the Patented Claims or the Unpatented Claims, whether or not of public record; (h) the paramount title of the United States; (i) pledges made with respect to any Operating Permits; (j) Encumbrances that arise solely by reason of acts of or with the approval of Buyer or its Affiliates; (k) Encumbrances not created by Sellers, any of their Affiliates or the Company that affect the underlying interest of any Membership Units or the Company Assets; (l) Governmental Orders; and (m) any Encumbrances included in the Disclosure Schedule.

"<u>Person</u>" means any individual, corporation or company with or without share capital, partnership, joint venture, association, trust, unincorporated organization, trustee, executor, administrator or other legal personal representative, Governmental Authority or entity however designated or constituted.

"<u>Privilege</u>" has the meaning set forth in <u>Section 2.5(a)</u>.

"Privileged Documents" has the meaning set forth in Section 2.5(a).

"<u>Reclamation Bonds</u>" has the meaning set forth in <u>Section 5.11</u>.

"Replacement Bonds" has the meaning set forth in Section 5.11.

"<u>Representative</u>" means, with respect to any Person, any and all directors, officers, members, managers, employees, consultants, financial advisors, counsel, accountants and other agents of such Person.

"<u>Rights-of-Way</u>" means the rights-of-way set forth in <u>Schedule 1.1(c)</u> of the Disclosure Schedule.

"<u>RMMC</u>" means Rocky Mountain Minerals Corp. a Nevada corporation and a wholly-owned subsidiary of Buyer.

"Royalty Deed" means the form of Royalty Deed attached to this Agreement as Exhibit C.

"<u>Rule 14f-1</u>" has the meaning set forth in <u>Section 5.21</u>.

"Sarbanes-Oxley Act" has the meaning set forth in Section 4.20(d).

"SEC" means the U.S. Securities and Exchange Commission.

"SEC Filings" has the meaning set forth in Section 4.6.

"Securities" means, collectively, the Consideration Shares, the Consideration Warrants and the Warrant Shares or any of them, as the circumstances require.

"Securities Act" means the United States Securities Act of 1933.

"SEDAR" means the System for Electronic Document Analysis and Retrieval of the Canadian Securities Administrators.

"<u>Seller Indemnified Parties</u>" means, at the time of the applicable Claim, each Seller and its Affiliates, and its and their respective Representatives.

"<u>Seller Marks</u>" means any trademark, trade name, service mark, trade dress, logo, copyright, signage, domain name, corporate or company name, or other designation that uses or is confusingly similar to the term "Barrick", in each case wherever used, whether or not filed for registration, registered or unregistered, and whether or not stylized.

"Sellers" has the meaning set forth in the Preamble.

"<u>Subscription Agreements</u>" has the meaning set forth in <u>Recital A</u>, being, more particularly, the subscription agreements dated as of the Execution Date among (a) Buyer and Augusta, (b) Buyer, Augusta and Donald Taylor, (c) Buyer, Augusta and Daniel Earle, (d) Buyer, Augusta and Purni Parikh, and (e) Buyer, Augusta and Maryse Belanger.

"Subsidiary" and "Subsidiaries" have the meanings set forth in Section 4.2.

"Survival Period" means a period of 365 consecutive days after the Closing Date.

"Tax" or "Taxes" means all federal, state, local, foreign and other income, gross receipts, sales, use, severance, depletion, production, ad valorem, transfer, franchise, registration, profits, license, lease, service, service use, withholding, payroll, employment, unemployment, estimated, excise, severance, environmental, stamp, occupation, premium, property (real or personal), real property gains, windfall profits, customs, duties or other taxes, fees, assessments or charges of any kind whatsoever, together with any interest, additions or penalties with respect thereto and any interest in respect of such additions or penalties.

"<u>Tax Return</u>" means any return, declaration, report, claim for refund, information return or statement or other document relating to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

"Technical Report" has the meaning set forth in Section 4.26.

"Term Sheet" has the meaning set forth in Recital A.

"<u>Third Party Claim</u>" has the meaning set forth in <u>Section 7.6(a)</u>.

"Transactions" means the transactions contemplated by the Transaction Documents, but excluding the Contribution and Conversion.

"<u>Transaction Documents</u>" means this Agreement, the Royalty Deed, the Indemnity Deed, the Investor Rights Agreement, the Subscription Agreements and any other agreement, instrument or deed entered into among the Augusta Group, Buyer, Sellers and any of their respective Affiliates or any of them at or prior to Closing.

"<u>Unpatented Claims</u>" means unpatented lode and placer mining claims, and millsites, owned by the Company set forth in <u>Schedule 1.1(b)</u> of the Disclosure Schedule.

"<u>US\$</u>" means "United States Dollars", the lawful currency of the United States of America.

"Voting Stock" means voting common stock of BBI.

"Voting Units" has the meaning set forth in Recital B.

"<u>Water Rights</u>" means water rights appurtenant to the Patented Claims and the Unpatented Claims set forth in <u>Schedule</u> <u>1.1(d)</u> of the Disclosure Schedule.

"Warrant Shares" means the Common Shares issuable on the exercise of the Consideration Warrants.

- 1.2 <u>Rules of Construction</u>.
  - (a) In this Agreement:

(i) unless the context otherwise clearly requires: (A) references to the plural include the singular, and references to the singular include the plural; (B) references to one gender include the other gender; (C) the words "include," "includes," and "including" do not limit the preceding terms or words and shall be deemed to be followed by the words "without limitation"; (D) the terms "hereof," "herein," "hereunder," "hereto," and similar terms refer to this entire Agreement and not to any particular provision of this Agreement, unless the provision otherwise provides; (E) "or" is used in the inclusive sense of "and/or"; (F) if a word or phrase is defined, then its other grammatical or derivative forms have a corresponding meaning; (G) a reference to Law or a statute, code, act, legislation, or to a provision thereof includes a modification, amendment, or substitution thereof or any successor Law, the rules and regulations promulgated thereunder, and the formal interpretations issued in accordance therewith; and (H) unless otherwise specified, the terms "day" and "days" mean and refer to calendar day(s);

(ii) unless otherwise specified, any reference to any document, instrument or agreement (including a reference to this Agreement): (A) includes and incorporates all exhibits, schedules, and other attachments thereto; (B) includes and incorporates all documents, instruments, deeds, or agreements issued or executed in connection therewith or in replacement thereof; and (C) means such document, instrument, deed, or agreement, or replacement or predecessor thereto, as amended, modified, or supplemented from time to time in accordance with its terms and in effect at the relevant time (except to the extent prohibited by this Agreement or such other agreement or document);

(iii) unless otherwise specified, all references to articles, sections, schedules and exhibits are to the Articles, Sections, Schedules, and Exhibits of this Agreement; and

(iv) the headings of this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.

(b) The Parties acknowledge that they and their respective legal counsel have reviewed and participated in negotiating and settling the terms of this Agreement and agree that no inference shall be drawn in favor of or against any Party by virtue of the fact that they or their respective legal counsel were or were not principally responsible for drafting this Agreement.

#### ARTICLE 2 PURCHASE AND SALE

2.1 <u>Purchase and Sale of the Membership Units</u>. Subject to the terms and conditions set forth in this Agreement, at the Closing:

(a) Homestake shall sell and Buyer shall purchase the Voting Units; and

(b) Lac Minerals shall sell and Buyer shall purchase the Non-Voting Units.

The Administrative Agent shall deliver the Voting Units and the Non-Voting Units to Buyer.

2.2 <u>Consideration</u>. In consideration for the purchase and sale of the Membership Units from Sellers pursuant to <u>Section</u> 2.1, Buyer shall deliver to the Administrative Agent at Closing the following (the "<u>Consideration</u>"):

(a) 54,600,000 units (the "<u>Consideration Units</u>"), with each Consideration Unit consisting of one restricted Common Share (the "<u>Consideration Shares</u>") and one whole warrant (the "<u>Consideration Warrants</u>") to purchase one Warrant Share at an exercise price of CDN\$0.30 or the prevailing equivalent United States Dollar price calculated using the average daily exchange rate of the Bank of Canada on the Business Day prior to the date of exercise, exercisable for four (4) years from the Closing Date in accordance with the terms of the Consideration Warrants, the form of which is attached hereto as <u>Exhibit D</u>;

(b) a 2% net smelter returns royalty granted on all minerals produced from the Patented Claims and the Unpatented Claims pursuant to the Royalty Deed;

(c) the Indemnity Deed; and

(d) certain investor rights, including antidilution rights, with regard to Buyer and the Consideration Shares, as set forth in the Investor Rights Agreement.

The Administrative Agent, in its sole and absolute discretion, shall allocate the Consideration between Sellers, to the extent applicable.

2.3 <u>Closing</u>. Subject to the terms and conditions of this Agreement, the purchase and sale of the Membership Units shall take place at a closing (the "<u>Closing</u>") beginning at 8:00 a.m., (Eastern Time) (the "<u>Closing Time</u>"), on the third Business Day after the last of the conditions to Closing set forth in <u>Article 6</u> have been satisfied or waived (other than conditions which, by their nature, are to be satisfied on the Closing Date), at the offices of Sichenzia Ross Ference LLP in New York, or at such other time or on such other date or at such other place as the Administrative Agent and Buyer may mutually agree upon in writing (the day on which the Closing may take place being the "<u>Closing Date</u>"). Subject to <u>Section 2.6</u>, in lieu of a physical closing, the Parties agree that the Closing may take place on the Closing Date on the exchange of counsels' undertakings which will involve each Party's counsel delivering to his or her counterpart all required documentation and payments, to be held in escrow and not released until all such documentation has been executed and delivered and all conditions have been satisfied or waived and each Party's solicitor has authorized in writing that the escrow is to be terminated.

2.4 Transactions at the Closing. At the Closing:

(a) The Administrative Agent shall deliver to Buyer the following:

(i) certificates evidencing the Voting Units, free and clear of all Encumbrances arising by, through or under Homestake, duly endorsed by Homestake in blank or accompanied by instruments of transfer duly executed in blank by Homestake;

(ii) certificates evidencing the Non-Voting Units, free and clear of all Encumbrances arising by, through or under Lac Minerals, duly endorsed by Lac Minerals in blank or accompanied by instruments of transfer duly executed in blank by Lac Minerals;

(iii) the Royalty Deed, duly executed by Sellers;

(iv) the Indemnity Deed, duly executed by Sellers;

(v) the Investor Rights Agreement, duly executed by Sellers or an Affiliate thereof that acquired the Consideration Units on Closing;

Closing Time;

(vi) resignations in writing of all directors, officers and managers of the Company effective as of the

Section 5.17;

(vii) the Books and Records of the Company, other than those Books and Records contemplated in

(viii) certificate of good standing or equivalent for the Company and each Seller issued by the State of Delaware and State of California, as applicable, within three days of the Closing Date; and

(ix) all other agreements, documents, instruments, deeds, or certificates required to be delivered by Sellers at or prior to the Closing pursuant to <u>Section 6.2</u> or reasonably requested by Buyer.

(b) Buyer shall deliver to the Administrative Agent the following:

(i) certificates representing the Consideration Shares, in the names of Homestake and Lac Minerals, or as otherwise directed by the Administrative Agent, and for the number of Consideration Shares issued to each of Homestake and Lac Minerals, or as otherwise directed by the Administrative Agent, as set forth in a written notice by the Administrative Agent to Buyer to be delivered at least two Business Days prior to Closing;

(ii) warrants representing the Consideration Warrants, in the names of Homestake and Lac Minerals, or as otherwise directed by the Administrative Agent, and for the number of Consideration Warrants issued to each of Homestake and Lac Minerals, or as otherwise directed by the Administrative Agent, as set forth in a written notice by the Administrative Agent to Buyer to be delivered at least two Business Days prior to Closing;

(iii) the Royalty Deed, duly executed by the Company;

(iv) the Indemnity Deed, duly executed by Buyer and/or the Company, as applicable;

(v) the Investor Rights Agreement, duly executed by Buyer and Augusta;

(vi) certificate of good standing or equivalent for Buyer issued by the State of Delaware within three days of the Closing Date; and

(vii) all other agreements, documents, instruments, deeds, or certificates required to be delivered by Buyer at or prior to the Closing pursuant to Section 6.3 or reasonably requested by the Administrative Agent.

## 2.5 Exclusions and Limitations.

(a) All communications and other documents exchanged between the Company or its Affiliates and legal counsel (including, as applicable, internal legal counsel) providing legal advice to the Company and its Affiliates, including documents and communications related to the Transactions and files maintained by legal counsel as a result of providing legal advice to the Company and its Affiliates (the "<u>Privileged Documents</u>"), that are subject to the attorney-client privilege, any similar privilege, or that constitute attorney work product (as applicable, a "<u>Privilege</u>"), specifically are excluded from the Transactions and shall be the property of Sellers and their respective Affiliates, as applicable. Neither the Company or Sellers, nor the applicable Affiliates, intend to waive any applicable Privilege, and any disclosure of any Privileged Documents, whether in the Books and Records or otherwise, shall be deemed to be inadvertent. Accordingly, Buyer acknowledges and agrees that a disclosure of any Privileged Documents will not constitute a waiver of such Privilege, and Buyer promptly shall return to the Administrative Agent, or with the consent of the Administrative Agent destroy, such inadvertently disclosed Privileged Documents.

(b) Pursuant to the Contribution and Conversion, all ownership and interest of BBI in and to Barrick Nevada Holding LLC, a Delaware limited company ("<u>Barrick Nevada Holding</u>"), Nevada Gold Mines LLC, a Delaware limited liability company ("<u>NGM</u>"), and any other assets which do not constitute Company Assets, will be transferred to Homestake and are not included in the Transactions. Buyer acknowledges such transfer, and covenants and agrees that it will not, now or in the future, make any claim of any legal or beneficial rights in or to the equity interests or assets of Barrick Nevada Holding or NGM or any other assets of the Company which may inadvertently be transferred pursuant to the Contribution and Conversion which do not constitute Company Assets.

(c) The Patented Claims, Unpatented Claims, Water Rights, Rights-of-Way and Operating Permits are in the name of BBI. Pursuant to the Contribution and Conversion, the name of BBI will be changed to the Company. The Administrative Agent will record the change in name from BBI to the Company with the appropriate Governmental Authorities in respect of the Patented Claims and the Unpatented Claims as part of the Contribution and Conversion. Subject to Section 5.8, Buyer, at its cost, shall be responsible and liable to use best efforts to file or record with the applicable Governmental Authorities all other name changes and qualifications within seven days of the Closing Date, including with regard to maintaining the qualification of the Company to conduct business in the State of California.

(d) In the event Buyer desires to revise any Patented Claims or Unpatented Claims, or any Water Rights, or the manner in which such Patented Claims or Unpatented Claims, or Water Rights, are recorded: (i) such revisions shall only be made following Closing; and (ii) Buyer shall be solely responsible for such recordings, and shall be solely liable for and shall pay all related recording fees and other costs, fees and expenses.

(e) In the event Buyer desires to amend, modify or revise any Operating Permits, or obtain additional Permits: (i) such amendments, modifications or revisions, or any action to obtain additional Permits, shall only be made following Closing; and (ii) Buyer shall be solely responsible for such amendments, modifications or revisions, or any action to obtain additional Permits, and shall be solely liable for and shall pay all related fees and other costs, including the cost of posting any bonds or other financial assurances related to any such amendments, modifications or revisions, or any action to obtain additional Permits.

2.6 <u>Simultaneous Transactions</u>. The Transactions, including the Transactions contemplated by the Subscription Agreements, and all deliveries contemplated by <u>Section 2.4</u> shall be deemed to occur simultaneously on the Closing Date, and none shall be deemed completed until all are completed. The escrow relating to Closing contemplated in <u>Section 2.3</u> shall not be terminated until the Parties have confirmed that the escrow arrangements relating to the closing of the Transactions contemplated by the Subscription Agreements will be concurrently released and that such closing will be deemed to have occurred concurrently with Closing.

2.7 <u>Termination of Lease and Option</u>. The Parties hereby agree that on and as of the Execution Date, the Lease and Option to Purchase dated March 23, 2015, as amended (the "<u>Lease and Option</u>"), between BBI and RMMC automatically shall terminate without any further action on the part of any Party thereto, and all rights and obligations of the parties and their respective Affiliates under the Lease and Option shall cease; *provided* that any provisions of the Lease and Option which are designated in the Lease and Option as surviving termination thereof shall be unaffected by the termination contemplated in this <u>Section 2.7</u>.

2.8 <u>Transfer Taxes</u>. Buyer shall be solely liable for and shall pay all applicable sales, transfer, use, stamp, conveyance, value-added, real property transfer, recording, and other similar Taxes, if any, together with all recording or filing fees, notarial fees and other similar costs of Closing, that may be imposed upon, or payable, collectible or incurred in connection with the Transactions.

2.9 <u>Administrative Agent</u>. Sellers hereby appoint Homestake as the Administrative Agent of Sellers under this Agreement, the Royalty Deed and the Indemnity Deed, to the extent that Sellers are party thereto, and each Seller hereby authorizes Homestake to act on behalf of it as its Administrative Agent in accordance with the terms of this Agreement, the Royalty Deed and the Indemnity Deed to the extent that it is a party thereto. Homestake hereby agrees to act as the Administrative Agent of Sellers as set forth in this Agreement, the Royalty Deed and the Indemnity Deed to which Sellers are party. Buyer hereby acknowledges and agrees that Homestake is acting as the Administrative Agent of Sellers under this Agreement, the Royalty Deed and the Indemnity Deed to which Sellers are party.

## 2.10 Legends.

(a) Sellers understand and agree that the certificates for the Securities shall bear substantially the following legend until: (i) such Securities shall have been registered under the Securities Act and effectively disposed of in accordance with a registration statement that has been declared effective; or (ii) in the opinion of counsel acceptable to Buyer, such Securities may be sold without registration under the Securities Act, as well as any applicable "blue sky" or state securities Law:

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), OR ANY APPLICABLE STATE SECURITIES LAWS. SUCH SECURITIES HAVE BEEN ACQUIRED FOR INVESTMENT PURPOSES AND MAY NOT BE OFFERED FOR SALE, SOLD, DELIVERED AFTER SALE, TRANSFERRED, PLEDGED OR HYPOTHECATED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT FILED BY THE COMPANY WITH THE U.S. SECURITIES AND EXCHANGE COMMISSION COVERING SUCH SECURITIES UNDER THE SECURITIES ACT OR AN OPINION OF COUNSEL THAT SUCH REGISTRATION IS NOT REQUIRED.

THE PRESENCE OF THIS LEGEND MAY IMPAIR THE ABILITY OF THE HOLDER HEREOF TO EFFECT "GOOD DELIVERY" OF THE SECURITIES REPRESENTED HEREBY ON A CANADIAN STOCK EXCHANGE."

(b) Sellers acknowledge that if any of the Securities are issued to a Person resident in Canada the certificates representing the Securities will bear, as of the Closing Date, a legend substantially in the following form and with the necessary information inserted:

"UNLESS PERMITTED UNDER SECURITIES LEGISLATION, THE HOLDER OF THIS SECURITY MUST NOT TRADE THE SECURITY BEFORE THE DATE THAT IS FOUR MONTHS AND A DAY AFTER THE CLOSING DATE."

#### ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF SELLERS

Except as set forth in the Disclosure Schedule, Sellers hereby jointly and severally represent and warrant to Buyer as of the date hereof and as of the Closing Date as follows:

## 3.1 Organization.

(a) Each Seller: (i) is duly organized, validly existing and in good standing under the Laws of the jurisdiction in which it was formed; and (ii) has the requisite company power and authority to own the Membership Units that will be held by it at the Closing Time.

(b) As of the date hereof and prior to the Contribution and Conversion, BBI is a corporation duly formed and in good standing under the Laws of the State of Delaware and is duly qualified to do business under the Laws of the States of California and Nevada.

(c) As of the Closing Time, the Company will be a limited liability company duly formed and in good standing under the Laws of the State of Delaware and will be duly qualified to do business under the Laws of the States of Nevada and California, subject to the filing by Buyer of name changes in the States of Nevada and California as required under <u>Section 2.5(c)</u>.

#### 3.2 Due Authorization, Execution and Delivery; Enforceability.

(a) Each Seller has the requisite company power and authority to enter into the Transaction Documents to which it is a party, to carry out its obligations under the Transaction Documents to which it is a party, and to consummate the transaction Documents to which it is a party, the performance by each Seller of its obligations under the Transaction Documents to which it is a party, and the consummation by each Seller of the Transactions to which it is a party have been duly authorized by all requisite company action. Each of the Transaction Documents to which a Seller is a party have been duly executed and delivered by such Seller and constitutes the legal, valid and binding obligation of such Seller, enforceable against such Seller in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting creditors' rights generally and by general principles of equity (regardless of whether enforcement is sought in a proceeding at law or in equity).

(b) As of the date hereof, and prior to the Contribution and Conversion, BBI has all necessary corporate power and authority to own and operate the Company Assets.

(c) At the Closing Time, Company will have all necessary corporate power and authority to own and operate the Company Assets.

3.3 <u>No Conflicts; Consents.</u> Except as set forth in the Disclosure Schedule and as would not have a Material Adverse Effect, the execution, delivery and performance by each Seller of the Transaction Documents to which it is a party, and the consummation of the Transactions to which it is a party: (a) do not: (i) conflict with or result in a violation or breach of any provision of the constating documents of either Seller or BBI or any contract or agreement with respect to BBI or the Company Assets to which Sellers or BBI is party (as of the date hereof, and prior to the Contribution and Conversion), or (ii) violate in any material respect any Law applicable to such Seller or the Company Assets; and (b) will not conflict with or result in a violation or breach of any provision of the constating or organizational documents of the Company or any contract or agreement to which the Company is party at the Closing Time. Upon the Closing in accordance with this Agreement, Buyer will acquire the Membership Units, representing 100% of the equity interests in the Company, free and clear of any Encumbrances arising by, through or under Sellers.

3.4 Capitalization of the Company; Title to Membership Units.

(a) As of the date hereof and prior to the Contribution and Conversion, the authorized equity of BBI consists of the Voting Shares and the Non-Voting Shares.

(b) At the Closing Time:

(i) the authorized equity of the Company will consist of the Voting Units and the Non-Voting Units;

(ii) all of the Membership Units will have been issued in accordance with the LLC Agreement;

(iii) Homestake will own all of the Voting Units free and clear of all Encumbrances arising by, through or under Homestake; and

(iv) Lac Minerals will own all of the Non-Voting Units free and clear of all Encumbrances arising by, through or under Lac Minerals.

(c) At the Closing Time, there will be no outstanding or authorized options, warrants, convertible securities or other rights, agreements, arrangements or commitments of any character relating to the equity of the Company or obligating any Seller or the Company to issue or sell any Membership Units or other equity interest in the Company.

3.5 <u>Company Assets</u>. At the Closing Time, the Company will not have any liabilities other than liabilities that directly relate to, arise from or otherwise have been incurred in connection with the ownership or operation of the Company Assets and the only material assets of the Company of any kind will be the following Company Assets:

(a) <u>Schedule 1.1(a)</u> of the Disclosure Schedule sets forth a true and complete list of the Patented Claims. Except as set forth in the Disclosure Schedule, at the Closing Time the Company will hold title to the Patented Claims, free and clear of all Encumbrances arising by, through or under Sellers or their Affiliates, except for Permitted Encumbrances.

(b) <u>Schedule 1.1(b)</u> of the Disclosure Schedule sets forth a true and complete list of the Unpatented Claims. Except as set forth in the Disclosure Schedule, at the Closing Time the Company will hold title to the Unpatented Claims, free and clear of all Encumbrances arising by, through or under Sellers or their Affiliates, except for Permitted Encumbrances. At the Closing Time, all maintenance fees on the Unpatented Claims held by the Company will have been paid on or before the date such maintenance fees have become due and payable.

(c) <u>Schedule 1.1(c)</u> of the Disclosure Schedule sets forth a true and complete list of all Rights-of-Way. Except as set forth in the Disclosure Schedule, at the Closing Time the Company will hold a valid and enforceable interest in the Rights-of-Way, free and clear of all Encumbrances, except for Permitted Encumbrances.

(d) <u>Schedule 1.1(d)</u> of the Disclosure Schedule sets forth a true and complete list of the Water Rights. Except as set forth in the Disclosure Schedule, at the Closing Time the Company will hold a valid and enforceable interest in the Water Rights, free and clear of all Encumbrances arising by, through or under Sellers or their Affiliates, except for Permitted Encumbrances.

(e) <u>Schedule 1.1(e)</u> the Disclosure Schedule sets forth a true and complete list of the Operating Permits. Except as set forth in the Disclosure Schedule, the Operating Permits are in force and effect and (i) as of the date hereof, BBI has not and (ii) at the Closing Time, the Company will not have, received any written notice from any Governmental Authority of any current and unresolved violation under any Operating Permit.

3.6 Legal Proceedings. As of the date hereof, BBI is not a party to any Claim with respect to the Company or the Company Assets. To the Knowledge of Sellers, no such Claim has been made or is threatened in writing which if determined adversely would result in a Material Adverse Effect or that challenges or seeks to prevent, enjoin or otherwise delay the Transactions.

3.7 <u>Compliance with Laws; Permits</u>. As of the date hereof and prior to the Contribution and Conversion, BBI is in compliance in all material respects with: (a) all Laws applicable to it, except where the failure to be in such compliance would not have a Material Adverse Effect; and (b) the Operating Permits and Water Rights are valid and in full force and effect, in each case except where the failure to obtain or maintain such Operating Permits and Water Rights would not have a Material Adverse Effect. At the Closing Time, the Company will be in compliance in all material respects with: (i) all Laws applicable to it, except where the failure to be in such compliance would not have a Material Adverse Effect; and (ii) the Operating Permits and Water Rights will be valid and in full force and effect, in each case (A) except where the failure to obtain or maintain such Operating Permits and Water Rights would not maintain such Operating Permits and Water Rights would not have a Material Adverse Effect; and (ii) the Operating Permits and Water Rights will be valid and in full force and effect, in each case (A) except where the failure to obtain or maintain such Operating Permits and Water Rights would not have a Material Adverse Effect and (B) subject to the filing by Buyer of name changes with the applicable Governmental Authorities as required under <u>Section 2.5(c)</u>.

3.8 Taxes. Except for matters that would not have a Material Adverse Effect:

(a) BBI has duly and timely filed (or there have been filed on its behalf) with the appropriate Governmental Authority all Tax Returns required to be filed by it for all periods ending prior to the date hereof. All such Tax Returns are complete and correct in all material respects and no material fact has been omitted therefrom. No extension of time in which to file any such Tax Returns is in effect. All Taxes shown on all such Tax Returns or any assessments or reassessments in respect of any such Tax Returns, and all Taxes required to be paid by BBI, have been paid in full.

(b) No Governmental Authority has asserted, or is threatening to assert, against BBI (or at the Closing Time, its successor, the Company) any deficiency or claim for Taxes, and no Governmental Authority has challenged or disputed a filing position taken by or on behalf of BBI in any Tax Return.

(c) (i) There are currently no tax actions, suits, proceedings, audits, investigations or claims now existing, pending or threatened against BBI (or at the Closing Time, its successor, the Company); (ii) no tax reassessments in respect of BBI (or at the Closing Time, its successor, the Company) have been issued and are outstanding; (iii) there are no outstanding issues or matters in respect of Taxes which have been raised and communicated to BBI (or at the Closing Time, its successor, the Company) by any Governmental Authority; and (iv) BBI (or at the Closing Time, its successor, the Company) has not executed or filed with any Governmental Authority any agreement or waiver extending the period for assessment, reassessment, payment or collection of any Taxes, or otherwise waived any statute of limitations in respect of Taxes.

(d) No Encumbrance for Taxes has been filed or exists in respect of BBI (or at the Closing Time, its successor, the Company) other than for Taxes not yet due and payable.

(e) The representations and warranties set forth in this <u>Section 3.8</u> are the sole and exclusive representations and warranties made by Sellers with respect to Taxes.

3.9 <u>Financial Advisors</u>. There is no Person acting or purporting to act on behalf of Sellers or any of their Affiliates who is entitled to any brokerage, agency or other fiscal advisory or similar fee in connection with the Transactions contemplated in this Agreement.

3.10 <u>Resale Restrictions.</u> Sellers acknowledge the Securities have not been registered under the Securities Act or any state securities Laws and may not be transferred or sold except pursuant to an effective registration statement or an available exemption therefrom. Sellers acknowledge an investment in the Securities involves a high degree of risk including the risks set forth in the SEC Filings. Subject to <u>Section 2.2</u>, Sellers, or upon written direction of the Sellers, an Affiliate thereof, are acquiring the Securities a principal for its or their own account and have no direct or indirect arrangement or understandings with any other Persons to distribute or regarding the distribution of such Securities.

## 3.11 No Other Representations and Warranties.

(a) Except for the representations and warranties specifically contained in this <u>Article 3</u> (including the Disclosure Schedule), Buyer hereby acknowledges and agrees that the Transactions contemplated by this Agreement are being completed on an "as is, where is" basis. None of Sellers or any of their Affiliates or Representatives, or any other Person, has made or makes any other express or implied representation or warranty, either written or oral, on behalf of Sellers, BBI or the Company, including any representatives (including any projections, information, documents or material made available to Buyer and its Representatives (including any projections, information, documents or material made available to Buyer or its Representatives in the Books and Records, management presentations or other communications with management, or in any other form in expectation of the Transactions), the sufficiency, merchantability or fitness for any particular purpose of the Company Assets, including the Operating Permits and the Water Rights, compliance with applicable Laws, or as to the future revenue, profitability or success of the Company, or any representation or warranty arising from statute or otherwise at law or in equity, all of which are hereby expressly disclaimed.

(b) WITHOUT LIMITING THE FOREGOING, NO SELLER MAKES ANY REPRESENTATIONS OR WARRANTIES WHATSOEVER CONCERNING THE EXISTENCE OR STATUS OF ANY MINES OR WORKINGS WITHIN THE AREA COVERED BY THE PATENTED CLAIMS AND UNPATENTED CLAIMS, INCLUDING THE EXISTENCE AND STATUS OF ANY ABANDONED MINES OR WORKINGS, THE STATUS OF ANY ROYALTIES IDENTIFIED IN THIS AGREEMENT (INCLUDING THE DISCLOSURE SCHEDULE) OR THE EXISTENCE OR STATUS OF ANY MINERAL RESERVES, OR RESOURCES, THE NATURE, LOCATION, AMOUNT OR VALUE OF ANY MINERALIZATION, MINERAL RESERVES, OR RESOURCES, THE ABILITY TO EXTRACT, PROCESS, OR SELL MINERALS BY ANY MEANS, WHETHER ANY NECESSARY PERMITS CAN BE OBTAINED IN A TIMELY MANNER, WHETHER ANY MINING CAN BE DONE ECONOMICALLY OR AT ALL, OR THAT THERE WILL BE NO THIRD PARTY CHALLENGE TO THE ISSUANCE OF ANY REQUIRED PERMIT OR ENVIRONMENTAL IMPACT STATEMENT REQUIRED FOR OPERATIONS WITH RESPECT TO THE COMPANY ASSETS, OR THAT THERE ARE NO RIGHTS (INCLUDING ROYALTIES, ACCESS RIGHTS, INFORMATION RIGHTS, RECONVEYANCE RIGHTS, REVERSIONARY RIGHTS OR OTHER RIGHTS OF PREDECESSORS IN INTEREST).

## ARTICLE 4 REPRESENTATIONS AND WARRANTIES OF BUYER

Buyer hereby represents and warrants to Sellers as of the date hereof and as of the Closing Date as follows:

4.1 Organization and Qualification. Buyer is a corporation duly organized, validly existing and in good standing under the Law of the State of Delaware. Buyer is duly qualified to do business and is in good standing in the states required due to: (i) the ownership or lease of real or personal property for use in the operation of Buyer's business; or (ii) the nature of the business conducted by Buyer. Buyer has all requisite power, right and authority to own, operate and lease its properties and assets, to carry on its business as now conducted, to execute, deliver and perform its obligations under the Transaction Documents to which it is a party, and to carry out the Transactions contemplated hereby and thereby, including the power and authority to create, issue and deliver the Consideration Units to be issued pursuant to this Agreement, and the Warrant Shares to be issued upon due exercise of the Consideration Warrants in accordance with their terms. All actions on the part of Buyer and its officers and directors necessary for the authorization, execution, delivery and performance of the Transaction Documents, the consummation of the Transactions contemplated hereby and thereby, and the performance of all of Buyer's obligations under the Transaction Documents, including the creation, issuance and delivery of the Consideration Units to be issued pursuant to this Agreement, and the Warrant Shares to be issued upon due exercise of the Consideration Warrants in accordance with their terms, have been taken or will be taken prior to the Closing. This Agreement has been, and the other Transaction Documents to which Buyer is a party on the Closing will be, duly executed and delivered by Buyer, and this Agreement is, and each of the other Transaction Documents to which it is a party on the Closing will be, a legal, valid and binding obligation of Buyer, enforceable against Buyer in accordance with its terms, except as enforceability of the obligations hereunder and thereunder are subject to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or Law).

4.2 <u>Subsidiaries</u>. The only subsidiaries of Buyer are Standard Gold Corp. and RMMC (each a "<u>Subsidiary</u>" and, collectively, the "<u>Subsidiaries</u>"). Each Subsidiary is a corporation duly organized, validly existing and in good standing under the Law of the State of Nevada. Each Subsidiary is duly qualified to do business and is in good standing in the states required due to: (i) the ownership or lease of real or personal property for use in the operation of its business; or (ii) the nature of the business conducted by such Subsidiary. Each Subsidiary has all requisite power, right and authority to own, operate and lease its properties and assets, to carry on its business as now conducted, to execute, deliver and perform its obligations under each agreement to which it is a party, and to carry out the transactions contemplated thereby. Buyer legally and beneficially owns all of the equity interests in the Subsidiaries, free and clear of any Encumbrance, and any agreement, option, right or privilege capable of becoming an Encumbrance. There are no options, warrants, conversion privileges or other rights, shareholder rights plans, agreements, arrangements or commitments (pre-emptive, contingent or otherwise) of any character whatsoever requiring or which may require the issuance, sale or transfer of any securities of any Subsidiary, or any securities or obligations convertible into, or exchangeable or exercisable for, or otherwise evidencing a right or obligation to acquire, any securities of either Subsidiary. All outstanding equity interests in each Subsidiary have been duly authorized and validly issued, and are fully paid and non-assessable.

4.3 <u>Issuance of Securities</u>. The Securities to be issued to Sellers or any of their Affiliates pursuant to this Agreement, and the Warrant Shares to be issued upon due exercise of the Consideration Warrants in accordance with their terms, when issued and delivered in accordance with the terms of this Agreement, will be duly and validly issued and will be fully paid and non-assessable.

4.4 Exemption from Prospectus Requirements. The issuance of the Securities will be exempt from the prospectus and registration requirements of applicable securities Law and no document will be required to be filed and no proceeding taken or approval, permit, consent, order or authorization obtained under Canadian Securities Law in connection with the first trade of the Securities; *provided* that: (a) Buyer is and has been a reporting issuer in a jurisdiction in Canada for the four months immediately preceding the trade; (b) at least four months have elapsed from the Closing Date; (c) the certificates representing the Consideration Shares and the Consideration Warrants carry a legend stating that "Unless permitted under securities legislation, the holder of this security must not trade the security before [insert date that is 4 months and a day after the Closing Date]"; (d) such trade is not a control distribution (as such term is defined in National Instrument 45-102 – Resale Restrictions); (e) no unusual effort is made to prepare the market or to create a demand for the Securities that are subject of the trade; and (f) no extraordinary commission or consideration is paid to a Person in respect of the trade.

4.5 Authorization; Enforcement. The execution, delivery and performance of this Agreement and the other Transaction Documents by Buyer, and the consummation of the Transactions contemplated hereby and thereby, including the creation, issuance and delivery of the Consideration Units to be issued to Sellers or any of their Affiliates pursuant to this Agreement, and the Warrant Shares to be issued upon due exercise of the Consideration Warrants in accordance with their terms, will not: (a) constitute a violation (with or without the giving of notice or lapse of time, or both) of any provision of any Law applicable to Buyer; (b) require any consent, approval or authorization of, or declaration, filing or registration with, any Person (except as may be required under applicable securities Law) except (i) such as have already been obtained, (ii) notification of a change of control to be provided by the Buyer following the Closing to the OTC Markets Group, Inc. and (iii) any acceptance of the CSE of the Transactions contemplated by the Transaction Documents, if necessary, which will be obtained by the Closing Date; (c) result in a default (with or without the giving of notice or lapse of time, or both) under, acceleration or termination of, or the creation in any party of the right to accelerate, terminate, modify or cancel, any agreement, lease, note or other restriction, Encumbrance, obligation or liability to which Buyer is a party or by which it is bound or to which any assets of Buyer are subject; (d) result in the creation of any Encumbrance upon the assets of Buyer, or upon any Common Shares or other securities of Buyer; (e) conflict with or result in a breach of or constitute a default under any provision of the certificate of incorporation or by-laws of Buyer; (f) invalidate or adversely affect any permit, license, authorization or status used in the conduct of the business of Buyer; or (g) restrict, hinder, impair or limit the ability of Buyer to carry on its business in any material respect as and where it is now being carried on.

4.6 <u>SEC Filings.</u> Buyer is subject to, and in material compliance with, the reporting requirements of Section 13 or 15(d) of the Exchange Act. Buyer has made available to Sellers through the EDGAR system true and complete copies of each of Buyer's Quarterly Reports on Form 10-Q, Annual Reports on Form 10-K and Current Reports on Form 8-K, in each case filed since January 1, 2018 (collectively, the "<u>SEC Filings</u>"). The SEC Filings, when they were filed with the SEC (or, if any amendment with respect to any such document was filed, when such amendment was filed), complied in all material respects with the applicable requirements of the Exchange Act and the rules and regulations thereunder and did not, as of such date, contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading. Buyer and each of the Subsidiaries, are engaged in all material respects only in the business of Buyer and the SEC Filings contain a complete and accurate description in all material respects of the business of Buyer and the Subsidiaries.

4.7 <u>Reporting Issuer Status.</u> Buyer is a reporting issuer in the provinces of British Columbia and Ontario, is not in default of any of the requirements of Canadian Securities Law in such provinces and its name does not appear on the lists of defaulting reporting issuers maintained by the Canadian Securities Authorities.

4.8 <u>Listing Status.</u> All of the issued and outstanding Common Shares are listed and posted for trading on the CSE and no order, ruling or determination having the effect of ceasing or suspending trading of any securities of Buyer or prohibiting from trading of Buyer's issued securities has been issued and no proceedings for such purpose have been instituted or, to the Knowledge of Buyer, are pending or contemplated or threatened.

4.9 <u>No Cease Trade.</u> The Common Shares are not subject to or affected by any actual or, to the Knowledge of Buyer, pending or threatened cease trade orders of, compliance or denial of use of exemption orders of, or action, investigation or proceeding by or before, any securities regulatory authority, court, administrative agency or other tribunal.

4.10 <u>Continuous Disclosure Documents.</u> Since January 1, 2018, Buyer has filed with all applicable securities regulatory authorities all forms, reports and documents required to be filed by it pursuant to all applicable Law, instruments, and policies, and all such filings, when made, complied in all material respects with the then applicable Law, instruments and policies. Such documents are accurate as at the date thereof, in all material respects, and may be relied upon by Sellers in consummating the Transactions contemplated hereby and thereby; and in particular, without limiting the foregoing, Buyer has made timely disclosure of all material changes relating to it and no such disclosure has been made on a confidential basis and there is no material change relating to Buyer which has occurred with respect to which the requisite material change report has not been filed.

4.11 <u>No Misrepresentation</u>. The documents publicly disclosed by Buyer do not contain any untrue statement of a material fact as at the date thereof nor do they omit to state a material fact which, at the date thereof, was required to have been stated or was necessary to prevent a statement that was made from being false or misleading in the circumstances in which it was made and were prepared in accordance with and complied with applicable securities Law.

4.12 <u>Compliance with Applicable Corporate and Securities Laws.</u> Buyer has complied and will comply in all material respects with the requirements of all applicable corporate and securities Law, including, without limitation, all applicable Law with respect to the issuance of the Consideration Units.

4.13 <u>Capitalization and Additional Issuances.</u> The authorized and outstanding capital stock of Buyer on a fully diluted basis are as set forth in the SEC Filings as of the dates set forth therein. As of the date of this Agreement, 161,230,237 Common Shares and 19,229,166 shares of Series B Preferred Stock (each convertible into one Common Share) are issued and are outstanding as fully paid and non-assessable, and no shares of Series A Preferred Stock are outstanding; all of the Common Shares and preferred stock have been issued in compliance with applicable Law, including applicable securities Law and the policies of the CSE. Other than pursuant to the transactions contemplated by the Subscription Agreements (the "Financing Transaction"), pursuant to the conversion of shares of Series B Preferred Stock as set forth above, and as set forth in Exhibit E, no Person has any agreement or option or right or privilege (whether at Law, pre-emptive or contractual) capable of becoming an agreement for the purchase, subscription or issuance of, or conversion into, any unissued shares, securities, warrants or convertible obligations of any nature of Buyer. The Consideration Units, upon issuance, will not be issued in violation of or subject to any pre-emptive rights, participation rights or other contractual rights to purchase securities issued by Buyer. The only officer, director, employee and consultant stock option or stock incentive plan or similar plan currently in effect is described in the SEC Filings.

4.14 <u>Shareholder Rights Plan.</u> Buyer has not entered into a shareholder rights plan agreement or put in place a shareholder rights plan.

4.15 <u>Transfer Agent.</u> National Securities Administrators Ltd. has been duly appointed as the transfer agent and registrar for the issued and outstanding Common Shares in the United States and Canada.

4.16 <u>Private Placement</u>. No registration under the Securities Act is required for the issuance of the Consideration Units by Buyer to Sellers or any of their Affiliates as contemplated hereby.

4.17 <u>Investment Company</u>. Buyer is not, and is not an affiliate of, and immediately after closing of the Transactions will not be an affiliate of, an "investment company" within the meaning of the Investment Company Act. Buyer shall conduct its business in a manner so that it will not become subject to the Investment Company Act.

4.18 <u>Bankruptcy</u>, <u>Insolvency and Reorganization</u>. None of Buyer or any Subsidiary is an insolvent Person within the meaning of the *Bankruptcy and Insolvency Act* (Canada) or under Title 11 United States Code Section 101, as applicable, and none of Buyer or any Subsidiary has made an assignment in favor of its creditors nor a proposal in bankruptcy to its creditors or any class thereof nor had any petition for a receiving order presented in respect of it nor undertaken a demerger or the transfer of its statutory seat or effective place of its management. None of Buyer or any Subsidiary has initiated proceedings with respect to a compromise or arrangement with its creditors or for its winding-up, liquidation or dissolution or a demerger or the transfer of its statutory seat or effective place of its management. No receiver is currently appointed in respect of Buyer or any Subsidiary, or their respective property or assets and no execution or distress is currently levied upon any of the property or assets of Buyer or any Subsidiary. No act or proceeding has been taken or authorized by or against Buyer or any Subsidiary with respect to any amalgamation, merger, consolidation, arrangement or reorganization of, or relating to Buyer or any Subsidiary nor have any such proceedings been authorized by any other Person.

4.19 <u>Material Contracts.</u> The material contracts of Buyer and the Subsidiaries, to which Buyer or any Subsidiary is a party or to which they, or any of their respective properties or assets, may be subject or by which Buyer or any Subsidiary is bound as of the Closing Date are: (a) the Transaction Documents; and (b) otherwise described in the SEC Filings (the "<u>Material Contracts</u>"). All Material Contracts are in full force and effect (other than the Transaction Documents to be entered into on Closing, which are in full force and effect on Closing only), and Buyer or a Subsidiary, as applicable, is entitled to all rights and benefits thereunder in accordance with the terms thereof. All of the Material Contracts are valid and binding obligations of Buyer or a Subsidiary, as applicable (other than the Transaction Documents to be entered into on Closing only), enforceable in accordance with their respective terms, except as may be limited by bankruptcy, insolvency and other applicable Law affecting the enforcement of creditors' rights generally and subject to the qualification that equitable remedies may only be granted in the discretion of a court of competent jurisdiction and there exists no default by Buyer or event of default or event, occurrence, condition or act, which, with the giving of notice, the lapse of time or the happening of any other event or condition, would become a default or event of default by Buyer under any Material Contract and, to the Knowledge of Buyer, all covenants to be performed by any other party to such agreements have been fully performed in all material respects and Buyer has not received notification from any party that it is in breach of or default under any Material Contracts.

## 4.20 Financial Statements; Internal Controls.

(a) Buyer's audited consolidated financial statements included in Buyer's Form 10-K for the year ended December 31, 2019 and the unaudited consolidated financial statements included in Buyer's quarterly reports on Form 10-Q for the quarters ended March 31, 2020 and June 30, 2020, filed in the SEC Filings (the "<u>Financial Statements</u>") were prepared in accordance with generally accepted accounting principles in the United States applied on a consistent basis ("<u>GAAP</u>") and PCAOB standards and present fairly in all material respects the financial condition and position of Buyer, taken as a whole and including all consolidation adjustments and other GAAP adjustments, as at the dates thereof and the results of operations and changes in financial position of Buyer for the periods covered by such financial statements subject to, in the case of the interim financial statements, normal and recurring year-end adjustments that will not, individually or in the aggregate, be material, and are in all material respects complete and correct, and do not contain or reflect any material inaccuracies or discrepancies. There has been no material change in the accounting policies of Buyer since the date of the last Financial Statement.

(b) The books of account and other financial records of Buyer: (i) reflect all items of income and expense and all assets and liabilities required to be reflected therein in accordance with local tax and accounting (excluding any purchase price allocation entries and other GAAP adjustments) regulations and GAAP applied on a consistent basis in accordance with the past practices of Buyer; and (ii) have been maintained in accordance with good business practices, GAAP and PCAOB requirements.

(c) Buyer maintains systems of internal accounting controls sufficient to provide reasonable assurance that: (i) transactions are executed in accordance with management's general or specific authorizations; (ii) transactions are recorded as necessary to permit preparation of consolidated financial statements for the group of entities in conformity with GAAP; and (iii) access to assets is permitted only in accordance with management's general or specific authorization.

(d) Each SEC Filing containing financial statements was accompanied by the certifications required to be filed or submitted by Buyer's chief executive officer and chief financial officer pursuant to the *Sarbanes-Oxley Act of 2002* (the "<u>Sarbanes-Oxley Act</u>"); at the time of filing or submission of each such certification, such certification was true and accurate and complied with the Sarbanes-Oxley Act and the rules and regulations promulgated thereunder; such certifications contain no qualifications or exceptions to the matters certified therein and have not been modified or withdrawn; and neither Buyer nor any of its officers has received notice from any Governmental Authority questioning or challenging the accuracy, completeness, form or manner of filing or submission of such certification.

(e) Save and except as disclosed in the Financial Statements, Buyer and each of its Affiliates does not have any loans, notes or other indebtedness outstanding to any of its shareholders, officers, directors or employees, past or present, or any Person not dealing at "arm's length" (as such term is used in the *Income Tax Act* (Canada)).

(f) The auditors of Buyer are independent public accountants as required by applicable securities legislation, including Article 2 of Regulation S-X, and there has not been any "reportable event" (within the meaning of National Instrument 51-102 – *Continuous Disclosure Obligations* of the Canadian Securities Administrators) or required disclosure under Item 4.01 of Form 8-K, that has not otherwise been reported, with the present or former auditor of Buyer and the auditor of Buyer has not provided any material comments or recommendations to Buyer regarding its accounting policies, internal control systems or other accounting or financial practices that have not been implemented by Buyer.

#### 4.21 <u>Tax Representations.</u>

(a) Buyer and each Subsidiary have duly and timely filed (or there have been filed on their behalf) with the appropriate Governmental Authority all Tax Returns required to be filed by or on their behalf for all periods ending prior to the date hereof. All such Tax Returns are complete and correct in all material respects and no material fact has been omitted therefrom. No extension of time in which to file any such Tax Returns is in effect. All Taxes shown on all such Tax Returns or any assessments or reassessments in respect of any such Returns, and all Taxes required to be paid by Buyer and each Subsidiary, have been paid in full. Buyer has made a provision for Taxes in the Financial Statements, which constitute an adequate provision for the payment of all Taxes.

(b) No Governmental Authority has asserted, or is threatening to assert, against Buyer or any Subsidiary any deficiency or claim for Taxes, and no Governmental Authority has challenged or disputed a filing position taken by Buyer or any Subsidiary in any Tax Return.

(c) There are currently no tax actions, suits, proceedings, audits, investigations or claims now existing, pending or threatened against Buyer or any Subsidiary. No tax reassessments in respect of Buyer or any Subsidiary have been issued and are outstanding, and there are no outstanding issues or matters in respect of taxes which have been raised and communicated to Buyer or any Subsidiary by any Governmental Authority. Neither Buyer nor any Subsidiary has executed or filed with any Governmental Authority any agreement or waiver extending the period for assessment, reassessment, payment or collection of any Taxes, or otherwise waived any statute of limitations in respect of Taxes.

(d) No Encumbrance for Taxes has been filed or exists in respect of Buyer or any Subsidiary other than for Taxes not yet due and payable.

#### 4.22 Employment Matters.

(a) Except for certain indebtedness of Buyer to David Beling, Buyer's Chief Executive Officer, in the amount of US\$654,000 as at June 30, 2020, and a severance payment to David Beling of US\$200,000 triggered on Closing, neither Buyer nor any Subsidiary is a party to or bound by any oral or written contract or commitment providing for: (i) severance, notice of termination or pay in lieu of notice of termination or termination, severance, retention or similar payments; or (ii) cash or other compensation or benefits to any director, officer, employee, consultant or agent upon or as a result of the execution of the Transaction Documents or the consummation of the Transactions contemplated in hereby or thereby.

(b) Buyer is in compliance, in all material respects, with the provisions of all federal, state, provincial, municipal, local and foreign Law respecting employment and employment practices, terms and conditions of employment and wages and hours.

(c) Buyer does not have, and has never had, employee plans other than Buyer's 2011 and 2017 Equity Incentive

# Plans.

#### 4.23 Environmental Matters.

(a) Buyer and each Subsidiary: (i) is in material compliance with all applicable Law relating to the protection of human health and safety, the environmental or hazardous or toxic substances or wastes, pollutants or contaminants (collectively, the "<u>Environmental Laws</u>"); (ii) has received all necessary Permits required of it under applicable Environmental Laws to conduct its business as presently conducted (collectively, the "<u>Environmental Permits</u>"); (iii) is in material compliance with all terms and conditions of each Environmental Permit; (iv) confirms that here have been no past, and there are no current claims, complaints, notices or requests for information received by Buyer or any Subsidiary with respect to any alleged violation of any Environmental Laws that, individually or in the aggregate, has or may reasonably be expected to have, a material adverse effect on the business, operators or financial condition of Buyer or any Subsidiary; and (v) to the Knowledge of Buyer, there are no events or circumstances that would reasonably be expected to form the basis of an order for clean-up or remediation, or an action, suit or proceeding by any other Person or governmental body or agency, against or affecting Buyer or any Subsidiary relating to Environmental Laws.

(b) Except in compliance with Environmental Permits and Environmental Laws, neither Buyer nor any Subsidiary has used or permitted to be used any of its assets or facilities, whether owned, leased, occupied, controlled or licensed or which it owned, leased, occupied, controlled or licensed at any prior time, to generate, manufacture, process, distribute, use, treat, store, dispose of, transport or handle any hazardous waste, material or substance as defined in or pursuant to any Environmental Laws by which Buyer is bound or subject.

## 4.24 Litigation Matters.

(a) There are no actions, suits, proceedings, inquiries or investigations existing, pending or, to the Knowledge of Buyer, threatened or adversely affecting Buyer or any Subsidiary or to which any of their property or assets is subject, at law or in equity, or before or by any Governmental Authority, and none of Buyer nor any Subsidiary are subject to any judgment, order, writ, injunction, decree or award of any Governmental Authority.

(b) To the Knowledge of Buyer, there is no pending change or contemplated change to any applicable Law that would or may reasonably be expected to have a material adverse effect on the business, operators or financial condition of Buyer or any Subsidiary.

(c) Neither Buyer nor any Subsidiary, nor any other Person associated with or acting on behalf of Buyer or any Subsidiary (including, without limitation, any Representative of Buyer or any Subsidiary) has, directly or indirectly, while acting on behalf of Buyer or any Subsidiary: (i) used any corporate funds for unlawful contributions, gifts or other unlawful expenses relating to political activity; (ii) made any unlawful payment to foreign or domestic governmental officials or employees or to foreign or domestic political parties or campaigns from corporate funds; or (iii) violated or alleged to have violated any provisions of the *Foreign Corrupt Practices Act of 1977* and the rules and regulations thereunder, the *Corruption of Foreign Public Officials Act* (Canada) or any applicable Law of similar effect in a jurisdiction.

(d) The operations of Buyer and each Subsidiary are and have been conducted at all times in compliance with applicable financial recordkeeping and reporting requirements of the anti-money laundering statutes of all applicable jurisdictions, the rules and regulations promulgated thereunder and any related or similar rules, regulations, guidelines issued, administered or enforced by each applicable Governmental Authority (collectively, the "<u>Anti-Money Laundering Laws</u>") and no action, suit or proceeding by or before any applicable court of governmental agency, authority or body or any arbitrator involving Buyer or any Subsidiary with respect to the Anti-Money Laundering Laws is pending or, to the Knowledge of Buyer, threatened.

(e) Neither Buyer, any Subsidiary nor, to the Knowledge of Buyer, any Representative thereof is currently the target of any U.S. sanctions administered by the Officer of Foreign Assets Control of the U.S. Treasury Department ("<u>OFAC</u>") and Buyer will not directly or indirectly use the proceeds of the Financing Transaction or lend, contribute or otherwise make available such proceeds to any subsidiary, joint venture partner or other Person, for the purpose of financing the activities of any Person that is currently the target of any U.S. sanctions administered by OFAC.

(f) No property or assets of Buyer or any Subsidiary has been taken or expropriated by any Governmental Authority nor has any notice or proceeding in respect thereof been given or commenced nor, to the Knowledge of Buyer, is there any intent or proposal to give any such notice or to commence any such proceeding.

4.25 <u>Material Properties</u>. Buyer's only interest in a mineral project on a property material to Buyer is the Bullfrog Gold Project located in Nye County, Nevada which, for greater certainty, shall be deemed not to include the Company Assets until immediately following Closing (the "<u>Bullfrog Project</u>").

4.26 <u>Scientific and Technical Information</u>. Buyer is in compliance with the provisions of National Instrument 43-101 – *Standards of Disclosure for Mineral Projects* ("<u>NI 43-101</u>") and has filed all technical reports in respect of its properties (and properties in respect of which it has a right to earn an interest) required thereby. The technical report of Buyer entitled "NI 43-101 Technical Report, Mineral Resource Estimate, Bullfrog Gold Project, Nye County, Nevada" dated effective August 9, 2017 (the "<u>Technical Report</u>") remains current as of the date hereof. The Technical Report complies in all material respects with the requirements of NI 43-101 and there is no new scientific or technical information concerning the Bullfrog Project since the date thereof that would require a new technical report in respect of such properties, respectively, to be issued under NI 43-101. Buyer and each Subsidiary has made available to the author of the Technical Report, prior to the issuance thereof, for the purpose of preparing such report, all information requested by the author and none of such information contained any misrepresentation at the time such information was provided. The information publicly disclosed by Buyer relating to scientific and technical information has been prepared in accordance with NI 43-101 and in compliance with all applicable securities Law.

## 4.27 Bullfrog Project.

(a) Buyer and/or each Subsidiary has sufficient title, clear of any material title defect or Encumbrance, to the Bullfrog Project and has good and sufficient title to all real property interests including, without limitation, fee simple estate of and in real property, leases, easements, rights of way, surface use agreements, permits or licenses from landowners or authorities permitting the use of land by Buyer and/or each Subsidiary necessary to permit the operation of their businesses as presently owned and conducted. Buyer and/or each Subsidiary holds all mineral rights, Permits and water rights required to continue their businesses and operations as currently conducted and Buyer and/or each Subsidiary has not received any notice, whether written or oral, from any Person with jurisdiction or applicable authority of any revocation or intention to revoke or cancel Buyer's or any Subsidiary's interest in any material mining rights, Permits or water rights nor has Buyer or any Subsidiary received any compliance orders, citations or notices relating to non-compliance or alleged non-compliance, relating to the mining rights, Permits and water rights are free and clear of Encumbrances and royalty burdens other than as described in the documents publicly disclosed by Buyer. Buyer and/or each Subsidiary, as applicable, has the exclusive right to deal with mineral rights held by Buyer and/or each Subsidiary, the Permits and the water rights.

(b) The mineral rights, Permits and water rights related to the Bullfrog Project which are held by Buyer and/or Subsidiaries are in good standing and have been properly located and recorded in compliance with applicable Law.

(c) All assessment or other work required to be performed in relation to the Bullfrog Project in order to maintain Buyer's and/or each Subsidiary's interests therein, if any, have been performed to date and Buyer and each Subsidiary has complied with all applicable Law in connection therewith.

(d) Any and all Taxes and other payments required to be paid in respect of the Bullfrog Project have been paid and any and all filings required to be filed in respect of the Bullfrog Project have been filed.

(e) Other than Buyer and/or each Subsidiary, no other Person has any interest in the Bullfrog Project or any right to acquire any such interest and there are no back-in rights, earn-in rights, rights of first refusal, royalty rights or similar provisions other than as disclosed by Buyer in the SEC Filings which would affect Buyer's or any Subsidiary's interests in the Bullfrog Project.

(f) There are no mining concessions, claims, interests in real and immoveable property, leases, licenses, permits or any other rights conflicting with the Bullfrog Project.

(g) Neither Buyer nor any Subsidiary has received any notice, whether written or oral from any Governmental Authority of any default under, or of any revocation or limitation, or intention to revoke or limit, any mineral rights, Permits or water rights held by Buyer and/or each Subsidiary in connection with the Bullfrog Project.

(h) Buyer and/or each Subsidiary possess all material licenses, Permits, certificates, consents, orders, grants, approvals, classifications, program participation requirements, sign-offs, registrations, or other authorizations of and from any Governmental Authority (collectively, the "<u>Bullfrog Permits</u>") necessary to carry on the business currently carried on, or contemplated to be carried on by them. The Bullfrog Permits are in compliance in all material respects with the terms and conditions of all the Bullfrog Permits and with all Law related to the operations thereof. Neither Buyer nor any Subsidiary has received any written or other notice of the modification, revocation, limitation or cancellation of, or any intention to modify, revoke, limit or cancel, or any proceedings relating to the modification, revocation, limitation or cancellation of any of the Bullfrog Permits. Buyer and each Subsidiary are in compliance in all material respects with the terms and conditions of all of the Bullfrog Permits and with all Law related to the operations thereof.

4.28 <u>COVID-19</u>. Except as mandated by, or in conformity with the recommendations of a Governmental Authority, there has been no closure, demobilization, shut-down, suspension, postponement or disruption to any of Buyer's or any Subsidiary's properties or to the operations and workforce productivity of Buyer and each Subsidiary as a result of the novel coronavirus outbreak.

4.29 <u>No Brokerage Fee.</u> There is no Person acting or purporting to act on behalf of Buyer who is entitled to any brokerage, agency or other fiscal advisory or similar fee in connection with the Transactions contemplated herein, other than Fort Capital Partners which is entitled to a cash payment of CDN\$1,200,000 on Closing.

4.30 <u>Minute Books</u>. The minute books of Buyer and each Subsidiary contains copies of all constating documents and all proceedings of securityholders and directors (and committees thereof) and are completed in all material respects.

4.31 <u>No Agreements Affecting Voting Control.</u> Other than the Investor Rights Agreement to be entered into on Closing, no agreement is in force or effect which will in any manner affect the voting or control of any of the securities of Buyer.

4.32 <u>Insurance</u>. The assets of Buyer and each Subsidiary and their businesses and operations are insured against loss or damage with responsible insurer on a basis consistent with insurance obtained by reasonably prudent participants in comparable businesses, and such coverage is in full force and effect, and neither Buyer nor any Subsidiary has breached the terms of any policies in respect thereof or failed to promptly give any notice or present any material claim thereunder.

4.33 <u>Full, True and Complete Disclosure</u>. Buyer has made available to Sellers and its advisors, or publicly disclosed, all of the information relating to Buyer that Sellers have requested for deciding whether to complete the Transactions; none of the foregoing representations, warranties and statements of fact and no other statement furnished by or on behalf of Buyer to Sellers or their Representatives in connection with the negotiation of the Transactions, contain any untrue statement of a material fact or omit to state any material fact necessary to make such statement or representation not misleading under the circumstances in which such statement or representation is made.

## 4.34 Independent Review.

(a) Buyer and its Affiliates and its and their respective Representatives have had sufficient access to and opportunity to review the Company Assets and the Books and Records and to ask questions of the management representatives and professional advisors of Sellers, as necessary for Buyer to investigate, analyze, and evaluate the Company Assets and the Books and Records, and to make its and their independent decision to consummate the Transactions.

(b) In making the decision to enter into the Transaction Documents to which Buyer is a party, and to consummate the Transactions, Buyer has conducted its own independent investigation, analysis, and evaluation of the Company, the Company Assets and the Books and Records (including Buyer's own estimate and appraisal of the extent, location and value of mineralization, mineral resources and reserves, undeveloped properties, and environmental obligations), and the financial condition of, operations, and prospects for, the Company.

#### ARTICLE 5 COVENANTS

# 5.1 <u>Conduct of Business Prior to the Closing</u>. From the Execution Date until the Closing:

(a) Sellers shall use reasonable commercial efforts to maintain the Company Assets as currently maintained, except: (i) as expressly required or permitted by this Agreement; (ii) as required pursuant to applicable Laws; (iii) as consented to in writing by Buyer (which consent shall not be unreasonably withheld, conditioned or delayed); or (iv) as required to reasonably and prudently respond to an emergency or disaster (including the right to take forthwith any action required to insure the safety and integrity of the Company Assets). For greater certainty, the Parties acknowledge and agree that the foregoing covenant shall not be construed as prohibiting or impairing the ability of Sellers to give effect to the Contribution and Conversion prior to Closing.

(b) Upon the completion of the Contribution and Conversion, Sellers: (i) shall maintain the Company in good standing under the Laws of the State of Delaware; (ii) shall maintain the qualification of the Company to do business under the Laws of the State of Nevada; (iii) shall operate the Company in material compliance with the LLC Agreement; and (iv) shall not issue additional Membership Units, or enter into any agreement or arrangement for the issuance of additional Membership Units or securities convertible into Membership Units.

5.2 <u>Access to Information</u>. From the Execution Date until the Closing, the Administrative Agent shall cause BBI or the Company, as applicable, to: (a) afford Buyer and its Representatives reasonable access to and the right to inspect the Company Assets and the Books and Records; and (b) instruct the Representatives of Sellers, BBI and the Company to reasonably cooperate with Buyer in its investigation of the Company Assets and Books and Records; *provided, however*, that any such investigation shall be conducted at the sole risk of Buyer and its Representatives during normal business hours upon reasonable advance notice to the Administrative Agent, under the supervision of the Administrative Agent and its Representatives in compliance with all applicable Laws and the health and safety procedures of Sellers. Notwithstanding anything to the contrary in this Agreement, neither Seller, BBI nor the Company shall be required to disclose any information to Buyer if such disclosure would, as determined in the discretion of the Administrative Agent: (i) jeopardize any Privilege; (ii) contravene any applicable Law or fiduciary duty, or binding agreement entered into prior to the Execution Date; or (iii) relate to Company assets other than the Company Assets or the transactions contemplated in connection with the Contribution and Conversion. The Books and Records will be provided to Buyer and its Representatives on an "as is, where is" basis.

5.3 <u>Notice of Certain Events</u>. Sellers and Buyer agree that, subject to applicable Law, each shall provide the other (with any notice to Sellers directed to the Administrative Agent) prompt notice in writing of:

(a) any notice or communication from any Person alleging that the consent of such Person is or may be required in connection with the Transactions;

(b) any material notice or communication from any Governmental Authority in connection with the Transactions;

(c) any material proceeding commenced or threatened against it or the Company, as applicable, which relates to the consummation of the Transactions or would reasonably be expected to have a Material Adverse Effect; and

(d) any failure by it to comply with or satisfy in any material respect any covenant, condition or agreement to be complied with or satisfied under this Agreement, and copies of all documents related thereto; *provided* that the giving of any such notice shall not in any way change or modify the representations and warranties of the Parties or the conditions in their favor contained in this Agreement or otherwise affect the remedies available to Sellers and Buyer under this Agreement.

#### 5.4 Announcements and Confidentiality.

(a) Subject to the terms of this Agreement and <u>Section 5.4(b)</u>, no announcement or communication (each an "<u>Announcement</u>") concerning the existence or content of this Agreement and documents referred to herein shall be made by Sellers or Buyer without the prior written approval of the other Party or Parties.

(b) Section 5.4(a) does not apply to any Announcement if, and to the extent that such is required by any Governmental Authority (or the rules thereof) to which Sellers or Buyer (as applicable) are subject, provided that Sellers or Buyer (as applicable) shall, to the extent permitted by applicable Law and so far as is practicable, inform the other of such requirement and the information required to be disclosed, consult with the other as to possible steps to avoid or limit disclosure, take such of those steps as the other may reasonably require and, where the disclosure is to be made by way of public announcement, make reasonable efforts to agree to the wording of the announcement with the other in advance. Without limiting the generality of the foregoing, the Parties acknowledge and agree that Buyer will be required to file a Current Report on Form 8-K with the SEC that discloses the material terms of this Agreement and, subject to Section 5.4(e), attaches a copy of this Agreement as an exhibit in accordance with the Exchange Act.

(c) Each Party shall and shall cause its Affiliates to, subject to <u>Section 5.4(d)</u>, treat as strictly confidential all information received or obtained as a result of entering into or performing this Agreement which relates to:

(i) the subject matter and provisions of the Transaction Documents;

(ii) the negotiations relating to the Transaction Documents; or

(iii) an other Party.

(d) Notwithstanding <u>Section 5.4(c)</u>, a Party may disclose information which would otherwise be confidential if and to the extent:

(i) the information is disclosed to Augusta, its Affiliates or its Representatives;

(ii) permitted by this Agreement (including <u>Section 5.5</u>);

(iii) required by applicable Law;

(iv) required by any Governmental Authority with authority over such Party or any of its Affiliates, subject to Section 5.4(b);

(v) disclosure is made to its Representatives, provided that any such Representative is first informed of the confidential nature of the information and such Representative acts in accordance with the provisions of <u>Section 5.4(c)</u> as if it were a party hereto;

(vi) the information has come into the public domain through no fault of that Party; or

(vii) the other Party or in the case of Sellers, the Administrative Agent, has given prior written approval to the disclosure, *provided* that, except as permitted pursuant to Section 5.4(d)(i), any disclosure shall, so far as is practicable, be made only after consultation with the other Party or Parties.

(e) Notwithstanding anything else in this Section 5.4, Sellers acknowledge and agree that if Buyer is required to file on SEDAR or EDGAR under applicable Law any Transaction Document to which either Seller is a party, Buyer may file a redacted form of such agreement subject to such reasonable redactions as the Administrative Agent may request; *provided* that such redactions are permitted under applicable Law. Any provision of any Transaction Document that has been so redacted shall continue to constitute confidential information for purposes of this Agreement and this Section 5.4; *provided*, *however*, that if any securities regulatory authority subsequently requires Buyer to disclose any such redacted information or such redacted information shall cease to be confidential information upon such disclosure; and (ii) Buyer shall not be in breach or violation of this Agreement with respect thereto. Buyer agrees that prior to filing any version of this Agreement or any related agreement with any securities regulatory authority, it shall provide Sellers with a reasonable opportunity to review and comment on all documents to be submitted in connection with such filing and shall consider in good faith the comments, if any, provided by the Administrative Agent in respect of such documents, provided that any decision regarding redactions will ultimately be determined by Buyer, acting reasonably.

5.5 <u>CSE Approval.</u> Buyer shall use its commercially reasonable efforts to obtain, on or before the Closing Date any necessary approvals of the CSE for the: (a) issuance of the Consideration Units on the terms and conditions set out in this Agreement; and (b) the listing of the Consideration Shares and the Warrant Shares on the CSE. Buyer shall, in a prompt manner, keep Sellers informed of all correspondence and discussions with the CSE in connection with the issuance of the Consideration Units and the listing of the Consideration Shares and the Common Shares issuable on the exercise of the Consideration Warrants, as applicable, and provide Sellers copies of all applications, forms and documents submitted to the CSE in connection therewith. If requested by Sellers, Buyer will allow Sellers and their legal counsel to participate in all discussions with the CSE relating to the Transactions.

5.6 <u>Governmental Approvals</u>. Each Party hereto shall, as promptly as practicable, use commercially reasonable efforts to obtain, or cause to be obtained, all consents, authorizations, orders and approvals from all Governmental Authorities that may be or become necessary for its execution and delivery of this Agreement and the consummation of the Transactions. Each Party shall cooperate fully with the other Party and its Affiliates in promptly seeking to obtain all such consents, authorizations, orders and approvals. The Parties hereto shall not willfully take any action that will have the effect of delaying, impairing or impeding the receipt of any required consents, authorizations, orders and approvals or the consummation of the Transactions.

5.7 <u>Contribution and Conversion</u>. Sellers shall take, or cause to be taken, all reasonable action and do, or cause to be done, all other things reasonably necessary, proper and advisable under applicable Law to complete the Contribution and Conversion as soon as reasonably practicable on terms satisfactory to Sellers in their sole discretion. The Administrative Agent shall provide written confirmation to Buyer upon the completion of the Contribution and Conversion.

5.8 <u>Seller Marks</u>. As soon as possible, but no later than 30 days after the Closing Date, Buyer shall discontinue the use of any Seller Marks, shall take all actions and make all filings necessary to discontinue any and all use of any Seller Marks, including as contemplated in <u>Section 2.5(c)</u>, and shall provide the Administrative Agent a certification no later than such deadline signed by an officer of Buyer confirming that Buyer has discontinued all use of Seller Marks.

5.9 <u>Closing Conditions</u>. From the Execution Date until the Closing, each Party shall, and Sellers shall cause BBI and the Company to, use commercially reasonable efforts to take such actions as are necessary to expeditiously satisfy the closing conditions set forth in Article 6 for which it is responsible.

5.10 <u>Further Assurances</u>. Following the Closing, each of the Parties hereto shall, and shall cause their respective Affiliates to, execute and deliver such additional documents, instruments, conveyances and assurances and take such further actions as may be reasonably required to carry out the provisions hereof and give effect to the Transactions.

5.11 <u>Replacement Reclamation Bonds</u>. Promptly after the Execution Date, the Administrative Agent or its Representatives and Buyer will meet with the relevant Governmental Authorities with respect to replacement of the sureties established by Sellers or their Affiliates to guarantee performance of the reclamation obligations of the Company related to the Company Assets as set forth on Schedule 5.11 of the Disclosure Schedule (the "Reclamation Bonds"). As quickly as possible thereafter, but in any event prior to Closing, Buyer shall deliver to the Administrative Agent or its Representatives evidence that Buyer has obtained, subject only to Closing, substitute guarantees, letters of credit, bonds, security deposits, cash or other surety obligations and evidence of financial capacity, in each case acceptable to the relevant Governmental Authorities in replacement of the Reclamation Bonds (the "Replacement Bonds"). As of the Closing Date, Buyer shall deliver to the applicable Governmental Authorities duly executed Replacement Bonds and Buyer shall use its best efforts to cause such agencies to fully and unconditionally release Sellers and their respective Affiliates from all obligations relating to the Reclamation Bonds and any liabilities related thereto.

5.12 Director and Officer Indemnification. Buyer agrees that all rights to indemnification, advancement of expenses and exculpation by the Company (including its predecessors) now existing in favor of each Person who is now, or has been at any time prior to the date hereof or who becomes prior to the Closing Date, an officer or director of the Company (including its predecessors), as provided in the LLC Agreement (or the constating documents of its predecessors), shall survive the Closing Date and shall continue in full force and effect in accordance with their respective terms. No amendment or other modification of the LLC Agreement after the Closing Date will adversely affect such rights to indemnification, advancement of expenses and exculpation.

### 5.13 Company Financial Statements.

(a) Buyer shall use reasonable commercial efforts prior to Closing to attempt to obtain an exemption from any requirement under the Exchange Act (the "Financial Statement Exemption") that necessitates the filing or disclosure by Buyer or its Affiliates of historical financial statements of the Company or its predecessors (the "Company Historical Financial Statements"), except such statements required to be filed pursuant to the Exchange Act as a result of the acquisition by Buyer of the Membership Units prepared on the basis that the Contribution and Conversion occurred immediately prior to the commencement of the relevant fiscal periods. Buyer shall provide to the Administrative Agent any proposed submission by Buyer or its Affiliates to any applicable Governmental Authorities related to the Financial Statement Exemption at least ten Business Days prior to the date on which Buyer or its Affiliates intend to make such submission (the "Financial Statement Exemption Review Period"). The Administrative Agent and its Representatives, acting reasonably shall have the right to review, comment and sign-off on any such written submission during the Financial Statement Exemption Review Period and Buyer shall give reasonable and due consideration to any comments made by the Administrative Agent and its Representatives. Buyer or its Affiliates shall not submit the Financial Statement Exemption to the applicable Governmental Authorities prior to the expiration of the Financial Statement Exemption Review Period without the prior written approval of the Administrative Agent, not to be unreasonably withheld. Buyer shall promptly (and in any event within two Business Days) provide a copy of its submission in respect of the Financial Statement Exemption to the relevant Governmental Authorities, and shall promptly provide copies of any correspondence or other communications related thereto, to the Administrative Agent.

(b) Sellers, acting through the Administrative Agent, shall provide the Company Historical Financial Statements to Buyer in accordance with the requirements of the Exchange Act, as modified by the Financial Statement Exemption, if applicable, at least three weeks prior to the deadline under the Exchange Act for Buyer to file or disclose such Company Historical Financial Statements to the applicable Governmental Authorities. The Company Historical Financial Statements will be prepared in accordance with GAAP, and will be independently audited or reviewed in accordance with the applicable auditing standards of the PCAOB.

5.14 <u>Directors' and Officers' Insurance</u>. Prior to the Closing Date, Buyer shall obtain and fully pay any premiums due in connection with a directors' and officers' insurance policy for all present and future directors and officers of Buyer, covering claims in respect of acts or omissions in their capacity as directors or officers of Buyer occurring on or after the Closing Date, on terms satisfactory to Sellers, acting reasonably (the "<u>D&O Insurance Policy</u>"). Buyer shall maintain at all times on and after the Closing Date the D&O Insurance Policy or a subsequent directors' and officers' insurance policy in an amount and with a scope no less favorable than the D&O Insurance Policy.

### 5.15 Buyer's Operations Prior to Closing.

(a) From the Execution Date until the Closing, except for the Transactions, (i) Buyer shall, and shall cause its Affiliates to, conduct its and their respective business and operations in the ordinary course of business and substantially as such business and operations are currently conducted; and (ii) Buyer shall not, and shall cause its Affiliates not to: (A) enter into any transaction or other arrangement that would have a material adverse effect on the business, results of operations, financial condition or assets of Buyer taken as a whole; (B) amend its constating documents; (C) adopt a plan of liquidation or resolutions providing for its liquidation or dissolution; or (D) sell, pledge or encumber (other than a Permitted Encumbrance) or otherwise transfer any assets having a value greater than US\$100,000 in the aggregate, other than in the ordinary course of business.

(b) From the Execution Date until the Closing, except for the Transactions, and except for any issuance of securities upon the conversion or exercise of any options, warrants or convertible preferred stock outstanding on the Execution Date, Buyer and its Affiliates shall not directly or indirectly: (i) issue, or enter into any agreements or other arrangements to issue, any capital stock or increase or decrease the number of shares of capital stock authorized for issuance; (ii) issue or authorize the issuance of, or enter into any agreements or other arrangements to issue, convertible securities or other rights, agreements, arrangements or commitments of any character relating to the capital stock of Buyer or any of its Affiliates to issue or sell any shares of capital stock or other equity interest in Buyer or any of its Affiliates; (iii) split, combine or reclassify any shares of its capital stock; (iv) redeem, repurchase, or otherwise acquire or offer to redeem, repurchase or otherwise acquire any of its outstanding securities; or (v) amend or agree to amend the terms of any of its outstanding securities.

(c) From the Execution Date until the Closing, Buyer shall not and shall cause its Affiliates not to enter into any agreement with any Person or group of Persons or take any steps to terminate its existence, to amalgamate or merge into another corporation, to continue into any other jurisdiction or to otherwise change its corporate existence or to support or cooperate with any Person or group of Persons that propose to make a tender offer or take-over bid for or otherwise acquire 50% or more of the then-outstanding Common Shares.

(d) From the Execution Date until the Closing, Buyer shall not negotiate, enter any agreement or provide information to or permit its employees, officers and agents to provide information to any Person in connection with: (i) any transactions that could interfere with or impede the Transactions contemplated by the Transaction Documents; or (ii) any merger, amalgamation or business combination.

(e) From the Execution Date until the Closing, Buyer shall use commercially reasonable efforts to: (i) maintain Buyer's status as a "reporting issuer" not in default under Canadian Securities Laws and to comply with its reporting obligations under the Exchange Act; and (ii) maintain the listing of the Common Shares on the CSE.

5.16 <u>Supplemental Disclosure</u>. Sellers may, from time to time prior to, but no less than five Business Days prior to, Closing, by written notice from the Administrative Agent to Buyer, amend, supplement, modify, create or add any section to the Disclosure Schedule in order to add information to, or change information previously set forth in, the Disclosure Schedule supplied to Buyer; *provided* that such amendment does not relate to a Material Adverse Effect. No such amendment, supplement, modification, creation or addition shall be evidence that the representations and warranties or disclosures previously made by Sellers were not true and correct as of the Execution Date or are no longer true or correct.

5.17 <u>Possession of Books and Records.</u> Buyer shall take possession of and transfer all Books and Records of the Company currently stored in Elko, NV and Salt Lake City, UT at its sole expense within ten Business Days of the Closing Date. Sellers shall provide access to such Books and Records during regular business hours to facilitate the satisfaction of this covenant by Buyer.

5.18 Post-Closing Access to Books and Records. The Administrative Agent may from time to time during normal business hours, after Buyer takes possession of the Books and Records in accordance with Sections 2.4(a)(vii) and 5.17, and upon reasonable notice and without undue interference to the business operations of Buyer, inspect and make copies (at its own expense) of the Books and Records, provided that its access to and use of such Books and Records will be limited to legal and regulatory purposes, including responding to tax audits, or otherwise dealing with Governmental Authorities.

5.19 <u>Final CSE Filings.</u> Promptly following Closing, Buyer shall make or cause to be made all filings, and shall pay all fees required to be given or made to the CSE in order to satisfy all of the conditions to the listing of the listing of the Consideration Shares and the Warrant Shares. Buyer shall promptly advise Sellers if any necessary approval of the CSE for the listing of the listing of the listing of the Consideration Shares and the Warrant Shares is not granted for any reason.

5.20 Improper or Unintended Transfers. To the extent that either Sellers or Buyer determine, after the Closing Date, that (i) any assets other than Company Assets (including contracts) or liabilities other than liabilities directly related to Company Assets were inadvertently not transferred or distributed by the Company to another Person or Persons as contemplated by the Contribution and Conversion, or (ii) Company Assets or liabilities directly related to Company Assets were inadvertently transferred or distributed by the Company to another Person or Persons in connection with the Contribution and Conversion, the Parties will work together to determine a mutually acceptable tax-efficient method to effect, to the greatest extent possible, the necessary transfers and assumptions to rectify such error or omission. Sellers shall reimburse Buyer and the Company for all reasonable and documented out-of-pocket costs and expenses (including Taxes) incurred by Buyer and the Company pursuant to this Section 5.20.

5.21 <u>Schedule 14F</u>. The Buyer shall prepare, file and mail to shareholders a Schedule 14F in accordance with Rule 14f-1 under the Exchange Act ("<u>Rule 14f-1</u>") as soon as practicable following receipt from Augusta of the information relating to the directors and officers to be appointed by Augusta required by the Buyer to comply with its obligations under Rule 14f-1 in satisfaction of the condition to Closing contained in <u>Section 6.3(e)</u>.

#### ARTICLE 6 CONDITIONS TO CLOSING

6.1 <u>Mutual Conditions.</u> The obligations of Sellers and Buyer to complete the sale and purchase of the Membership Units is conditional on the satisfaction of all of the following conditions, each of which is for the benefit of Sellers and Buyer and may only be waived by the mutual written consent of the Administrative Agent and Buyer:

(a) no Governmental Authority shall have issued a Governmental Order restraining or enjoining the Transactions and there shall not have been enacted or made applicable any Law that makes the Transactions illegal or otherwise prohibited;

(b) Buyer shall have completed all necessary filings with the CSE and received any necessary approvals from the CSE to allow for the issuance and listing of the Consideration Shares and the Warrant Shares on or prior to the Closing Date subject only to ordinary conditions typical for the issuance and listing of common shares;

(c) the Closing having occurred on or prior to the Drop Dead Date; *provided* that, with respect to any extension of the Drop Dead Date by the Parties pursuant to <u>Section 8.1(b)(i)</u>, the CSE shall have confirmed in writing, to the extent requested by the Administrative Agent, that: (i) the Consideration Shares and Consideration Warrants can be issued on the terms contemplated herein; and (ii) the Augusta Group Units can be issued on the terms contemplated in the Term Sheet, in each case at Closing; and

(d) the Transactions contemplated by the Subscription Agreements shall have been completed on the terms set forth therein at or contemporaneously with the Closing; *provided* that, to the extent that any of the Subscription Agreements is proposed to be amended by the parties thereto, the Administrative Agent shall have been provided with notice of the proposed amendments, including a copy of the proposed amending document, prior to the execution thereof and Buyer shall not have entered into or otherwise agreed to any such amendment without the prior written consent of the Administrative Agent.

6.2 <u>Conditions for the Benefit of Buyer</u>. The obligations of Buyer to complete the purchase of the Membership Units is conditional on the satisfaction of all of the following conditions, each of which is for the benefit of Buyer and may only be waived by Buyer in writing:

(a) the representations and warranties made by Sellers in <u>Article 3</u> shall be true and correct in all respects without regard to any materiality or the Material Adverse Effect qualifications contained in them, except where the failure of such representations and warranties to be true and correct do not constitute in aggregate a Material Adverse Effect, in each case, as of the Closing Date as if made on and as of such date (except to the extent that such representations and warranties speak to an earlier date, then as of such date, or except as affected by the Transactions contemplated or permitted by the Transaction Documents), and each Seller shall have provided to Buyer a certificate dated the Closing Date executed by a senior officer to the foregoing effect;

(b) each Seller shall have duly performed and complied in all material respects with all agreements, covenants and conditions required by the Transaction Documents to which it is a party to be performed or complied with by each such Seller prior to or on the Closing Date, including completion of the Contribution and Conversion, and each such Seller shall have delivered to Buyer a certificate dated the Closing Date executed by a senior officer to the foregoing effect with respect to such Seller's agreements, covenants and conditions;

(c) since the Execution Date, no Material Adverse Effect shall have occurred; and

(d) the Administrative Agent shall have delivered to Buyer each other document, certificate, deed and instrument set forth in Sections 2.4(a)(i) through (vii) inclusive.

6.3 <u>Conditions for the Benefit of Sellers.</u> The obligations of Sellers to complete the sale of the Membership Units is conditional on the satisfaction of all of the following conditions, each of which is for the benefit of Sellers and may only be waived by the Administrative Agent in writing:

(a) the representations and warranties made by Buyer in <u>Article 4</u> shall be true and correct in all respects without regard to any materiality or the material adverse effect qualifications contained in them, except where the failure of such representations and warranties to be true and correct do not constitute in aggregate a material adverse effect, in each case, as of the Closing Date as if made on and as of such date (except to the extent that such representations and warranties speak to an earlier date, then as of such date, or except as affected by the Transactions contemplated or permitted by the Transaction Documents), and Buyer shall have provided to Sellers a certificate dated the Closing Date executed by a senior officer to the foregoing effect;

(b) Buyer shall have duly performed and complied in all material respects with all agreements, covenants and conditions required by the Transaction Documents to which it is a party to be performed or complied with by Buyer prior to or on the Closing Date and Buyer shall have delivered to the Administrative Agent a certificate dated the Closing Date executed by a senior officer to the foregoing effect with respect to Buyer's agreements, covenants and conditions;

(c) Buyer shall have delivered written evidence satisfactory to the Administrative Agent in its sole discretion that the applicable Governmental Authorities have accepted the Replacement Bonds as replacements for the Reclamation Bonds;

discretion:

(d) the Contribution and Conversion shall have been completed on terms satisfactory to Sellers in their sole

(e) the Governance Changes shall have become effective and the Board Designee shall have been appointed to Buyer board of directors in accordance with the Investor Rights Agreement at or contemporaneously with the Closing, it being acknowledged that the Buyer will be required to file and mail an information statement on Schedule 14F in accordance with Rule 14f-1 at least 10 days prior to effecting such governance changes to the Buyer's board of directors; and

(f) Buyer shall have delivered to the Administrative Agent each other document, certificate, deed and instrument set forth in  $\underline{\text{Sections } 2.4(b)(i)}$  through  $\underline{(vi)}$  inclusive.

### ARTICLE 7 INDEMNIFICATION

7.1 <u>Indemnification Obligations of Sellers</u>. From and after the Closing, subject to the Liability Limits and <u>Section 7.3(b)</u>, Sellers shall jointly and severally indemnify, defend and hold harmless the Buyer Indemnified Parties from and against any Loss directly arising out of:

(a) any breach of a representation or warranty of Sellers set forth in <u>Article 3</u> of this Agreement, or in any certificate delivered pursuant to this Agreement by or on behalf of Sellers;

(b) any breach or violation of any covenant, agreement or undertaking in this Agreement made by Sellers; and

(c) any fraud, willful misrepresentation or intentional misconduct of Sellers in connection with this Agreement.

7.2 <u>Indemnification Obligations of Buyer</u>. Subject to the Indemnity Deed, from and after the Closing, subject to the Liability Limits and <u>Section 7.3(c)</u>, Buyer shall indemnify, defend and hold harmless Seller Indemnified Parties from and against any Loss directly arising out of:

(a) any breach of a representation or warranty of Buyer set forth in <u>Article 4</u> of this Agreement, or in any certificate delivered pursuant to this Agreement by Buyer;

(b) any breach or violation of any covenant, agreement or undertaking in this Agreement made by Buyer; and

(c) any fraud, willful misrepresentation or intentional misconduct of Buyer in connection with this Agreement.

#### 7.3 Survival of Representations and Warranties; Survival of Covenants.

(a) Except as set forth in Section 7.3(b) and Section 7.3(c), and subject to the limitations and other provisions of this Agreement, the representations and warranties contained herein shall survive the Closing and shall not merge. None of the covenants or other agreements contained in this Agreement shall survive the Closing Date other than those which by their terms contemplate performance after the Closing Date, and each such surviving covenant and agreement shall survive the Closing for the period contemplated by its terms.

(b) Sellers shall not be required to indemnify or save harmless the Buyer Indemnified Parties pursuant to <u>Section</u> <u>7.1</u>, unless Buyer shall have provided to the Administrative Agent a Notice of Claim in respect thereof within the following time limits:

(i) with respect to the representations and warranties set forth in <u>Sections 3.1</u> (Organization), <u>3.2</u> (Due Authorization, Execution and Delivery; Enforceability) and <u>3.4</u> (Capitalization of the Company; Title to Membership Interests), at any time after Closing, subject to the applicable statute of limitations; and

(ii) with respect to all other representations, warranties and covenants of Sellers contained in this Agreement, the Survival Period; *provided* that any claim for breach of the representations and warranties or covenants of Sellers contained in this Agreement involving fraud, gross negligence or intentional misconduct shall be subject to the applicable statute of limitations.

(c) Buyer shall not be required to indemnify or save harmless Seller Indemnified Parties pursuant to <u>Section 7.2</u>, unless the Administrative Agent shall have provided a Notice of Claim in respect thereof within the following time limits:

(i) with respect to the representations and warranties set forth in Sections 4.1 (Organization and Qualification), 4.3 (Issuance of Securities), 4.5 (Authorization; Enforcement) and 4.11 (No Misrepresentation), at any time after Closing, subject to the applicable statute of limitations; and

(ii) with respect to all other representations, warranties and covenants of Buyer contained in this Agreement, the Survival Period; *provided* that any claim for breach of the representations and warranties or covenants of Buyer contained in this Agreement involving fraud, gross negligence or intentional misconduct shall be subject to the applicable statute of limitations.

7.4 <u>Notice of Claims</u>. Notice of any Claim under this <u>Article 7</u> based on any inaccuracy or breach of a representation or warranty must be given prior to the expiration of the applicable survival period related to the representation and warranty as set forth in <u>Sections 7.3(b)</u> or <u>7.3(c)</u>, as applicable, and any Claim not made within such period shall be of no force or effect and shall not give rise to any obligation of the Indemnifying Party to indemnify or save harmless any Indemnified Party. Notwithstanding the foregoing, if, before the close of business on the last day of the applicable claims period set forth above, an Indemnifying Party shall have been notified in writing of a Claim for indemnity hereunder in accordance with the terms hereof, and such Claim shall not have been finally resolved or disposed of at such date, such Claim shall continue to survive and shall remain a basis for indemnity hereunder until such Claim is finally resolved or disposed of in accordance with the terms hereof.

7.5 <u>Certain Limitations.</u> The indemnification provided for herein shall be subject to the following limitations (the "Liability Limits"):

(a) The Buyer Indemnified Parties shall not be entitled to require payment in respect of any Loss pursuant to the indemnities contained in <u>Section 7.1</u> and Sellers shall not be liable for any indemnity payment thereunder unless:

(i) the amount finally agreed or adjudicated of any such individual Loss exceeds CDN\$200,000 (the "<u>De Minimis Amount</u>");

(ii) either alone or together with the amount finally agreed or adjudicated to be payable in respect of other Losses for which Buyer would otherwise be entitled to require payment under such indemnities, such Loss exceeds 2% of the total value of the Consideration Units determined as of the close of trading on the CSE on the Closing Date (the "<u>Deductible</u>"); *provided* that (A) any individual amount used to calculate the Deductible shall be no less than the De Minimis Amount, and (B) once the Deductible has been exceeded, Buyer shall only be entitled to require payment on such indemnities on the portion of Losses that exceeds the Deductible; and

(iii) notwithstanding any provision in this Agreement to the contrary, (A) the aggregate amount of all Losses for which Sellers shall be liable pursuant to <u>Section 7.1</u> shall not exceed 15% of the total value of the Consideration Units determined as of the close of trading on the CSE on the Closing Date.

(b) Subject to Buyer's obligations under the Indemnity Deed, notwithstanding any provision in this Agreement to the contrary, Buyer's aggregate liability to Seller Indemnified Parties pursuant to <u>Sections 7.2</u> shall in no event exceed the total value of the Consideration Units determined as of the close of trading on the CSE on the Closing Date.

(c) Losses arising out of separate sets of facts, matters or circumstances will not be treated as an individual Loss, even if each set of facts, matters or circumstances may be a breach of the same representation and warranty, and Losses of the same or similar nature arising out of the same or similar facts, matters and circumstances will be treated as an individual Loss.

(d) Payments by an Indemnifying Party in respect of any Loss shall be reduced by an amount equal to any Tax benefit realized, or reasonably expected to be realized, as a result of such Loss by the Indemnified Party.

(e) Each Indemnified Party shall take, and cause its Affiliates to take, all reasonable steps to mitigate any Loss upon becoming aware of any event or circumstance that would be reasonably expected to, or does, give rise thereto, including incurring costs only to the minimum extent necessary to remedy the breach that gives rise to such Loss.

(f) Where an Indemnified Party is, or would be likely to be, entitled to recover or be compensated or indemnified by another Person, whether by way of contract, indemnity or otherwise (including under a policy of insurance), any amount in respect of a Claim made by the Indemnified Party, the Indemnified Party shall promptly notify the Indemnifying Party of such right or entitlement, take all reasonable steps to seek recovery of that amount and keep the Indemnifying Party at all times fully and promptly notified of the status of such recovery. The amount of the Claim by the Indemnified Party shall be reduced by any amount actually recovered by the Indemnified Party (net of all reasonable out of pocket costs and expenses incurred in doing so and any Tax paid or payable on the amount recovered).

(g) If, after an Indemnifying Party has made a payment in respect of a Claim, an Indemnified Party recovers from or is paid by another Person any amount in respect of the Loss that gave rise to the Claim, the Indemnified Party shall promptly, and in any event within 10 Business Days, pay to the Indemnifying Party, the lesser of (i) the amount of the Loss that was recovered or paid and (ii) the amount paid by the Indemnifying Party to the Indemnified Party in respect of the Claim, in either case net of all reasonable out of pocket costs and expenses incurred in obtaining the recovery or payment and any Tax paid or payable as a result of receiving such recovery or payment.

#### 7.6 Notice of Claim.

(a) An Indemnified Party that may be entitled to make a Claim shall give written notification to the Indemnifying Party of such Claim (a "<u>Notice of Claim</u>") promptly upon becoming aware of the Claim, but in no event later than the relevant date, if any, specified in <u>Section 7.3(b)</u> or <u>7.3(c)</u>, as applicable. The Notice of Claim shall specify whether the Claim arises as a result of a Claim by a third party against the Indemnified Party (a "<u>Third Party Claim</u>") or whether the Claim does not so arise (a "<u>Direct Claim</u>"), and shall also specify with reasonable particularity, to the extent that the information is available, the factual basis for the Claim and the amount of the Claim.

(b) If an Indemnified Party fails to provide the Indemnifying Party with a Notice of Claim promptly as required by <u>Section 7.6(a)</u>, the Indemnifying Party shall be relieved of the obligation to pay damages to the extent it can show that it was prejudiced in its defence of the Claim or in proceeding against a third party who would have been liable to it but for the fact of the delay, but the failure to provide such Notice of Claim promptly shall not otherwise release the Indemnifying Party from its obligations under this <u>Article 7</u>.

(c) If the date by which a Notice of Claim must be given as set out in Section 7.3(b) or 7.3(c), as applicable, in respect of a breach of representation and warranty has passed without any Notice of Claim having been given to the Indemnifying Party, then the related Claim shall be forever extinguished, notwithstanding that by the date specified in Section 7.3(b) or 7.3(c), as applicable, the Indemnified Party did not know, and in the exercise of reasonable care could not have known, of the existence of the Claim.

7.7 <u>Direct Claims.</u> With respect to any Direct Claim, following receipt of notice from the Indemnified Party of the Claim, the Indemnifying Party shall have 45 days to make such investigation of the Claim as is considered necessary or desirable. For the purpose of such investigation, the Indemnified Party shall make available to the Indemnifying Party the information relied upon by the Indemnified Party to substantiate the Claim, together with all such other information as the Indemnifying Party may reasonably request. If both Parties agree at or prior to the expiration of such 45-day period (or any mutually agreed upon extension thereof) to the validity and amount of such Claim, the Indemnifying Party shall immediately pay to the Indemnified Party the full agreed-upon amount of the Claim (subject to the limitations set out in <u>Section 7.5</u>), failing which the Indemnified Party is free to pursue all rights and remedies available to it, subject to this Agreement.

### 7.8 Third Party Claims.

(a) Subject to <u>Section 7.8(d)</u>, upon receiving a Notice of Claim, the Indemnifying Party may participate in the investigation and defense of the Third Party Claim, and may also elect to assume the investigation and defense of the Third Party Claim with counsel satisfactory to the Indemnified Party, acting reasonably; *provided* that the Indemnifying Party shall not have the right to assume such investigation and defense, and shall pay the fees and expenses of counsel retained by the Indemnified Party, if the Third Party Claim involves a claim that, in the good faith judgment of the Indemnified Party, the Indemnifying Party failed or is failing to vigorously prosecute or defend. The Indemnified Party shall have the right, at its own cost and expense, to participate in the defense of any Third Party Claim with counsel selected by it subject to the Indemnifying Party's right to control the defense thereof.

(b) In order to assume the investigation and defense of a Third Party Claim, the Indemnifying Party must give the Indemnified Party written notice of its election within 20 days of the Indemnifying Party's receipt of the Notice of Claim.

Claim:

(c) Subject to Section 7.8(d), if the Indemnifying Party assumes the investigation and defense of a Third Party

(i) the Indemnifying Party will pay for all reasonable costs and expenses of the investigation and defense of the Third Party Claim except that the Indemnifying Party will not, so long as it diligently conducts such defense, be liable to the Indemnified Party for any fees of other counsel or any other expenses with respect to the defense of the Third Party Claim, incurred by the Indemnified Party after the date the Indemnifying Party validly exercised its right to assume the investigation and defense of the Third Party Claim;

(ii) the Indemnifying Party will reimburse the Indemnified Party for all reasonable costs and expenses incurred by the Indemnified Party in connection with the investigation and defense of the Third Party Claim prior to the date the Indemnifying Party validly exercised its right to assume the investigation and defense of the Third Party Claim; and

(iii) if the Indemnifying Party thereafter fails to defend the Third Party Claim within a reasonable time, the Indemnified Party shall be entitled to assume such defense and the Indemnifying Party shall be bound by the results obtained by the Indemnified Party with respect to the Third Party Claim.

(d) Where the named parties to any Third Party Claim include the Indemnified Party as well as the Indemnifying Party and the Indemnified Party determines in good faith, based on advice from its legal counsel, that joint representation would be inappropriate due to the actual or potential differing interests between them or there may be one or more legal defenses available to the Indemnified Party which are different from or in addition to those available to the Indemnifying Party, and the Indemnified Party notifies the Indemnifying Party in writing that it elects to retain separate counsel, the Indemnifying Party shall not have the right to assume the defense of such Third Party Claim on behalf of the Indemnified Party but shall be liable to pay the reasonable fees and expenses of counsel of the Indemnified Party. In no event, however, shall the Indemnifying Party be liable hereunder to pay the fees and disbursements of more than one counsel in any one jurisdiction acting as counsel on behalf of all Indemnified Parties.

(e) If the Indemnified Party undertakes the defense of the Third Party Claim, the Indemnifying Party will not be bound by any compromise or settlement of the Third Party Claim effected without the consent of the Indemnifying Party (which consent may not be unreasonably withheld, conditioned or delayed).

(f) The Indemnifying Party will not be permitted to compromise and settle or to cause a compromise and settlement of a Third Party Claim without the prior written consent of the Indemnified Party, which consent may not be unreasonably withheld, conditioned or delayed; *provided, however*, that no such consent shall be required if:

(i) the terms of the compromise and settlement require only the payment of money for which the Indemnified Party is entitled to full indemnification under this Agreement and the Indemnifying Party agrees to timely pay such amount in full; and

(ii) the Indemnified Party is not required to admit any wrongdoing, take or refrain from taking any action, acknowledge any rights of the Person making the Third Party Claim or waive any rights that the Indemnified Party may have against the Person making the Third Party Claim.

(g) No Party shall be liable to pay any amount in discharge of a Claim under this Agreement unless and until the liability in respect of which the Claim is made has become due and payable.

7.9 <u>Tax Treatment of Indemnification Payments</u>. All indemnification payments made under this Agreement shall be treated by the Parties as an adjustment to the value of the Consideration for Tax purposes, unless otherwise required by Law.

7.10 <u>Exclusive Remedies.</u> If the Closing occurs, no Party may make any Claim for damages in respect of this Agreement, or in respect of any breach hereof, against any other Party except by making a Claim pursuant to and in accordance with this <u>Article 7</u>. The provisions of this <u>Article 7</u> in its entirety shall survive the termination of this Agreement.

7.11 <u>No Liability of Administrative Agent</u>. The Administrative Agent shall not be liable to Buyer or any Buyer Indemnified Party for any actions or failures to act by the Administrative Agent in its capacity as Administrative Agent.

### ARTICLE 8 TERMINATION

8.1 Termination. This Agreement may be terminated at any time prior to the Closing:

(a) by the mutual written consent of Sellers and Buyer, any such consent of Sellers to be delivered by the Administrative Agent;

(b) by Buyer by written notice to Sellers if:

(i) the Closing has not occurred on or prior to the date that is 90 days from the date of this Agreement (or, if such date is not a Business Day, the next subsequent Business Day), or such later date as may be agreed in writing by the Parties (the "<u>Drop Dead Date</u>"), except that the right to terminate this Agreement under this <u>Section 8.1(b)(i)</u> shall not be available to Buyer if its failure to fulfill any of its covenants or obligations or if the breach of any of its representations and warranties under this Agreement has been the cause of, or resulted in, the failure of the Closing to occur by the Drop Dead Date; or

(ii) any of the conditions set forth in <u>Sections 6.1</u> or <u>6.2</u> shall not have been satisfied or waived by the Drop Dead Date or is incapable of satisfaction by the Drop Dead Date; *provided* that Buyer is not then in breach of this Agreement so as to cause any of the conditions in <u>Section 6.1</u> or <u>6.3</u> not to be satisfied; or

(c) by Sellers by written notice of the Administrative Agent to Buyer if:

(i) the Closing has not occurred on or prior to the Drop Dead Date, except that the right to terminate this Agreement under this <u>Section 8.1(c)(i)</u> shall not be available to Sellers if the failure of Sellers to fulfill any of Sellers' covenants or obligations or if the breach of any of the representations and warranties of Sellers under this Agreement has been the cause of, or resulted in, the failure of the Closing to occur by the Drop Dead Date; or

(ii) any of the conditions set forth in Section 6.1 or 6.3 shall not have been satisfied or waived by the Drop Dead Date or is incapable of satisfaction provided that Sellers are not then in breach of this Agreement so as to cause any of the conditions in Section 6.1 or 6.2 not to be satisfied.

### 8.2 Effect of Termination.

(a) Notwithstanding the termination of this Agreement by Buyer pursuant to <u>Section 8.1(b)</u>, Buyer may bring an action against Sellers for Losses suffered by any Buyer Indemnified Parties where the event giving rise to the right of termination is a result of a breach of covenant, representation or warranty by Sellers.

(b) Notwithstanding the termination of this Agreement by Sellers pursuant to <u>Section 8.1(c)</u>, Sellers may bring an action against Buyer for Losses suffered by any Seller Indemnified Parties where the event giving rise to the right of termination is a result of a breach of covenant, representation or warranty by Buyer.

(c) Notwithstanding any other provisions of this Agreement, if this Agreement is terminated, the provisions of Sections 1.1, 1.2, 2.5(a), and 5.4, this Section 8.2 and Article 7 and Article 9 (subject to any time limitations referred to therein) shall survive such termination and remain in full force and effect, along with any other provisions of this Agreement which expressly or by their nature survive the termination hereof.

#### ARTICLE 9 MISCELLANEOUS

9.1 <u>Expenses</u>. Except as otherwise expressly provided herein (including in <u>Sections 2.5(c)</u>, 2.5(d), 2.5(e), 2.8, 5.17 and 5.20), all costs and expenses, including, without limitation, fees and disbursements of counsel, financial advisors and accountants, incurred in connection with this Agreement and the Transactions shall be paid by the Party incurring such costs and expenses, whether or not the Closing shall have occurred.

9.2 <u>Notices</u>. All notices, requests, consents, claims, demands, waivers and other communications hereunder shall be in writing and shall be deemed to have been given: (a) when delivered by hand (with written confirmation of receipt), (b) when received by the addressee if sent by a nationally recognized overnight courier (receipt requested), (c) on the date sent by facsimile or e-mail of a PDF document (with confirmation of transmission) if sent during normal business hours of the recipient, and on the next Business Day if sent after normal business hours of the recipient, or (d) when received by the addressee if mailed, by certified or registered mail, return receipt requested, postage prepaid. Such communications must be sent to the respective Parties at the following addresses (or at such other address for a Party as shall be specified in a notice given in accordance with this <u>Section 9.2</u>):

If to Sellers:

Administrative Agent c/o Barrick Gold Corporation Brookfield Place TD Canada Trust Tower 161 Bay Street, Suite 3700 P.O. Box 212 Toronto, Canada M5J 2S1 Attn: General Counsel

Fax: 416-861-9717

with a copy to:	Barrick Gold of North America Inc. 310 S. Main Street, Suite 1150 Salt Lake City, Utah 84101 Attn: Michael McCarthy, General Counsel (North America)	
	Fax: 801-359-0875	
If to Buyer:	Bullfrog Gold Corp. 897 Quail Run Drive Grand Junction, Colorado 81505	

Attn: David Beling, CEO

with a copy to:

9.3 <u>Severability</u>. If any provision of this Agreement is determined by a court of competent jurisdiction to be invalid, illegal or unenforceable in any respect, all other provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the Transactions contemplated hereby is not affected in a manner materially adverse to a Party.

9.4 <u>Entire Agreement</u>. This Agreement, the Indemnity Deed, the Royalty Deed and the Investor Rights Agreement constitute the sole and entire agreement of the Parties with respect to the subject matter contained herein and supersede all prior and contemporaneous representations, warranties, understandings and agreements, both written and oral, with respect to such subject matter.

9.5 <u>Successors and Assigns</u>. This Agreement shall be binding upon and shall inure to the benefit of the Parties hereto and their respective successors and permitted assigns. No Party may assign its rights or obligations hereunder without the prior written consent of the other Parties, which consent shall not be unreasonably withheld, conditioned or delayed. No assignment shall relieve the assigning Party of any of its obligations hereunder.

9.6 <u>Beneficiaries</u>. This Agreement includes rights and benefits for the Indemnified Parties, and the Parties will exercise their respective rights and obligations under this Agreement with due consideration for the rights and benefits of the Indemnified Parties. Subject to the foregoing, this Agreement is for the sole benefit of the Parties and their respective successors and permitted assigns and nothing herein, express or implied, is intended to or shall confer upon any other Person any legal or equitable right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.

9.7 <u>Amendment and Modification; Waiver</u>. This Agreement may only be amended, modified or supplemented by an agreement in writing signed by each Party hereto or, in the case of Sellers, by the Administrative Agent. No waiver by any Party of any of the provisions hereof shall be effective unless explicitly set forth in writing and signed by the Party so waiving or, in the case of Sellers, by the Administrative Agent. No waiver by any Party shall operate or be construed as a waiver in respect of any failure, breach or default not expressly identified by such written waiver, whether of a similar or different character, and whether occurring before or after that waiver. No failure to exercise, or delay in exercising, any right, remedy, power or privilege arising from this Agreement shall operate or be construed as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege.

## 9.8 Governing Law; Submission to Jurisdiction.

(a) This Agreement shall be governed by and construed in accordance with the internal laws of the State of Nevada without giving effect to any choice or conflict of law provision or rule (whether of the State of Nevada or any other jurisdiction).

(b) Any Claim arising out of or based upon this Agreement may be instituted in the state courts of Nevada or the federal courts of the United States, in each case located in Reno, Nevada, and each Party irrevocably submits to the exclusive jurisdiction of such courts in any such Claim. Service of process, summons, notice or other document by mail to such Party's address set forth herein shall be effective service of process for any Claim brought in any such court. The Parties irrevocably and unconditionally waive any objection to the laying of venue of any Claim in such courts and irrevocably waive and agree not to plead or claim in any such court that any such Claim brought in any such court has been brought in an inconvenient forum.

(c) EACH PARTY HEREBY IRREVOCABLY WAIVES ALL RIGHTS TO A JURY TRIAL IN ANY ACTION, SUIT, OR PROCEEDING OF ANY KIND DIRECTLY OR INDIRECTLY ARISING OUT OF OR IN ANY WAY RELATING TO THIS AGREEMENT. THE JURY TRIAL WAIVER CONTAINED IN THIS AGREEMENT IS INTENDED TO APPLY, TO THE FULLEST EXTENT PERMITTED BY LAW, TO ANY AND ALL DISPUTES AND CONTROVERSIES THAT ARISE OUT OF OR IN ANY WAY RELATE TO ANY OR ALL OF THE MATTERS DESCRIBED IN THE PRECEDING SENTENCE, INCLUDING WITHOUT LIMITATION CONTRACT CLAIMS, TORT CLAIMS, AND ALL OTHER COMMON LAW AND STATUTORY CLAIMS OF ANY KIND. THIS AGREEMENT MAY BE FILED WITH ANY COURT OF COMPETENT JURISDICTION AS A PARTY'S WRITTEN CONSENT TO SUCH PARTY'S WAIVER OF A JURY TRIAL.

9.9 <u>Disclosure Schedule</u>. The information in the Disclosure Schedule constitutes exceptions or qualifications to representations and warranties of Sellers as set forth in this Agreement. Any disclosure made in the Disclosure Schedule shall be deemed to be disclosures made with respect to all representations and warranties contained in this Agreement to the extent reasonably apparent on their face, regardless of whether or not a specific cross-reference is made thereto. The Parties acknowledge and agree that the information and disclosures contained in the Disclosure Schedule do not constitute or imply, and will not be construed as: (a) any representation, warranty, covenant or agreement which is not expressly set out in the body of this Agreement; (b) an admission that the information is material or would constitute a Material Adverse Effect; or (c) an expansion of the scope or effect of any of the representations, warranties and/or covenants set out in this Agreement. No disclosure in the Disclosure Schedule relating to a possible breach or violation of any contract or Law shall be construed as an admission or indication that a breach or violation exists or has actually occurred. Capitalized terms used in the Disclosure Schedules that are not defined therein shall have the meaning given them in this Agreement.

9.10 <u>Counterparts</u>. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together shall be deemed to be one and the same agreement. A signed copy of this Agreement delivered by facsimile, e-mail or other means of electronic transmission shall be deemed to have the same legal effect as delivery of an original signed copy of this Agreement.

[Signature Page Follows]

The Parties hereto have caused this Agreement to be executed as of the Execution Date by their respective duly authorized Representatives.

# SELLERS

HOMESTAKE MINING COMPANY OF CALIFORNIA, a California corporation

By:	/s/ Michael R. McCarthy
Name:	Michael R. McCarthy
Title:	Director

LAC MINERALS (USA) LLC, a Delaware limited liability company

By:	/s/ Michael R. McCarthy
Name:	Michael R. McCarthy
Title:	Director
By:	/s/ Patrick Malone
	/s/ Patrick Malone Patrick Malone

### **BUYER**

# BULLFROG GOLD CORP., a Delaware corporation

By:	/s/ David Beling
Name:	David Beling
Title:	Chief Executive Officer and President



## Exhibit 10.1

#### **Execution Version**

### SUBSCRIPTION AGREEMENT FOR UNITS OF BULLFROG GOLD CORP.

This Subscription Agreement (this "<u>Agreement</u>") is being delivered to the purchaser identified on the signature page to this Agreement, being Augusta Investments Inc. ("<u>Augusta</u>" or the "<u>Subscriber</u>"), in connection with its investment in **Bullfrog Gold Corp.**, a Delaware corporation (the "<u>Company</u>"). The Company is conducting a private placement (the "<u>Offering</u>") of units ("<u>Units</u>"), as more fully described below, at a purchase price of C\$0.20 per Unit (the "<u>Purchase Price</u>"). Each Unit will consist of: (i) one (1) share of the Company's common stock (the "<u>Common Stock</u>"), par value US\$0.0001 per share (the "<u>Shares</u>"), and (ii) one (1) four (4) year warrant to purchase one additional share of the Company's common stock (a "<u>Warrant Share</u>") at a per share exercise price of C\$0.30 (or the prevailing equivalent United States dollar price calculated using the daily average exchange rate of the Bank of Canada on the Business Day (as defined below) prior to the date of exercise) (the "<u>Exercise Price</u>"), in the form attached hereto as <u>Exhibit D</u> (the "<u>Warrants</u>"). For purposes of this Agreement, the term "<u>Securities</u>" shall refer collectively to the Units, the Shares, the Warrants and the Warrant Shares or any of them, as the circumstances require. The aggregate gross proceeds to be raised from the Offering is C\$22,000,000.00.

#### RECITALS

- A. The Company entered into the binding term sheet dated September 7, 2020, a first amendment thereto on October 7, 2020 and a second amendment thereto on October 8, 2020, attached hereto as <u>Exhibit A</u> (as amended, the "<u>Term Sheet</u>"), that sets forth the principal understanding between Homestake Mining Company of California, a California corporation ("<u>Homestake</u>"), Lac Minerals (USA) LLC, a Delaware limited liability company ("<u>Lac Minerals</u>" and together with Homestake, the "<u>Barrick Parties</u>"), Augusta and the Company, with respect to an integrated transaction involving (i) the purchase by the Company from the Barrick Parties of all of the equity interests (the "<u>Equity Interests</u>") in Bullfrog Mines LLC, the successor by conversion of Barrick Bullfrog Inc. (the "<u>Target</u>"), a Delaware corporation (the "<u>Acquisition Transaction</u>"), and (ii) the concurrent financing pursuant to which Augusta and/or its affiliates and certain other individuals identified by Augusta (together, the "<u>Augusta Group</u>"), as subscribers, will acquire Units pursuant to the Offering for cash consideration under the terms of this Agreement;
- B. Pursuant to the Term Sheet, the Company and the Barrick Parties entered into a Membership Interest Purchase Agreement related to the Acquisition Transaction as of the date hereof, in substantially the form attached hereto as Exhibit B (the "Purchase Agreement");
- C. On the Closing Date (as defined below) (i) the Company and the Target, as applicable, and the Barrick Parties will enter into the Royalty Deed (as defined in the Term Sheet) and the "<u>Indemnity Deed</u>" (referred to in the Term Sheet as the Operating and Indemnity Agreement); and (ii) the Company, the Barrick Parties or an affiliate thereof and Augusta will enter into an Investor Rights Agreement, in the form attached hereto as Exhibit C (the "<u>Investor Rights Agreement</u>");
- D. The Target's sole assets will be the mining claims described in Exhibit A to the Term Sheet (the "<u>Target Mining</u> <u>Claims</u>"), certain rights-of-way described in Exhibit A to the Term Sheet (the "<u>Target Rights-of-Way</u>"), the current permits described in Exhibit B to the Term Sheet (the "<u>Target Permits</u>"), and the water rights held by the Target relating to the Mining Claims described in Exhibit C to the Term Sheet (the "<u>Target Water Rights</u>"); and
- E. The concurrent closing of the Acquisition Transaction is a condition of closing of the Offering.

### 1. SUBSCRIPTION AND PURCHASE PRICE

(a) Offering. The Company will raise C\$22,000,000 on the Closing Date.

(b) <u>Subscription</u>. Subject to the conditions in its favour set forth in Sections 6(a) and (b) hereof, the Subscriber hereby subscribes for and agrees to purchase the number of Units indicated on page 19 hereof on the terms and conditions described herein.

(c) <u>Purchase of Units</u>. The Subscriber understands and acknowledges that the purchase price to be remitted to the Company in exchange for the Units shall be C\$0.20 per Unit for an aggregate purchase price as set forth on page 19 hereof (the "<u>Aggregate Purchase Price</u>"). The Subscriber shall effect payment for the Units subscribed for hereunder at Closing, in Canadian dollars by wire transfer of immediately available funds in accordance with the wire instructions provided in <u>Exhibit E</u>. The Subscriber understands and agrees that, subject to Section 6 and applicable laws, by executing this Agreement, it is entering into a binding agreement.

## 2. ACCEPTANCE, OFFERING TERM AND CLOSING PROCEDURES

(a) <u>Acceptance</u>. Subject to full, faithful and punctual performance and discharge by the Company of all of its duties, obligations and responsibilities as set forth in this Agreement, the Subscriber shall be legally bound to purchase the Units pursuant to the terms and conditions set forth in this Agreement. In the event the Closing (as defined below) does not take place because of the conditions in this Agreement for the benefit of a party are not satisfied or waived by the Outside Date (as defined below), this Agreement shall thereafter be terminated and have no force or effect, and the parties shall take all steps to ensure that the Aggregate Purchase Price submitted to the Company's bank account as specified in <u>Exhibit E</u> herein shall promptly be returned or caused to be returned to the Subscriber without interest thereon or deduction therefrom in accordance with the wire transfer instructions provided to the Company by the Subscriber for such purpose.

(b) <u>Closing</u>. The closing of the purchase and sale of the Units hereunder (the "<u>Closing</u>") shall take place at the offices of Sichenzia Ross Ference LLP in New York, or such other place as determined by the Company, concurrent with the closing of the Acquisition Transaction. The Closing shall take place beginning at 8:00 a.m. (Eastern Time) on the third Business Day after the last of the conditions to Closing set forth in Section 6 below have been satisfied or waived (other than conditions which, by their nature, are to be satisfied on the Closing Date) or such other time and date as may be determined by the Company (the "<u>Closing</u> <u>Date</u>"). "<u>Business Day</u>" shall mean from the hours of 9:00 a.m. (Eastern Time) through 5:00 p.m. (Eastern Time) of a day other than a Saturday, Sunday or other day on which commercial banks located in Vancouver, British Columbia, Toronto, Ontario or Denver, Colorado are authorized or required to be closed. The Units purchased by the Subscriber will be delivered by the Company on the Closing Date. In lieu of a physical closing, the Company and the Subscriber agree that the Closing may take place on the Closing Date on the exchange of solicitors' undertakings which will involve solicitors for each of the Company and Augusta delivering to his or her counterpart all required documentation and payments, to be held in escrow and not released until all such documentation has been executed and delivered and all conditions have been satisfied or waived and solicitors for each of the Company and Augusta has authorized in writing that the escrow is to be terminated.

(c) <u>Following Acceptance</u>. The Subscriber acknowledges and agrees that this Agreement and any other documents delivered in connection herewith will be held by the Company. If the Closing does not occur, the documents delivered in connection herewith will be returned to the Subscriber at the address of the Subscriber as set forth in this Agreement.

### 3. THE SUBSCRIBER'S REPRESENTATIONS, WARRANTIES AND COVENANTS

The Subscriber hereby acknowledges, agrees with and represents, warrants and covenants to the Company, as follows:

(a) The Subscriber has full power and authority to enter into this Agreement, the execution and delivery of which has been duly authorized, if applicable, and this Agreement constitutes a valid and legally binding obligation of the Subscriber, except as may be limited by bankruptcy, reorganization, insolvency, moratorium and similar laws of general application relating to or affecting the enforcement of rights of creditors, and except as enforceability of the obligations hereunder are subject to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or law).

<u>- 2</u> -

(b) The Subscriber acknowledges its understanding that the Offering is intended to be exempt from registration under the Securities Act of 1933, as amended (the "Securities Act"), by virtue of Section 4(a)(2) of the Securities Act and/or the provisions of Regulation D promulgated thereunder ("Regulation D"). In furtherance thereof, the Subscriber represents and warrants to the Company and its affiliates as follows:

(i) The Subscriber realizes that the basis for the exemption from registration may not be available if, notwithstanding the Subscriber's representations contained herein, the Subscriber is merely acquiring the Units for a fixed or determinable period in the future, or for a market rise, or for sale if the market does not rise. The Subscriber does not have any such intention.

(ii) The Subscriber realizes that the basis for the exemption would not be available if the Offering is part of a plan or scheme to evade registration provisions of the Securities Act or any applicable state or federal securities laws.

(iii) The Subscriber is acquiring the Units solely for the Subscriber's own beneficial account, for investment purposes, and not with a view towards, or resale in connection with, any distribution of the Units.

(iv) The Subscriber has the financial ability to bear the economic risk of the Subscriber's investment, has adequate means for providing for its current needs and contingencies, and has no need for liquidity with respect to an investment in the Company.

(v) The Subscriber and the Subscriber's attorney, accountant, purchaser representative and/or tax advisor, if any (collectively, the "<u>Advisors</u>") has such knowledge and experience in financial and business matters as to be capable of evaluating the merits and risks of a prospective investment in the Units. If other than an individual, the Subscriber also represents it has not been organized solely for the purpose of acquiring the Units.

(vi) The Subscriber (together with its Advisors, if any) has received all documents requested by the Subscriber, if any, has carefully reviewed them and understands the information contained therein, prior to the execution of this Agreement.

(c) The Subscriber is not relying on the Company or any of its employees, agents, sub-agents or advisors with respect to the legal, tax, economic and related considerations involved in this investment. The Subscriber has relied on the advice of, or has consulted with, only its Advisors.

(d) The Subscriber has carefully considered the potential risks relating to the Company and a purchase of the Units, and fully understands that the Units are a speculative investment that involves a high degree of risk of loss of the Subscriber's entire investment. Among other things, the Subscriber has carefully considered each of the risks described under the heading "*Risk Factors*" in the Company's SEC Filings, which risk factors are incorporated herein by reference, and any additional disclosures in the nature of risk factors described herein.

(e) The Subscriber will not sell or otherwise transfer any Securities without registration under the Securities Act or an exemption therefrom, and fully understands and agrees that the Subscriber must bear the economic risk of its purchase because, among other reasons, the Securities have not been registered under the Securities Act or under the securities laws of any state and, therefore, cannot be resold, pledged, assigned or otherwise disposed of unless they are subsequently registered under the Securities Act and under the applicable securities laws of such states, or an exemption from such registration is available. In particular, the Subscriber is aware that the Securities are "restricted securities," as such term is defined in Rule 144 promulgated under the Securities Act ("<u>Rule 144</u>"), and they may not be sold pursuant to Rule 144 unless all of the conditions of Rule 144 are met. The Subscriber understands that any sales or transfers of the Securities are further restricted by state securities laws and the provisions of this Agreement.

3 -

(f) The Subscriber understands and agrees that the certificates for the Securities shall bear substantially the following legend until (i) such Securities shall have been registered under the Securities Act and effectively disposed of in accordance with a registration statement that has been declared effective or (ii) in the opinion of counsel acceptable to the Company, such Securities may be sold without registration under the Securities Act, as well as any applicable "blue sky" or state securities laws:

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), OR ANY APPLICABLE STATE SECURITIES LAWS. SUCH SECURITIES HAVE BEEN ACQUIRED FOR INVESTMENT PURPOSES AND MAY NOT BE OFFERED FOR SALE, SOLD, DELIVERED AFTER SALE, TRANSFERRED, PLEDGED OR HYPOTHECATED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT FILED BY THE COMPANY WITH THE U.S. SECURITIES AND EXCHANGE COMMISSION COVERING SUCH SECURITIES UNDER THE SECURITIES ACT OR AN OPINION OF COUNSEL THAT SUCH REGISTRATION IS NOT REQUIRED.

THE PRESENCE OF THIS LEGEND MAY IMPAIR THE ABILITY OF THE HOLDER HEREOF TO EFFECT "GOOD DELIVERY" OF THE SECURITIES REPRESENTED HEREBY ON A CANADIAN STOCK EXCHANGE."

(g) The Subscriber acknowledges that if the Subscriber is resident in Canada the certificates representing the Securities will bear, as of the Closing Date, a legend substantially in the following form and with the necessary information inserted:

"UNLESS PERMITTED UNDER SECURITIES LEGISLATION, THE HOLDER OF THIS SECURITY MUST NOT TRADE THE SECURITY BEFORE THE DATE THAT IS 4 MONTHS AND A DAY AFTER THE CLOSING DATE."

(h) Neither the SEC, the Canadian Securities Exchange, nor any U.S. or Canadian securities commission has approved the Securities or passed upon or endorsed the merits of the Offering. There is no government or other insurance covering any of the Securities.

(i) The Subscriber has taken no action that would give rise to any claim by any person for brokerage commissions, finders' fees or the like relating to this Agreement or the transactions contemplated hereby.

(j) The Subscriber acknowledges that any estimates or forward-looking statements or projections furnished by the Company to the Subscriber were prepared by the management of the Company in good faith, but that the attainment of any such projections, estimates or forward-looking statements cannot be guaranteed by the Company or its management and should not be relied upon.

(k) If the Subscriber is resident in Canada, the Company has advised the Subscriber that the Company is relying on an exemption from the requirements to provide the Subscriber with a prospectus and to sell the Units through a person registered to sell securities under the *Securities Act* (British Columbia) and, as a consequence of acquiring the Units pursuant to these exemptions, certain protections, rights and remedies provided by the *Securities Act* (British Columbia), including statutory rights of rescission or damages, will not be available to the Subscriber.

(1) The Subscriber is, and on each date on which the Subscriber acquires Units under the Offering will be an "Accredited Investor" (i) if the Subscriber is resident in the United States, as defined in Rule 501(a) under the Securities Act; or (ii) if the Subscriber is resident in Canada, (A) under National Instrument 45-106 – *Prospectus Exemptions* ("<u>NI 45-106</u>"), or (B) the *Securities Act* (Ontario). If the Subscriber is resident in Canada, the Subscriber has properly completed, executed and delivered to the Company this Agreement and Schedule A to this Agreement and the acknowledgments, representations, warranties, covenants and information contained herein and therein are true and correct as of the date hereof and will be true and correct as of the Closing Date.

(m) The execution and delivery of this Agreement, the performance and compliance with the terms hereof, the subscription for the Units and the completion of the transactions contemplated hereby will not result in any breach of, or be in conflict with or constitute a default under, or create a state of facts which, after notice or lapse of time, or both, would constitute a default under any term or provision of the constating documents, by-laws or resolutions of the Subscriber (if not an individual), applicable securities laws or any other applicable law, any agreement to evaluate which the Subscriber is a party or any applicable regulation, judgment, decree, order or ruling to which the Subscriber is bound.

- 4 -

(n) The Subscriber acknowledges that it is bound by the provisions of applicable securities laws which impose obligations on a person who becomes an Insider (as defined in the *Securities Act* (British Columbia)) of an issuer, or on a person who holds sufficient securities exercisable into voting securities of an issuer to become an Insider. The Subscriber acknowledges that such obligations may include, but are not necessarily limited to: the filing of insider reports on the System for Electronic Disclosure by Insiders (SEDI); the filing of early warning reports; the filing of reports of acquisitions; and the filing of a Personal Information Form or similar document with the applicable stock exchange. The Subscriber further acknowledges that it has been advised to consult its own legal advisors with respect to such obligations, and that it is solely responsible for complying with such obligations, and covenants and agrees with the Company that it will comply with all of such obligations, if applicable to the Subscriber, in a timely manner, whether arising at or after the Closing.

(o) The funds representing the Aggregate Purchase Price which will be advanced by the Subscriber hereunder will not represent proceeds of crime for the purposes of the *Proceeds of Crime (Money Laundering) and Terrorist Financing Act* (Canada), as may be amended from time to time (the "<u>PCMLTFA</u>") and for the purposes of the *United States Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act*, as may be amended from time to time (the "<u>PATRIOT Act</u>") and the Subscriber acknowledges that the Company may in the future be required by law to disclose the Subscriber's name and other information relating to this Agreement and the Subscriber's subscription hereunder, on a confidential basis, pursuant to the PCMLTFA and the PATRIOT Act. To the best of its knowledge, none of the funds representing the Aggregate Purchase Price to be provided by the Subscriber: (i) have been or will be derived from or related to any activity that is deemed criminal under the laws of Canada, the United States or any other jurisdiction; or (ii) are being tendered on behalf of a person or entity who has not been identified to the Subscriber. The Subscriber shall promptly notify the Company if it discovers that any of such representations cease to be true and provide the Company with appropriate information in connection therewith.

(p) The Subscriber is not engaged in the business of trading in securities or exchange contracts as a principal or agent and does not hold himself, herself or itself out as engaging in the business of trading in securities or exchange contracts as a principal or agent, or is otherwise exempt from any requirements to be registered as a dealer under National Instrument 31-103 - Registration Requirements, Exemptions and Ongoing Registrant Obligations.

(q) The Subscriber is not a Control Person (as defined in the Securities Act (British Columbia)) of the Company.

(r) If the Subscriber is resident in Canada, this Agreement and Schedule A attached hereto require the Subscriber to provide certain personal information relating to the Subscriber to the Company. The Subscriber acknowledges that such information is being collected and will be used by the Company for the purposes of completing the Offering, which includes, without limitation, determining the Subscriber's eligibility to purchase the Units under Canadian securities laws, preparing and registering certificates representing securities or arranging for non-certificated, electronic delivery of same, and completing filings required by any securities regulatory authority or exchange. Such personal information may be disclosed by the Company to (a) securities regulatory authorities and commissions, or stock exchanges, (b) the Company's registrar and transfer agent, (c) any government agency (including any taxing authorities), board or other entity, and (d) any of the other parties involved in the Offering, including the legal counsel of the Company, and may be included in record books in connection with the Offering. By executing this Agreement, the Subscriber consents to the foregoing collection, use and disclosure of such personal information. The Subscriber also consents to the filing of copies or originals of any of the Subscriber's documents described herein as may be required to be filed with any securities regulatory authority in connection with the transactions contemplated hereby.

- 5 -

(s) If the Subscriber is resident or otherwise subject to the applicable securities legislation of a jurisdiction in Canada, the Subscriber acknowledges that: (i) the Company will deliver to the applicable securities regulatory authority or regulator certain personal information pertaining to the Subscriber, including such Subscriber's full name, residential address and telephone number, email address, the number of Units purchased by such Subscriber, the Aggregate Purchase Price paid for such Units, the prospectus exemption relied on and the date of distribution of the Units; (ii) such information is being collected indirectly by the applicable securities regulatory authority or regulator under the authority granted to it in securities legislation; (iii) such information is being collected for the purposes of the administration and enforcement of the securities legislation of the local Canadian jurisdiction; and (iv) the Subscriber may contact the public officials listed on Schedule B with respect to questions about the security regulatory authority's or regulator's indirect collection of such information.

### 4. THE COMPANY'S REPRESENTATIONS, WARRANTIES AND COVENANTS

The Company hereby acknowledges, agrees with and represents, warrants and covenants to the Subscriber, on the date of this Agreement and on the Closing Date, as follows:

(a) Organization and Qualification. The Company is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. The Company is duly qualified to do business and is in good standing in the states required due to: (i) the ownership or lease of real or personal property for use in the operation of the Company's business; or (ii) the nature of the business conducted by the Company. The Company has all requisite power, right and authority to own, operate and lease its properties and assets, to carry on its business as now conducted, to execute, deliver and perform its obligations under this Agreement, the Purchase Agreement, the Investor Rights Agreement, the Royalty Deed, the Indemnity Deed and any other agreement entered into among the Subscriber (or any member of the Augusta Group), the Company, the Barrick Parties and any of their respective affiliates, or any of them, at or prior to Closing (the "Transaction Documents") to which it is a party, and to carry out the transactions contemplated hereby and thereby, including the power and authority to create, issue and deliver the Units to be issued to the Subscriber pursuant to this Agreement, and the Warrant Shares to be issued upon due exercise of the Warrants in accordance with their terms. All actions on the part of the Company and its officers and directors necessary for the authorization, execution, delivery and performance of the Transaction Documents, the consummation of the transactions contemplated hereby and thereby, and the performance of all of the Company's obligations under the Transaction Documents, including the creation, issuance and delivery of the Units to be issued to the Subscriber pursuant to this Agreement, and the Warrant Shares to be issued upon due exercise of the Warrants in accordance with their terms, have been taken or will be taken prior to the Closing. This Agreement has been, and the other Transaction Documents to which the Company is a party on the Closing will be, duly executed and delivered by the Company, and this Agreement is, and each of the other Transaction Documents to which it is a party on the Closing will be, a legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, except as enforceability of the obligations hereunder and thereunder are subject to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or law).

(b) <u>No Conflicts</u>. The execution, delivery and performance by the Company of the Transaction Documents to which it is a party, and the consummation of the transactions contemplated hereby and thereby do not: (i) conflict with or result in a violation or breach of any provision of the constating documents of the Company, or (ii) violate in any material respect any law applicable to the Company.

(c) <u>Subsidiaries</u>. The only subsidiaries of the Company are Standard Gold Corp. and Rocky Mountain Minerals Corp. (each a "<u>Subsidiary</u>" and, collectively, the "<u>Subsidiaries</u>"). Each Subsidiary is a corporation duly organized, validly existing and in good standing under the laws of the State of Nevada. Each Subsidiary is duly qualified to do business and is in good standing in the states required due to: (i) the ownership or lease of real or personal property for use in the operation of its business; or (ii) the nature of the business conducted by such Subsidiary. Each Subsidiary has all requisite power, right and authority to own, operate and lease its properties and assets, to carry on its business as now conducted, to execute, deliver and perform its obligations under each agreement to which it is a party, and to carry out the transactions contemplated thereby. The Company legally and beneficially owns all of the equity interests in the Subsidiaries, free and clear of any hypothecations, mortgages, pledges, assignments, liens, charges, security interests, encumbrances and adverse rights, other third person interest or encumbrance of any kind, whether contingent or absolute, and any agreement, option, right or privilege capable of becoming any of the foregoing (collectively, "Liens"). There are no options, warrants, conversion privileges or other rights, shareholder rights plans, agreements, arrangements or commitments (pre-emptive, contingent or otherwise) of any character whatsoever requiring or which may require the issuance, sale or transfer of any securities of any Subsidiary, or any securities or obligations convertible into, or exchangeable or exercisable for, or otherwise evidencing a right or obligation to acquire, any securities of either Subsidiary. All outstanding equity interests in each Subsidiary have been duly authorized and validly issued, and are fully paid and non-assessable.

- 6 -

(d) <u>Issuance of Securities</u>. The Securities to be issued to the Subscriber pursuant to this Agreement, and the Warrant Shares to be issued upon due exercise of the Warrants in accordance with their terms, when issued and delivered in accordance with the terms of this Agreement, will be duly and validly issued and will be fully paid and non-assessable.

(e) Exemption from Prospectus Requirements. The issuance of the Securities will be exempt from the prospectus and registration requirements of applicable securities laws and no document will be required to be filed and no proceeding taken or approval, permit, consent, order or authorization obtained under applicable securities laws in Canada in connection with the first trade of the Securities; provided that: (i) the Company is and has been a reporting issuer in a jurisdiction in Canada for the four (4) months immediately preceding the trade; (ii) at least four (4) months have elapsed from the Closing Date; (iii) the certificates representing the Shares and the Warrants carry a legend stating that "Unless permitted under securities legislation, the holder of this security must not trade the security before [insert date that is 4 months and a day after the Closing Date]"; (iv) such trade is not a control distribution (as such term is defined in National Instrument 45-102 – Resale Restrictions); (v) no unusual effort is made to prepare the market or to create a demand for the Securities that are subject of the trade; and (vi) no extraordinary commission or consideration is paid to a person in respect of the trade.

(f) Authorization; Enforcement. The Offering, the execution, delivery and performance of this Agreement and the other Transaction Documents by the Company, and the consummation of the transactions contemplated hereby and thereby, including the creation, issuance and delivery of the Units to be issued to the Subscriber pursuant to this Agreement, and the Warrant Shares to be issued upon due exercise of the Warrants in accordance with their terms, will not: (i) constitute a violation (with or without the giving of notice or lapse of time, or both) of any provision of any law or any judgment, decree, order, regulation or rule of any court, agency or other governmental authority applicable to the Company; (ii) require any consent, approval or authorization of, or declaration, filing or registration with, any person (except as may be required under applicable securities laws) except (A) such as have already been obtained, (B) notification of a change of control to be provided by the Company following the Closing to the OTC Markets Group, Inc. and (C) any acceptance of the Canadian Securities Exchange of the Offering, if necessary, which will be obtained by the Closing Date; (iii) result in a default (with or without the giving of notice or lapse of time, or both) under, acceleration or termination of, or the creation in any party of the right to accelerate, terminate, modify or cancel, any agreement, lease, note or other restriction, encumbrance, obligation or liability to which the Company is a party or by which it is bound or to which any assets of the Company are subject; (iv) result in the creation of any lien or encumbrance upon the assets of the Company, or upon any shares of Common Stock or other securities of the Company; (v) conflict with or result in a breach of or constitute a default under any provision of the certificate of incorporation or by-laws of the Company; (vi) invalidate or adversely affect any permit, license, authorization or status used in the conduct of the business of the Company; or (vii) restrict, hinder, impair or limit the ability of the Company to carry on its business in any material respect as and where it is now being carried on.

(g) <u>SEC Filings</u>. The Company is subject to, and in material compliance with, the reporting requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, as amended (the "<u>Exchange Act</u>"). The Company has made available to each Subscriber through the EDGAR system true and complete copies of each of the Company's Quarterly Reports on Form 10-Q, Annual Reports on Form 10-K and Current Reports on Form 8-K, in each case filed since January 1, 2018 (collectively, the "<u>SEC Filings</u>"). The SEC Filings, when they were filed with the SEC (or, if any amendment with respect to any such document was filed, when such amendment was filed), complied in all material respects with the applicable requirements of the Exchange Act and the rules and regulations thereunder and did not, as of such date, contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading. The Company and each of the Subsidiaries, are engaged in all material respects only in the business described in the SEC Filings, and the SEC Filings contain a complete and accurate description in all material respects of the business of the Company and the SUBSIDIARY subsidiaries.

(h) <u>Reporting Issuer Status</u>. The Company is a reporting issuer in the provinces of British Columbia and Ontario, is not in default of any of the requirements of the applicable securities laws in such provinces and its name does not appear on the lists of defaulting reporting issuers maintained by the securities regulatory authorities in those provinces.

- 7 -

(i) <u>Listing Status</u>. All of the issued and outstanding shares of the Common Stock are listed and posted for trading on the Canadian Securities Exchange and no order, ruling or determination having the effect of ceasing or suspending trading of any securities of the Company or prohibiting from trading of the Company's issued securities has been issued and no proceedings for such purpose have been instituted or, to the knowledge of the Company, are pending or contemplated or threatened.

(j) <u>No Cease Trade</u>. The shares of the Common Stock are not subject to or affected by any actual or, to the knowledge of the Company, pending or threatened cease trade orders of, compliance or denial of use of exemption orders of, or action, investigation or proceeding by or before, any securities regulatory authority, court, administrative agency or other tribunal.

(k) <u>Continuous Disclosure Documents</u>. Since January 1, 2018, the Company has filed with all applicable securities regulatory authorities all forms, reports and documents required to be filed by it pursuant to all applicable laws, instruments, and policies, and all such filings, when made, complied in all material respects with the then applicable laws, instruments and policies. Such documents are accurate as at the date thereof, in all material respects, and may be relied upon by the Subscriber in consummating the transactions contemplated hereby; and in particular, without limiting the foregoing, the Company has made timely disclosure of all material changes relating to it and no such disclosure has been made on a confidential basis and there is no material change relating to the Company which has occurred with respect to which the requisite material change report has not been filed.

(1) <u>No Misrepresentation</u>. The documents publicly disclosed by the Company do not contain any untrue statement of a material fact as at the date thereof nor do they omit to state a material fact which, at the date thereof, was required to have been stated or was necessary to prevent a statement that was made from being false or misleading in the circumstances in which it was made and were prepared in accordance with and complied with applicable securities laws.

(m) <u>Compliance with Applicable Corporate and Securities Laws.</u> The Company has complied and will comply in all material respects with the requirements of all applicable corporate and securities laws and administrative policies and directions, including, without limitation, all applicable laws with respect to the sale of the Units.

(n) <u>No Financial Advisor</u>. The Company acknowledges and agrees that the Subscriber is acting solely in the capacity of an arm's length purchaser with respect to the Units and the transactions contemplated hereby. The Company further acknowledges that the Subscriber is not acting as a financial advisor or fiduciary of the Company (or in any similar capacity) with respect to this Agreement and the transactions contemplated hereby and any advice given by the Subscriber or any of its representatives or agents in connection with this Agreement and the transactions contemplated hereby is merely incidental to the Subscriber's purchase of the Units. The Company further represents to the Subscriber that the Company's decision to enter into this Agreement has been based solely on the independent evaluation of the transactions contemplated hereby by the Company and its representatives.

(o) <u>Indemnification</u>. The Company will indemnify and hold harmless the Subscriber and, where applicable, its directors, officers, employees, agents, advisors and shareholders, from and against any and all Loss arising out of or based upon any representation or warranty of the Company contained herein or in any document furnished by the Company to the Subscriber in connection herewith being untrue in any material respect or any breach or failure by the Company to comply with any covenant or agreement made by the Company to the Subscriber in connection therewith; <u>provided</u>, <u>however</u>, that the Company's liability shall not exceed the Subscriber's Aggregate Purchase Price tendered hereunder.

(p) <u>Capitalization and Additional Issuances</u>. The authorized and outstanding capital stock of the Company on a fully diluted basis are as set forth in the SEC Filings as of the dates set forth therein. As of the date of this Agreement, 161,230,237 shares of Common Stock and 19,229,166 shares of preferred stock series B (the "<u>Preferred Stock</u>") are issued and are outstanding as fully paid and non-assessable; and no shares of preferred stock series A are outstanding; all of the shares of Common Stock and Preferred Stock have been issued in compliance with applicable laws, including applicable securities laws and the policies of the Canadian Securities Exchange. Other than pursuant to the Offering and as set forth in <u>Exhibit F</u> hereto, no person has any agreement or option or right or privilege (whether at law, pre-emptive or contractual) capable of becoming an agreement for the purchase, subscription or issuance of, or conversion into, any unissued shares, securities, warrants or convertible obligations of any nature of the Company. The Units, upon issuance, will not be issued in violation of or subject to any pre-emptive rights, participation rights or other contractual rights to purchase securities issued by the Company. The only officer, director, employee and consultant stock option or stock incentive plan or similar plan currently in effect is described in the SEC Filings.

- 8 -

(q) <u>Shareholder Rights Plan.</u> The Company has not entered into a shareholder rights plan agreement or put in place a shareholder rights plan.

(r) <u>Transfer Agent</u>. National Securities Administrators Ltd. has been duly appointed as the transfer agent and registrar for the issued and outstanding shares of Common Stock in the United States and Canada.

(s) <u>Private Placements</u>. Assuming the accuracy of the Subscriber's representations and warranties set forth in Section 3, no registration under the Securities Act is required for the offer and sale of the Units by the Company to the Subscribers as contemplated hereby.

(t) <u>Investment Company</u>. The Company is not, and is not an affiliate of, and immediately after receipt of payment for the Units will not be an affiliate of, an "investment company" within the meaning of the Investment Company Act of 1940, as amended. The Company shall conduct its business in a manner so that it will not become subject to the Investment Company Act of 1940, as amended.

(u) <u>Bankruptcy, Insolvency and Reorganization</u>. None of the Company or any Subsidiary is an insolvent person within the meaning of the *Bankruptcy and Insolvency Act* (Canada) or under Title 11 United States Code Section 101, as applicable, and none of the Company or any Subsidiary has made an assignment in favour of its creditors nor a proposal in bankruptcy to its creditors or any class thereof nor had any petition for a receiving order presented in respect of it nor undertaken a demerger or the transfer of its statutory seat or effective place of its management. None of the Company or any Subsidiary has initiated proceedings with respect to a compromise or arrangement with its creditors or for its winding-up, liquidation or dissolution or a demerger or the transfer of its statutory seat or effective place of its management. No receiver is currently appointed in respect of the Company or any Subsidiary, or their respective property or assets and no execution or distress is currently levied upon any of the property or assets of the Company or any Subsidiary. No act or proceeding has been taken or authorized by or against the Company or any Subsidiary with respect to any amalgamation, merger, consolidation, arrangement or reorganization of, or relating to the Company or any Subsidiary nor have any such proceedings been authorized by any other person.

(v) <u>Material Contracts</u>. The material contracts of the Company and the Subsidiaries, to which the Company or any Subsidiary is a party or to which they, or any of their respective properties or assets, may be subject or by which the Company or any Subsidiary is bound as of the Closing Date are: (i) the Transaction Documents; and (ii) otherwise described in the SEC Filings (the "<u>Material Contracts</u>"). All Material Contracts are in full force and effect (other than the Transaction Documents to be entered into on Closing, which are in full force and effect on Closing only), and the Company or a Subsidiary, as applicable, is entitled to all rights and benefits thereunder in accordance with the terms thereof. All of the Material Contracts are valid and binding obligations of the Company or a Subsidiary, as applicable, (other than the Transaction Documents to be entered into on Closing, which are valid and binding on Closing only) enforceable in accordance with their respective terms, except as may be limited by bankruptcy, insolvency and other laws affecting the enforcement of creditors' rights generally and subject to the qualification that equitable remedies may only be granted in the discretion of a court of competent jurisdiction and there exists no default by the Company or event of default or event, occurrence, condition or act, which, with the giving of notice, the lapse of time or the happening of any other event or condition, would become a default or event of default by the Company under any Material Contract and, to the knowledge of the Company, all covenants to be performed by any other party to such agreements have been fully performed in all material respects and the Company has not received notification from any party that it is in breach of or default under any Material Contracts.

- 9 -

### (w) Financial Statements; Internal Controls.

(i) The Company's audited consolidated financial statements included in the Company's Form 10-K for the year ended December 31, 2019 and the unaudited consolidated financial statements included in the Company's quarterly reports on Form 10-Q for the quarters ended March 31, 2020 and June 30, 2020, filed in the SEC Filings (the "<u>Financial Statements</u>") were prepared in accordance with generally accepted accounting principles in the United States applied on a consistent basis ("<u>GAAP</u>") and Public Company Accounting Oversight Board ("<u>PCAOB</u>") standards and present fairly in all material respects the financial condition and position of the Company, taken as a whole and including all consolidation adjustments and other GAAP adjustments, as at the dates thereof and the results of operations and changes in financial statements, normal and recurring year-end adjustments that will not, individually or in the aggregate, be material, and are in all material respects complete and correct, and do not contain or reflect any material inaccuracies or discrepancies. There has been no material change in the accounting policies of the Company since the date of the last Financial Statement.

(ii) The books of account and other financial records of the Company: (A) reflect all items of income and expense and all assets and liabilities required to be reflected therein in accordance with local tax and accounting (excluding any purchase price allocation entries and other GAAP adjustments) regulations and GAAP applied on a consistent basis in accordance with the past practices of the Company; and (B) have been maintained in accordance with good business practices, GAAP and PCAOB requirements.

(iii) The Company maintains systems of internal accounting controls sufficient to provide reasonable assurance that: (A) transactions are executed in accordance with management's general or specific authorizations; (B) transactions are recorded as necessary to permit preparation of consolidated financial statements for the group of entities in conformity with GAAP; and (C) access to assets is permitted only in accordance with management's general or specific authorization.

(iv) Each SEC Filing containing financial statements was accompanied by the certifications required to be filed or submitted by the Company's chief executive officer and chief financial officer pursuant to the Sarbanes-Oxley Act of 2002 (the "<u>Sarbanes-Oxley Act</u>"); at the time of filing or submission of each such certification, such certification was true and accurate and complied with the Sarbanes-Oxley Act and the rules and regulations promulgated thereunder; such certifications contain no qualifications or exceptions to the matters certified therein and have not been modified or withdrawn; and neither the Company nor any of its officers has received notice from any governmental entity questioning or challenging the accuracy, completeness, form or manner of filing or submission of such certification.

(v) Save and except as disclosed in the Financial Statements, the Company and each of its affiliates does not have any loans, notes or other indebtedness outstanding to any of its shareholders, officers, directors or employees, past or present, or any person not dealing at "arm's length" (as such term is used in the *Income Tax Act* (Canada)).

(vi) The auditors of the Company are independent public accountants as required by applicable securities legislation, including Article 2 of Regulation S-X, and there has not been any "reportable event" (within the meaning of National Instrument 51-102 – *Continuous Disclosure Obligations* of the Canadian Securities Administrators) or required disclosure under Item 4.01 of Form 8-K, that has not otherwise been reported, with the present or former auditor of the Company and the auditor of the Company has not provided any material comments or recommendations to the Company regarding its accounting policies, internal control systems or other accounting or financial practices that have not been implemented by the Company.

#### (x) Tax Representations.

(i) The Company and each Subsidiary have duly and timely filed (or there have been filed on their behalf) with the appropriate governmental entity or authority all tax returns (the "<u>Returns</u>") required to be filed by or on their behalf for all periods ending prior to the date hereof. All such Returns are complete and correct in all material respects and no material fact has been omitted therefrom. No extension of time in which to file any such Returns, and all taxes required to be paid by the Company and each Subsidiary, have been paid in full. The Company has made a provision for taxes in the Financial Statements, which constitute an adequate provision for the payment of all taxes.

(ii) No governmental entity or authority has asserted, or is threatening to assert, against the Company or any Subsidiary any deficiency or claim for taxes, and no governmental entity or authority has challenged or disputed a filing position taken by the Company or any Subsidiary in any Return.

#### - 10 -

(iii) There are currently no tax actions, suits, proceedings, audits, investigations or claims now existing, pending or threatened against the Company or any Subsidiary. No tax reassessments in respect of the Company or any Subsidiary have been issued and are outstanding, and there are no outstanding issues or matters in respect of taxes which have been raised and communicated to the Company or any Subsidiary by any governmental entity or authority. Neither the Company nor any Subsidiary has executed or filed with any governmental entity or authority any agreement or waiver extending the period for assessment, reassessment, payment or collection of any taxes, or otherwise waived any statute of limitations in respect of taxes.

(iv) No lien for taxes has been filed or exists in respect of the Company or any Subsidiary other than for taxes not yet due and payable.

### (y) Employment Matters.

(i) Except for certain indebtedness of the Company to David Beling, the Company's Chief Executive Officer, in the amount of US\$654,000 as at June 30, 2020 and a severance payment to David Beling of US\$200,000 triggered on Closing, neither the Company nor any Subsidiary is a party to or bound by any oral or written contract or commitment providing for: (A) severance, notice of termination or pay in lieu of notice of termination or termination, severance, retention or similar payments; or (B) cash or other compensation or benefits to any director, officer, employee, consultant or agent upon or as a result of the execution of the Purchase Agreement or the consummation of the transactions contemplated in the Acquisition Transaction.

(ii) The Company is in compliance, in all material respects, with the provisions of all federal, state, provincial, municipal, local and foreign laws and regulations respecting employment and employment practices, terms and conditions of employment and wages and hours.

(iii) The Company does not have, and has never had, employee plans, other than the Company's 2011 and 2017 Equity Incentive Plans.

### (z) Environmental Matters.

(i) The Company and each Subsidiary: (A) is in material compliance with all applicable federal, state, provincial, municipal, local and foreign laws and regulations relating to the protection of human health and safety, the environmental or hazardous or toxic substances or wastes, pollutants or contaminants (collectively, the "<u>Environmental Laws</u>"); (B) has received all necessary permits, licenses or other approvals required of it under applicable Environmental Laws to conduct its business as presently conducted (collectively, the "<u>Environmental Permits</u>"); (C) is in material compliance with all terms and conditions of each Environmental Permit; (D) confirms that here have been no past, and there are no current claims, complaints, notices or requests for information received by the Company or any Subsidiary with respect to any alleged violation of any Environmental Laws that, individually or in the aggregate, has or may reasonably be expected to have, a material adverse effect on the business, operators or financial condition of the Company or any Subsidiary; and (E) to the knowledge of the Company after due inquiry, there are no events or circumstances that would reasonably be expected to form the basis of an order for clean-up or remediation, or an action, suit or proceeding by any other Person or governmental body or agency, against or affecting the Company or any Subsidiary relating to Environmental Laws.

(ii) Except in compliance with Environmental Permits and Environmental Laws, neither the Company nor any Subsidiary has used or permitted to be used any of its assets or facilities, whether owned, leased, occupied, controlled or licensed or which it owned, leased, occupied, controlled or licensed at any prior time, to generate, manufacture, process, distribute, use, treat, store, dispose of, transport or handle any hazardous waste, material or substance as defined in or pursuant to any Environmental Laws by which the Company is bound or subject.

- 11 -

### (aa) Litigation Matters.

(i) There are no actions, suits, proceedings, inquiries or investigations existing, pending or, to the knowledge of the Company after due inquiry, threatened or adversely affecting the Company or any Subsidiary or to which any of their property or assets is subject, at law or in equity, or before or by any court, federal, provincial, state, municipal or other governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, and none of the Company nor any Subsidiary are subject to any judgment, order, writ, injunction, decree or award of any governmental authority.

(ii) The Company is not aware of any pending change or contemplated change to any applicable law or regulation or governmental position that would or may reasonably be expected to have a material adverse effect on the business, operators or financial condition of the Company or any Subsidiary.

(iii) Neither the Company nor any Subsidiary, nor any other person associated with or acting on behalf of the Company or any Subsidiary (including, without limitation, any director, officer, employee or agent of the Company or any Subsidiary) has, directly or indirectly, while acting on behalf of the Company or any Subsidiary: (A) used any corporate funds for unlawful contributions, gifts or other unlawful expenses relating to political activity; (B) made any unlawful payment to foreign or domestic governmental officials or employees or to foreign or domestic political parties or campaigns from corporate funds; or (C) violated or alleged to have violated any provisions of the *Foreign Corrupt Practices Act of 1977* (as amended, supplemented, modified, restated or replaced from time to time) and the rules and regulations thereunder, the *Corruption of Foreign Public Officials Act* (Canada) or any applicable law of similar effect in a jurisdiction.

(iv) The operations of the Company and each Subsidiary are and have been conducted at all times in compliance with applicable financial recordkeeping and reporting requirements of the anti-money laundering statutes of all applicable jurisdictions, the rules and regulations promulgated thereunder and any related or similar rules, regulations, guidelines issued, administered or enforced by each applicable governmental agency (collectively, the "<u>Anti-Money</u> <u>Laundering Laws</u>") and no action, suit or proceeding by or before any applicable court of governmental agency, authority or body or any arbitrator involving the Company or any Subsidiary with respect to the Anti-Money Laundering Laws is pending or, to the knowledge of the Company, threatened.

(v) Neither the Company, any Subsidiary nor, to the knowledge of the Company, any director, officer, agent, employee, affiliates or other person acting on behalf of the Company or any Subsidiary is currently the target of any U.S. sanctions administered by the Officer of Foreign Assets Control of the U.S. Treasury Department ("<u>OFAC</u>") and the Company will not directly or indirectly use the proceeds of the Offering or lend, contribute or otherwise make available such proceeds to any subsidiary, joint venture partner or other person or entity, for the purpose of financing the activities of any person that is currently the target of any U.S. sanctions administered by OFAC.

(vi) No property or assets of the Company or any Subsidiary has been taken or expropriated by any governmental authority nor has any notice or proceeding in respect thereof been given or commenced nor, to the knowledge of the Company, is there any intent or proposal to give any such notice or to commence any such proceeding.

(bb) <u>Material Properties</u>. The Company's only interest in a mineral project on a property material to the Company is the Bullfrog Gold Project (the "<u>Bullfrog Project</u>") located in Nye County, Nevada which, for greater certainty, shall be deemed not to include the Target Mining Claims, the Target Rights-of-Way, the Target Permits or the Target Water Rights until immediately following Closing.

(cc) <u>Scientific and Technical Information</u>. The Company is in compliance with the provisions of National Instrument 43-101 – *Standards of Disclosure for Mineral Projects* ("<u>NI 43-101</u>") and has filed all technical reports in respect of its properties (and properties in respect of which it has a right to earn an interest) required thereby. The technical report of the Company entitled "NI 43-101 Technical Report, Mineral Resource Estimate, Bullfrog Gold Project, Nye County, Nevada" dated effective August 9, 2017 (the "<u>Technical Report</u>") remains current as of the date hereof. The Technical Report complies in all material respects with the requirements of NI 43-101 and there is no new scientific or technical information concerning the Bullfrog Project since the date thereof that would require a new technical report in respect of such properties, respectively, to be issued under NI 43-101. The Company and each Subsidiary has made available to the author of the Technical Report, prior to the issuance thereof, for the purpose of preparing such report, all information requested by the author and none of such information contained any misrepresentation at the time such information was provided. The information publicly disclosed by the Company relating to scientific and technical information has been prepared in accordance with NI 43-101 and in compliance with all applicable securities laws.

- 12 -

### (dd) Bullfrog Project.

(i) The Company and/or each Subsidiary has sufficient title, clear of any material title defect or encumbrance, to the Bullfrog Project and has good and sufficient title to all real property interests including, without limitation, fee simple estate of and in real property, leases, easements, rights of way, surface use agreements, permits or licenses from landowners or authorities permitting the use of land by the Company and/or each Subsidiary necessary to permit the operation of their businesses as presently owned and conducted. The Company and/or each Subsidiary holds all mineral rights, permits and water rights required to continue their businesses and operations as currently conducted and the Company and/or each Subsidiary has not received any notice, whether written or oral, from any person with jurisdiction or applicable authority of any revocation or intention to revoke or cancel the Company's or any Subsidiary's interest in any material mining rights, permits or water rights nor has the Company or any Subsidiary received any compliance orders, citations or notices relating to non-compliance or alleged non-compliance, relating to the mining rights, permits or water rights are free and clear of encumbrances and royalty burdens other than as described in the documents publicly disclosed by the Company. The Company and/or each Subsidiary, as applicable, has the exclusive right to deal with mineral rights held by the Company and/or each Subsidiary, the permits and the water rights.

(ii) The mineral rights, permits and water rights related to the Bullfrog Project which are held by the Company and/or Subsidiaries are in good standing and have been properly located and recorded in compliance with applicable laws.

(iii) All assessment or other work required to be performed in relation to the Bullfrog Project in order to maintain the Company's and/or each Subsidiary's interests therein, if any, have been performed to date and the Company and each Subsidiary has complied with all applicable governmental laws, regulations and policies in connection therewith.

(iv) Any and all taxes and other payments required to be paid in respect of the Bullfrog Project have been paid and any and all filings required to be filed in respect of the Bullfrog Project have been filed.

(v) Other than the Company and/or each Subsidiary, no other person has any interest in the Bullfrog Project or any right to acquire any such interest and there are no back-in rights, earn-in rights, rights of first refusal, royalty rights or similar provisions other than as disclosed by the Company in the SEC Filings which would affect the Company's or any Subsidiary's interests in the Bullfrog Project.

(vi) There are no mining concessions, claims, interests in real and immoveable property, leases, licenses, permits or any other rights conflicting with the Bullfrog Project.

(vii) Neither the Company nor any Subsidiary has received any notice, whether written or oral from any governmental entity or any person with jurisdiction or applicable authority of any default under, or of any revocation or limitation, or intention to revoke or limit, any mineral rights, permits or water rights held by the Company and/or each Subsidiary in connection with the Bullfrog Project.

(viii) The Company and/or each Subsidiary possess all material licenses, permits, certificates, consents, orders, grants, approvals, classifications, program participation requirements, sign-offs, registrations, or other authorizations of and from any governmental entity or authority (collectively, the "Bullfrog Permits") necessary to carry on the business currently carried on, or contemplated to be carried on by them. The Bullfrog Permits are in compliance in all material respects with the terms and conditions of all the Bullfrog Permits and with all laws, regulations, tariffs, rules, orders and directives related to the operations thereof. Neither the Company nor any Subsidiary has received any written or other notice of the modification, revocation, limitation or cancellation of, or any intention to modify, revoke, limit or cancel, or any proceedings relating to the modification, revocation, limitation or cancellation of any of the Bullfrog Permits. The Company and each Subsidiary are in compliance in all material respects with the terms and conditions of all of the Bullfrog Permits and with all laws, regulations of all of the Bullfrog Permits and with all laws, regulations, tariffs, rules, orders and directives related to the operations thereof.

- 13 -

(ee) <u>COVID-19</u>. Except as mandated by, or in conformity with the recommendations of a governmental authority, there has been no closure, demobilization, shut-down, suspension, postponement or disruption to any of the Company's or any Subsidiary's properties or to the operations and workforce productivity of the Company and each Subsidiary as a result of the novel coronavirus outbreak.

(ff) <u>No Brokerage Fee</u>. There is no person acting or purporting to act on behalf of the Company who is entitled to any brokerage, agency or other fiscal advisory or similar fee in connection with the transactions contemplated herein, other than Fort Capital Partners which is entitled to a cash payment of C\$1.2 million on Closing.

(gg) <u>Minute Books</u>. The minute books of the Company and each Subsidiary contains copies of all constating documents and all proceedings of securityholders and directors (and committees thereof) and are completed in all material respects.

(hh) <u>No Agreements Affecting Voting Control</u>. Other than the Investor Rights Agreement to be entered into on Closing, no agreement is in force or effect which will in any manner affect the voting or control of any of the securities of the Company.

(ii) <u>Insurance</u>. The assets of the Company and each Subsidiary and their businesses and operations are insured against loss or damage with responsible insurer on a basis consistent with insurance obtained by reasonably prudent participants in comparable businesses, and such coverage is in full force and effect, and neither the Company nor any Subsidiary has breached the terms of any policies in respect thereof or failed to promptly give any notice or present any material claim thereunder.

(jj) <u>Full, True and Complete Disclosure</u>. The Company has made available to the Subscriber and its advisors, or publicly disclosed, all of the information relating to the Company that the Subscriber has requested for deciding whether to complete the transactions contemplated in this Agreement and the Term Sheet; none of the foregoing representations, warranties and statements of fact and no other statement furnished by or on behalf of the Company to the Subscriber or its advisors in connection with the negotiation of the transactions contemplated by this Agreement and the Term Sheet, contain any untrue statement of a material fact or omit to state any material fact necessary to make such statement or representation not misleading under the circumstances in which such statement or representation is made.

## 5. OTHER AGREEMENTS OF THE PARTIES

(a) <u>Extraordinary Events Regarding Common Stock</u>. The Company shall not, prior to the Closing, (i) issue additional shares of Common Stock as a dividend or other distribution on outstanding Common Stock, (ii) subdivide its outstanding shares of Common Stock, (iii) combine its outstanding shares of Common Stock into a smaller number of shares of Common Stock, (iv) increase or decrease the number of shares of Common Stock authorized for issuance, (v) redeem, repurchase or otherwise acquire or offer to redeem, repurchase or otherwise acquire any of its outstanding securities, or (vi) amend or agree to amend the terms of any of its outstanding securities. For the avoidance of any doubt, the Company may issue shares of Common Stock from its treasury upon the conversion or exercise of any options, warrants or convertible Preferred Stock outstanding on the date hereof.

- 14 -

(b) Furnishing of Information. As long as any Subscriber owns Securities, the Company covenants to use commercially reasonable efforts to timely file (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the date hereof pursuant to the Exchange Act. As long as any Subscriber owns Securities, if the Company is not required to file reports pursuant to the Exchange Act, it will use commercially reasonable efforts to prepare and furnish to the Subscribers and make publicly available in accordance with Rule 144(c) under the Securities Act such information as is required for the Subscribers to sell the Securities under Rule 144. The Company further covenants that it will take such further action as any holder of Securities may reasonably request, all to the extent required from time to time to enable such person to sell such Securities without registration under Rule 144 under the Securities Act, provided such Securities are then eligible to be sold under Rule 144, provided further, that the Subscriber provides any information reasonably requested by the Company which for the avoidance of doubt may include a broker's representation letter that there is an intent to sell such Securities. Notwithstanding the foregoing, the Company agrees to timely take all reasonable action(s) necessary to have the Securities Act restrictive legends removed from Securities following any valid request for legend removal under Rule 144 by the Subscriber or its broker, including, without limitation, (i) authorizing the transfer agent to remove the restrictive legend, (ii) expediting the acquisition of a legal opinion from the Company's authorized counsel at the Company's expense, and (iii) delivering any additional documentation that may be reasonably required by the Subscriber, its broker or the transfer agent in connection with the legend removal request, including Rule 144 share representation letters and a resolution of the board of directors of the Company evidencing proper issuance of the Securities, as soon as reasonably possible; provided that, the Subscriber will provide any documentation or information reasonably requested by the Company, the Company's counsel, or the transfer agent in connection with such request.

(c) <u>Securities Laws Disclosure</u>; <u>Publicity</u>. The Company shall not publicly disclose the name of any Subscriber, or include the name of any Subscriber in any filing with the SEC or any regulatory agency, without the prior written consent of such Subscriber, except to the extent such disclosure is required by law, provided that, for the avoidance of doubt, this provision will not restrict the right of the Company to file a complete copy of this Agreement with the SEC or the securities regulatory authority in any Canadian jurisdiction.

(d) <u>Integration</u>. The Company shall not, and shall use its best efforts to ensure that no affiliate of the Company shall, after the date hereof, sell, offer for sale or solicit offers to buy or otherwise negotiate in respect of any security that would be integrated with the offer or sale of the Units in a manner that would require the registration under the Securities Act of the sale of the Units to the Subscribers.

(e) <u>Reservation of Securities</u>. The Company will use commercially reasonable efforts to maintain sufficient duly authorized shares of Common Stock for issuance pursuant to the Transaction Documents in such amount as may be required to fulfill its obligations in full under the Transaction Documents. In the event that at any time the then authorized shares of Common Stock are insufficient for the Company to satisfy its obligations in full under the Transaction Documents, the Company shall promptly use commercially reasonable efforts to take such actions as may be required to increase the number of authorized shares, including without limitation promptly preparing and filing any proxy or information statement with the SEC for purposes of a special meeting (or action by written consent, as applicable) to authorize the issuance of additional shares to Subscribers, which in no event shall be filed more than ten (10) business days following the date on which the Company becomes aware that it may have inadequate authorized capital.

(f) <u>Use of Proceeds</u>. The Company covenants and agrees that the proceeds of the Offering will be used for exploration and other work on the Bullfrog Project, including the Target Claims, and for working capital and general corporate purposes.

(g) <u>Maintain Corporate Existence</u>. As long as any Subscriber owns Securities, the Company shall use commercially reasonable efforts to maintain its existence as a corporation validly subsisting under the laws of its jurisdiction of existence, licensed, registered or qualified as an extra-provincial or foreign corporation in all jurisdictions where the character of its properties owned or leased or the nature of the activities conducted by it make such licensing, registration or qualification necessary and shall carry on its business in the ordinary course and in compliance in all material respects with all applicable laws, rules and regulations of each such jurisdiction.

(h) <u>Maintain Reporting Issuer Status</u>. As long as any Subscriber owns Securities, the Company shall use commercially reasonable efforts to maintain its status as a reporting issuer under the applicable securities laws in the provinces of British Columbia and Ontario not in default of any material requirement of such applicable securities laws provided that this covenant shall not prevent the Company from completing any transaction which would result in the Company ceasing to be a "reporting issuer" so long as the holders of shares of Common Stock receive securities of an entity which is listed on a stock exchange in Canada or the U.S. or cash, or a combination of both, or the holders of the shares of Common Stock have approved the transaction in accordance with the requirements of applicable securities laws and corporate laws.

- 15 -

(i) <u>Listing Status</u>. The Company shall ensure that the shares of Common Stock comprising part of the Units (including the Warrant Shares) are approved for listing on the Canadian Securities Exchange as of the Closing Date. As long as any Subscriber owns Securities, the Company shall not take any action which would reasonably be expected to result in the delisting or suspension of the shares of Common Stock on or from the Canadian Securities Exchange or on or from any securities exchange, market, or trading or quotation facility on which the shares of Common Stock are then listed or quoted, provided that this covenant shall not prevent the Company from completing any transaction which would result in the Company ceasing to be listed on the Canadian Securities Exchange in Canada or the U.S. or cash or the holders of the shares of Common Stock have approved the transaction in accordance with the requirements of applicable securities laws and corporate laws.

(j) <u>Maintain Quotation</u>. As long as any Subscriber owns Securities, the Company shall use commercially reasonable efforts to maintain eligibility for the shares of Common Stock on the OTCQB or a national securities exchange in the U.S.

(k) <u>DTC Eligibility</u>. For as long as any Subscriber owns Securities, the Company shall use its best efforts to maintain full eligibility of the Company's Common Stock for electronic clearance and settlement services through the Depository Trust Company.

(1) <u>Schedule 14F</u>. The Company shall prepare, file and mail to shareholders a Schedule 14F in accordance with Rule 14f-1 under the Exchange Act ("<u>Rule 14f-1</u>") as soon as practicable following receipt from Augusta of the information relating to the directors and officers to be appointed by Augusta required by the Company to comply with its obligations under Rule 14f-1 (the "<u>Required Information</u>") in satisfaction of the condition to Closing contained in Section 6(b)(iv).

(m) <u>Provision of Information for Schedule 14F</u>. Augusta shall provide the Required Information to the Company within three Business Days of the date of this Agreement.

## 6. CONDITIONS TO CLOSING

(a) <u>Mutual Conditions</u>. The obligations of the Company and the Subscriber to complete the sale and purchase of the Units is conditional on the satisfaction of all of the following conditions, each of which is for the benefit of the Company and the Subscriber and may only be waived by the mutual written consent of the Company and the Subscriber:

(i) no governmental authority shall have issued an order restraining or enjoining the transactions contemplated by the Transaction Documents and there shall not have been enacted or made applicable any law that makes such transactions illegal or otherwise prohibited;

(ii) the Company shall have completed all necessary filings with the Canadian Securities Exchange and received any necessary approvals from the Canadian Securities Exchange to allow for the issuance and listing of the Shares and the Warrants Shares on or prior to the Closing Date subject only to ordinary conditions typical for the issuance and listing of common shares;

(iii) the Closing having occurred on or prior to the date that is 90 days from the date of this Agreement (or, if such date is not a Business Day, the next subsequent Business Day), or such later date as may be agreed in writing by the Company and Augusta (the "<u>Outside Date</u>") provided that, with respect to any such extension of the Outside Date by the Company and Augusta, the Canadian Securities Exchange has confirmed in writing, to the extent requested by Augusta, that the Shares and the Warrants to be issued pursuant to this Agreement can be issued on the terms contemplated herein at Closing;

(iv) the transactions contemplated by the Purchase Agreement shall have been completed on the terms set forth therein at or contemporaneously with the Closing; provided that, to the extent that the Purchase Agreement is proposed to be amended by the parties thereto, Augusta shall have been provided with notice of the proposed amendments, including a copy of the proposed amending document, prior to the institution thereof and the Company shall not have entered into or otherwise agreed to any such amendment without the prior written consent of Augusta; and

- 16 -

(v) the Investor Rights Agreement shall have been executed and delivered by the Barrick Parties or an affiliate thereof.

(b) <u>Conditions for the Benefit of the Subscriber</u>. The obligations of the Subscriber to complete the purchase of the Units is conditional on the satisfaction of all of the following conditions, each of which is for the benefit of the Subscriber and may only be waived in writing by the Subscriber:

(i) the representations and warranties made by the Company in Article 4 hereof shall be true and correct in all respects without regard to any materiality or the material adverse effect qualifications contained in them, except where the failure of such representations and warranties to be true and correct do not constitute in aggregate a material adverse effect, in each case, as of the Closing Date as if made on and as of such date (except to the extent that such representations and warranties speak to an earlier date, then as of such date);

(ii) the Company shall have duly performed and complied in all material respects with all agreements, covenants and conditions required by the Transaction Documents to which it is a party to be performed or complied with by the Company prior to or on the Closing Date;

(iii) the Investor Rights Agreement shall have been executed and delivered by the Company; and

(iv) the Board of Directors and Management of the Company being reconstituted on the Closing Date in accordance with the governance changes described in the Term Sheet, it being acknowledged that the Company will be required to file and mail an information statement on Schedule 14F in accordance with Rule 14f-1 at least 10 days prior to the Board of Directors being reconstituted in such manner.

(c) <u>Conditions for the Benefit of the Company</u>. The obligations of the Company to complete the sale of the Units is conditional on the satisfaction of all of the following conditions, each of which is for the benefit of the Company and may only be waived by the Company in writing:

(i) the representations and warranties made by the Subscriber in Article 3 hereof shall be true and correct in all respects without regard to any materiality or the material adverse effect qualifications contained in them, except where the failure of such representations and warranties to be true and correct do not constitute in aggregate a material adverse effect, in each case, as of the Closing Date as if made on and as of such date (except to the extent that such representations and warranties speak to an earlier date, then as of such date, or except as affected by the transactions contemplated or permitted by the Transaction Documents);

(ii) the Subscriber shall have duly performed and complied in all material respects with all agreements, covenants and conditions required by this Agreement to be performed or complied with by it prior to or on the Closing Date; and

(iii) the Investor Rights Agreement shall have been executed and delivered by Augusta.

#### 7. MISCELLANEOUS PROVISIONS

(a) The parties hereto have been represented by counsel, and no inference shall be drawn in favor of or against any party by virtue of the fact that such party's counsel was or was not the principal draftsman of this Agreement.

(b) Each of the parties hereto shall be responsible to pay the costs and expenses of its own legal counsel in connection with the preparation and review of this Agreement and related documentation.

(c) Except as otherwise explicitly contemplated in this Agreement, neither this Agreement nor any of the provisions hereof, shall be waived, modified, discharged or terminated except by an instrument in writing signed by the party against whom any waiver, modification, discharge or termination is sought.

- 17 -

(d) The representations, warranties, covenants and agreements of the Subscriber and the Company made in this Agreement shall survive the execution and delivery of this Agreement and the delivery of the Units.

(e) This Agreement and the Term Sheet constitute the sole and entire agreement of the parties with respect to the subject matter contained herein and supersede all prior and contemporaneous representations, warranties, understandings and agreements, both written and oral, with respect to such subject matter. The Subscriber acknowledges and agrees to abide by the confidentiality provisions in the Term Sheet under the heading "General Provisions – Confidentiality" as if an original party thereto.

(f) Either party may send any notice, request, demand, claim or other communication hereunder to the Subscriber at the address set forth on the signature page of this Agreement or to the Company at its primary office (including personal delivery, expedited courier, messenger service, fax, ordinary mail or electronic mail), but no such notice, request, demand, claim or other communication will be deemed to have been duly given unless and until it actually is received by the intended recipient. Either party may change the address to which notices, requests, demands, claims and other communications hereunder are to be delivered by giving the other party written notice in the manner herein set forth.

(g) Except as otherwise provided herein, this Agreement shall be binding upon, and inure to the benefit of, the parties to this Agreement and their heirs, executors, administrators, successors, legal representatives and assigns. If the Subscriber is more than one person or entity, the obligation of the Subscriber shall be joint and several and the agreements, representations, warranties and acknowledgments contained herein shall be deemed to be made by, and be binding upon, each such person or entity and its heirs, executors, administrators, successors, legal representatives and assigns. This Agreement sets forth the entire agreement and understanding between the parties as to the subject matter hereof and merges and supersedes all prior discussions, agreements and understandings of any and every nature between them.

(h) This Agreement is not transferable or assignable by the Subscriber or the Company.

(i) This Agreement shall be governed by and construed in accordance with the laws of the Province of Ontario and the federal laws applicable therein.

(j) Each of the Company and the Subscriber irrevocably and unconditionally: (i) submits to the non-exclusive jurisdiction of the courts of the Province of Ontario over any action or proceeding arising out of or relating to this Agreement, (ii) waives any objection that it might otherwise be entitled to assert to the jurisdiction of such courts; and (iii) agrees not to assert that such courts are not a convenient forum for the determination of any such action or proceeding.

(k) This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

- 18 -

### ALL SUBSCRIBERS MUST COMPLETE THIS PAGE AS WELL AS THE PERSONAL INFORMATION AND INVESTOR QUESTIONNAIRE FOLLOWING THE ACCEPTANCE PAGE

## TO: BULLFROG GOLD CORP.

Units subscribed for

1. The Subscriber irrevocably subscribes for and agrees to purchase from the Company the following securities:

x C\$0.20 for each Unit =

Aggregate Purchase Price

2. The Subscriber and the Company agree that the offering of the Units shall be on the terms and conditions specified in this Agreement (including the schedules attached hereto). The Subscriber hereby makes the representations, warranties, acknowledgments and agreements set out in Schedule A hereto and in all applicable Forms, and acknowledges and agrees that the Company and its counsel will and can rely on such representations, warranties, acknowledgments and agreements should this subscription be accepted.

3. Identity of and execution by Subscriber:

BOX A: SUBSCRIBER INFORMATION AND EXECUTION				
(name of subscriber)				
(address – include city, pr	ovince and postal code)			
(telephone number)	(email address)	X (signature of subscriber/authorized signatory)		
		(if applicable, print name of signatory and office)		

Execution hereof by the Subscriber shall constitute an offer and agreement to subscribe for the Units set out in Item 1 above pursuant to the provisions of Item 2 above, and acceptance by the Company shall effect a legal, valid and binding agreement between the Company and the Subscriber. This subscription may be executed and delivered by facsimile or other electronic means and shall be deemed to bear the date of acceptance below.

4. If the Units are to be registered other than as set out in Box A, the Subscriber directs the Company to register and deliver the Units as follows:

# BOX B: ALTERNATE REGISTRATION INSTRUCTIONS

(name of registered holder)

(address of registered holder – include city, province and postal code)

(registered holder: contact name, contact telephone number and contact email address)

5. If the Units are to be delivered other than as set out in Box A (or if completed, Box B):

# BOX C: ALTERNATE DELIVERY INSTRUCTIONS

(name of recipient)

(address of recipient – include city, province and postal code)

(recipient: contact name, contact telephone number and contact email address)

- 19 -

# ACCEPTANCE

) )))

)

This subscription is accepted and agreed to by the Company

as of the \_\_\_\_\_ day of \_\_\_\_\_, 2020.

BULLFROG GOLD CORP.

Per:

# Authorized Signatory

#### **PERSONAL INFORMATION**

#### 1. <u>Present Ownership of Securities of the Company</u>. The Subscriber either [check appropriate box]:

- [] owns, directly or indirectly, or exercises control or direction over, no shares of the Company's common stock or securities convertible into shares of the Company's common stock; or
- [] owns, directly or indirectly, or exercises control or direction over, \_\_\_\_\_\_ shares of the Company's common stock and securities convertible or exercisable to acquire an additional \_\_\_\_\_\_ shares of the Company's common stock.

#### 2. <u>Insider Status</u>. The Subscriber either [check appropriate box]:

- [] is an "Insider" of the Company as defined in the Securities Act (British Columbia), determined as follows:
  - (a) a director or officer of the Company;
  - (b) a director or officer of a person that is itself an insider or subsidiary of the Company; or
  - (c) a person that has direct or indirect:
    - (1) beneficial ownership of or control or direction over; or
    - (2) a combination of beneficial ownership of and control or direction over,

securities of the Company carrying more than 10% of the voting rights attached to all the Company's outstanding voting securities, excluding, for the purpose of the calculation of the percentage held, any securities held by the person as underwriter in the course of a distribution.

[] is not an Insider of the Company.

#### 3. <u>Registrants</u>. The Subscriber either [check appropriate box]:

- [] is a dealer, adviser, investment fund manager, an ultimate designated person or chief compliance officer as those terms are used pursuant to applicable Canadian securities laws, or a person registered or otherwise required to be registered under applicable Canadian securities laws; or
- [] is not a Registrant



#### INVESTOR QUESTIONNAIRE

Instructions: Check all boxes below which correctly describe you.

- You are (i) a bank, as defined in Section 3(a)(2) of the Securities Act of 1933, as amended (the "Securities Act"), (ii) [] a savings and loan association or other institution, as defined in Section 3(a)(5)(A) of the Securities Act, whether acting in an individual or fiduciary capacity, (iii) a broker or dealer registered pursuant to Section 15 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), (iv) an insurance company as defined in Section 2(13) of the Securities Act, (v) an investment company registered under the Investment Company Act of 1940, as amended (the "Investment Company Act"), (vi) a business development company as defined in Section 2(a)(48) of the Investment Company Act, (vii) a Small Business Investment Company licensed by the U.S. Small Business Administration under Section 301 (c) or (d) of the Small Business Investment Act of 1958, as amended, (viii) a plan established and maintained by a state, its political subdivisions, or an agency or instrumentality of a state or its political subdivisions, for the benefit of its employees and you have total assets in excess of \$5,000,000, or (ix) an employee benefit plan within the meaning of the Employee Retirement Income Security Act of 1974, as amended ("ERISA") and (1) the decision that you shall subscribe for and purchase shares of common stock and warrants to purchase common stock (the "Units"), is made by a plan fiduciary, as defined in Section 3(21) of ERISA, which is either a bank, savings and loan association, insurance company, or registered investment adviser, or (2) you have total assets in excess of \$5,000,000 and the decision that you shall subscribe for and purchase the Shares is made solely by persons or entities that are accredited investors, as defined in Rule 501 of Regulation D promulgated under the Securities Act ("Regulation D") or (3) you are a self-directed plan and the decision that you shall subscribe for and purchase the Units is made solely by persons or entities that are accredited investors.
- [] You are a private business development company as defined in Section 202(a)(22) of the Investment Advisers Act of 1940, as amended.
- [] You are an organization described in Section 501(c)(3) of the Internal Revenue Code of 1986, as amended (the "<u>Code</u>"), a corporation, Massachusetts or similar business trust or a partnership, in each case not formed for the specific purpose of making an investment in the Units and its underlying securities in excess of \$5,000,000.
- [] You are a director or executive officer of the Company.
- [] You are a natural person whose individual net worth, or joint net worth with your spouse, exceeds \$1,000,000 (excluding residence) at the time of your subscription for and purchase of the Units.
- [] You are a natural person who had an individual income in excess of \$200,000 in each of the two most recent years or joint income with your spouse in excess of \$300,000 in each of the two most recent years, and who has a reasonable expectation of reaching the same income level in the current year.
- [] You are a trust, with total assets in excess of \$5,000,000, not formed for the specific purpose of acquiring the Units and whose subscription for and purchase of the Units is directed by a sophisticated person as described in Rule 506 (b)(2)(ii) of Regulation D.
- [] You are an entity in which all of the equity owners are persons or entities described in one of the preceding paragraphs.

- 22 -

#### Check all boxes below which correctly describe you.

With respect to this investment in the Units, your:

Investment Objectives:	[ ] Aggressive Growth	[] Speculation	
Risk Tolerance:	[ ] Low Risk	[] Moderate Risk	[] High Risk

Are you associated with a FINRA Member Firm? [ ] Yes [ ] No

Your initials (Subscriber) are required for each item below:

\_\_\_\_\_ I/We understand that this investment is not guaranteed.

\_\_\_\_\_ I/We are aware that this investment is not liquid.

- \_\_\_\_\_ I/We are sophisticated in financial and business affairs and are able to evaluate the risks and merits of an investment in this offering.
- I/We confirm that this investment is considered "high risk." (This type of investment is considered high risk due to the inherent risks including lack of liquidity and lack of diversification. Success or failure of private placements such as this is dependent on the corporate issuer of these securities and is outside the control of the investors. While potential loss is limited to the amount invested, such loss is possible.)

The Subscriber hereby represents and warrants that all of its answers to this Investor Questionnaire are true as of the date of its execution of the Subscription Agreement pursuant to which it purchased the Units.

Name of Subscriber [please print]

Signature of Subscriber (Entities please provide signature of Subscriber's duly authorized signatory)

Name of Signatory (Entities only)

Title of Signatory (Entities only)

[SIGNATURE PAGE FOR INVESTOR QUESTIONNAIRE]

- 23 -

# <u>Exhibit A</u>

# Term Sheet

See Attached.

- 24 -

# <u>Exhibit B</u>

# Purchase Agreement

See Attached.

- 25 -

# <u>Exhibit C</u>

# Investor Rights Agreement

See Attached.

- 26 -

# <u>Exhibit D</u>

# Form of Warrant

See Attached.

- 27 -

# <u>Exhibit E</u>

# Wire Instructions

# <u>Exhibit F</u>

# OUTSTANDING CONVERTIBLE SECURITIES OF THE COMPANY

# Stock Options

Manal 20, 2025	
March 30, 2025	4,500,000
November 30, 2027	5,000,000
January 6, 2030	350,000
August 3, 2030	500,000
TOTAL	10,350,000
	November 30, 2027 January 6, 2030 August 3, 2030

### <u>Warrants</u>

Exercise Price	Expiry Date	Number Outstanding
US\$0.10	February 11, 2021	8,200,000
US\$0.10	March 25, 2021	2,924,000
US\$0.15	May 23, 2021	9,750,000
C\$0.20	January 15, 2022	8,207,058
	TOTAL	29,081,058

- 29 -

#### SCHEDULE A

# CANADIAN ACCREDITED INVESTOR CERTIFICATE

#### To: Bullfrog Gold Corp. (the "Company")

The categories listed herein contain certain specifically defined terms. If you are unsure as to the meanings of those terms or are unsure as to the applicability of any category below, please contact your broker and/or legal advisor before completing this certificate. Capitalized terms not specifically defined in this Schedule A have the meanings ascribed to them in the Subscription Agreement to which this Schedule A is attached.

In connection with the purchase by the undersigned Subscriber of the Units, the Subscriber hereby represents, warrants, covenants and certifies to the Company (and acknowledges that the Company and its counsel are relying thereon) that:

- (a) the Subscriber is resident in or otherwise subject to the securities laws of one of the Provinces or Territories of Canada;
- (b) the Subscriber is purchasing the Units as principal for its own account and not for the benefit of any other person or is deemed to be purchasing as principal pursuant to NI 45-106;
- (c) the Subscriber is, and at the Closing Date will be, an "accredited investor" within the meaning of NI 45-106 on the basis that the undersigned fits within one of the categories of an "accredited investor" reproduced below beside which the undersigned has indicated the undersigned belongs to such category;
- (d) the Subscriber was not created or is not used, solely to purchase or hold securities as an accredited investor as described in paragraph (m) below; and
- (e) upon execution of this Schedule A by the Subscriber, including, if applicable, Appendix 1 to this Schedule A, this Schedule A shall be incorporated into and form a part of the Subscription Agreement.

#### (PLEASE CHECK THE BOX OF THE APPLICABLE CATEGORY OF ACCREDITED INVESTOR)

NOTE: If you check the box beside <u>paragraphs (j)</u>, (k) or (l) below, you <u>must</u> complete and execute Appendix 1 to this Schedule A.

- [] (a) a Canadian financial institution, or a Schedule III bank (or in Ontario, a bank listed in Schedule I, II, or III of the *Bank Act* (Canada));
- [] (b) the Business Development Bank of Canada incorporated under the *Business Development Bank of Canada Act* (Canada);
- [] (c) a subsidiary of any person referred to in paragraphs (a) or (b), if the person owns all of the voting securities of the subsidiary, except the voting securities required by law to be owned by directors of that subsidiary;
- [] (d) a person registered under the securities legislation of a jurisdiction of Canada as an adviser or dealer;
- [] (e) an individual registered under the securities legislation of a jurisdiction of Canada as a representative of a person referred to in paragraph (d);



- [] (e.1) an individual formerly registered under the securities legislation of a jurisdiction of Canada, other than an individual formerly registered solely as a representative of a limited market dealer under one or both of the *Securities Act* (Ontario) or the *Securities Act* (Newfoundland and Labrador);
- [] (f) the Government of Canada or a jurisdiction of Canada, or any crown corporation, agency or wholly owned entity of the Government of Canada or a jurisdiction of Canada;
- [] (g) a municipality, public board or commission in Canada and a metropolitan community, school board, the Comité de gestion de la taxe scolaire de l'île de Montréal or an intermunicipal management board in Québec;
- [] (h) any national, federal, state, provincial, territorial or municipal government of or in any foreign jurisdiction, or any agency of that government;
- [] (i) a pension fund that is regulated by either the Office of the Superintendent of Financial Institutions (Canada), a pension commission or similar regulatory authority of a jurisdiction of Canada;
- [] (j) an individual who, either alone or with a spouse, beneficially owns, financial assets having an aggregate realizable value that, before taxes but net of any related liabilities, exceeds \$1,000,000;

#### If you check beside paragraph (j) above, you <u>must</u> complete and execute Appendix 1 to this Schedule A.

- [] (j.1) an individual who beneficially owns financial assets having an aggregate realizable value that, before taxes but net of any related liabilities, exceeds \$5,000,000;
- [] (k) an individual whose net income before taxes exceeded \$200,000 in each of the 2 most recent calendar years or whose net income before taxes combined with that of a spouse exceeded \$300,000 in each of the 2 most recent calendar years and who, in either case, reasonably expects to exceed that net income level in the current calendar year;

#### If you check beside paragraph (k) above, you <u>must</u> complete and execute Appendix 1 to this Schedule A.

(1) an individual who, either alone or with a spouse, has net assets of at least \$5,000,000;

[]

#### If you check beside paragraph (l) above, you <u>must</u> complete and execute Appendix 1 to this Schedule A.

- [] (m) a person, other than an individual or investment fund, that has net assets of at least \$5,000,000 as shown on its most recently prepared financial statements;
- [] (n) an investment fund that distributes or has distributed its securities only to (i) a person that is or was an accredited investor at the time of the distribution, (ii) a person that acquires or acquired securities in the circumstances referred to in sections 2.10 [*Minimum amount investment*], or 2.19 [*Additional investment in investment funds*] of NI 45-106, or (iii) a person described in paragraph (i) or (ii) that acquires or acquired securities under section 2.18 [*Investment fund reinvestment*] of NI 45-106;
- [] (o) an investment fund that distributes or has distributed securities under a prospectus in a jurisdiction of Canada for which the regulator or, in Québec, the securities regulatory authority, has issued a receipt;

- 31 -

- [] (p) a trust company or trust corporation registered or authorized to carry on business under the *Trust and Loan Companies Act* (Canada) or under comparable legislation in a jurisdiction of Canada or a foreign jurisdiction, acting on behalf of a fully managed account managed by the trust company or trust corporation, as the case may be;
- [] (q) a person acting on behalf of a fully managed account managed by that person, if that person is registered or authorized to carry on business as an adviser or the equivalent under the securities legislation of a jurisdiction of Canada or a foreign jurisdiction;
- [] (r) a registered charity under the *Income Tax Act* (Canada) that, in regard to the trade, has obtained advice from an eligibility adviser or an adviser registered under the securities legislation of the jurisdiction of the registered charity to give advice on the securities being traded;
- [] (s) an entity organized in a foreign jurisdiction that is analogous to any of the entities referred to in paragraphs (a) to (d) or paragraph (i) in form and function;
- [] (t) a person in respect of which all of the owners of interests, direct, indirect or beneficial, except the voting securities required by law to be owned by directors, are persons that are accredited investors;
- [] (u) an investment fund that is advised by a person registered as an adviser or a person that is exempt from registration as an adviser;
- [] (v) a person that is recognized or designated by the securities regulatory authority or, except in Ontario and Québec, the regulator as an accredited investor; or
- [] (w) a trust established by an accredited investor for the benefit of the accredited investor's family members of which a majority of the trustees are accredited investors and all of the beneficiaries are the accredited investor's spouse, a former spouse of the accredited investor or a parent, grandparent, brother, sister, child or grandchild of that accredited investor, of that accredited investor's spouse or of that accredited investor's former spouse.

For the purposes hereof, the following definitions are included for convenience:

- A. "bank" means a bank named in Schedule I or II of the Bank Act (Canada);
- B. "Canadian financial institution" means (i) an association governed by the *Cooperative Credit Associations Act* (Canada) or a central cooperative credit society for which an order has been made under section 473(1) of that Act, or (ii) a bank, loan corporation, trust company, trust corporation, insurance company, treasury branch, credit union, caisse populaire, financial services cooperative, or credit union league or federation that, in each case, is authorized by an enactment of a statute of Canada or a jurisdiction of Canada to carry on business in Canada or a jurisdiction of Canada;
- C. "company" means any corporation, incorporated association, incorporated syndicate or other incorporated organization;
- D. "entity" means a company, syndicate, partnership, trust or unincorporated organization;
- E. "financial assets" means (i) cash, (ii) securities, or (iii) a contract of insurance, a deposit or an evidence of a deposit that is not a security for the purposes of securities legislation;
- F. "founder" means, in respect of an issuer, a person who, (i) acting alone, in conjunction, or in concert with one or more persons, directly or indirectly, takes the initiative in founding, organizing or substantially reorganizing the business of the issuer, and (ii) at the time of the distribution or trade is actively involved in the business of the issuer;
- G. "fully managed account" means an account of a client for which a person makes the investment decisions if that person has full discretion to trade in securities for the account without requiring the client's express consent to a transaction;

- 32 -

- H. "individual" means a natural person, but does not include a partnership, unincorporated association, unincorporated organization, trust, or a natural person in his or her capacity as trustee, executor, administrator or other legal personal representative;
- I. "investment fund" means a mutual fund or a non-redeemable investment fund, and, for greater certainty in British Columbia, includes an employee venture capital corporation that does not have a restricted constitution, and is registered under Part 2 of the *Employee Investment Act* (British Columbia), R.S.B.C. 1996 c. 112, and whose business objective is making multiple investments and a venture capital corporation registered under Part 1 of the *Small Business Venture Capital Act* (British Columbia), R.S.B.C. 1996 c. 429 whose business objective is making multiple investments;
- J. "person" includes (a) an individual, (b) a corporation, (c) a partnership, trust, fund and an association, syndicate, organization or other organized group of persons, whether incorporated or not, and (d) an individual or other person in that person's capacity as a trustee, executor, administrator or personal or other legal representative;
- K. "related liabilities" means (i) liabilities incurred or assumed for the purpose of financing the Amalgamation or ownership of financial assets or (ii) liabilities that are secured by financial assets;
- L. "Schedule III bank" means an authorized foreign bank named in Schedule III of the Bank Act (Canada);
- M. "spouse" means an individual who (i) is married to another individual and is not living separate and apart within the meaning of the *Divorce Act* (Canada), from the other individual, (ii) is living with another individual in a marriage-like relationship, including a marriage-like relationship between individuals of the same gender, or (iii) in Alberta, is an individual referred to in paragraph (i) or (ii), or is an adult interdependent partner within the meaning of the *Adult Interdependent Relationships Act* (Alberta); and
- N. "subsidiary" means an issuer that is controlled directly or indirectly by another issuer and includes a subsidiary of that subsidiary.

In NI 45-106 a person or company is an affiliate of another person or company if one is a subsidiary of the other, or if each of them is controlled by the same person or company.

In NI 45-106 and except in Part 2 Division 4 of NI 45-106, a person (first person) is considered to control another person (second person) if (a) the first person beneficially owns or directly or indirectly exercises control or direction over securities of the second person carrying votes which, if exercised, would entitle the first person to elect a majority of the directors of the second person, unless that first person holds the voting securities only to secure an obligation, (b) the second person is a partnership, other than a limited partnership, and the first person holds more than 50% of the interests of the partnership, or (c) the second person is a limited partnership and the general partner of the limited partnership is the first person.

In NI 45-106 a trust company or trust corporation described in paragraph (p) above of the definition of "accredited investor" (other than in respect of a trust company or trust corporation registered under the laws of Prince Edward Island that is not registered or authorized under the *Trust and Loan Companies Act* (Canada) or under comparable legislation in another jurisdiction of Canada) is deemed to be purchasing as principal.

In NI 45-106 a person described in paragraph (q) above of the definition of "accredited investor" is deemed to be purchasing as principal.

- 33 -

The foregoing representations contained in this certificate are true and accurate as of the date of this certificate and will be true and accurate as of the Closing Date. If any such representations shall not be true and accurate prior to the Closing Date, the undersigned shall give immediate written notice of such fact to the Company prior to the Closing Date.

DATED:

SIGNED :

Print the name of Subscriber

If Subscriber is not an Individual, print name and title of Authorized Signing Officer

- 34 -

# APPENDIX 1 TO SCHEDULE A RISK ACKNOWLEDGEMENT CERTIFICATE

# Form 45-106F9 Form for Individual Accredited Investors

WARNING!

This investment is risky. Don't invest unless you can afford to lose all the money you pay for this investment.

SECTION 1 TO BE COMPLETED BY THE ISSUER OR SELLING SECURITY HOLDER				
1. About your investment				
Type of securities: Units Issuer: Bullfrog Gold Corp.				
SECTIONS 2 TO 4 TO BE COMPLETED BY THE SUBSCRIBER				
2. Risk acknowledgement				
This investment is risky. <b>Initial</b> to the right of each category that you understand that:	Your initials			
Risk of loss – You could lose your entire investment of \$ [Instruction: Insert the total dollar amount of the investment.]				
Liquidity risk – You may not be able to sell your investment quickly – or at all.				
Lack of information – You may receive little or no information about your investment.				
<b>Lack of advice</b> – You will not receive advice from the salesperson about whether this investment is suitable for you unless the salesperson is registered. The salesperson is the person who meets with, or provides information to, you about making this investment. To check whether the salesperson is registered, go to <u>www.aretheyregistered.ca</u> .				
3. Accredited investor status				
You must meet at least <b><u>one</u></b> of the following criteria to be able to make this investment. Initial the statement that applies to you. (You may initial more than one statement.) The person identified in section 6 is responsible for ensuring that you meet the definition of accredited investor. That person, or the salesperson identified in section 5, can help you if you have questions about whether you meet these criteria.	Your initials			
• Your net income before taxes was more than \$200,000 in each of the 2 most recent calendar years, and you expect it to be more than \$200,000 in the current calendar year. (You can find your net income before taxes on your personal income tax return.)				
• Your net income before taxes combined with your spouse's was more than \$300,000 in each of the 2 most recent calendar years, and you expect your combined net income before taxes to be more than \$300,000 in the current calendar year.				
• Either alone or with your spouse, you own more than \$1 million in cash and securities, after subtracting any debt related to the cash and securities.				
• Either alone or with your spouse, you have net assets worth more than \$5 million. (Your net assets are your total assets (including real estate) minus your total debt.)				
4. Your name and signature				
By signing this form, you confirm that you have read this form and you understand the risks of making this investment as identified this form.	ntified in			

- 35 -

Signature: x	Date:, 2020		
SECTION 5 TO BE COMPLETED BY THE SALESPERS	SON		
5. Salesperson information			
	r provides information to, the subscriber with respect to making this or selling security holder, a registrant or a person who is exempt from the		
First and last name of salesperson (please print):			
Telephone:	E-mail:		
Name of firm (if registered):			
SECTION 6 TO BE COMPLETED BY THE ISSUER OR	SELLING SECURITY HOLDER		
6. For more information about this investment			
Bullfrog Gold Corp.			
897 Quail Run Drive Grand Junction, Colorado 81505			
Attention: David Beling Email: dave@bullfroggold.com			

2. The subscriber must sign this form. Each of the subscriber and the issuer or selling security holder must receive a copy of this form signed by the subscriber. The issuer or selling security holder is required to keep a copy of this form for 8 years after the distribution.

- 36 -

#### SCHEDULE B

# CONTACT INFORMATION – CANADIAN PROVINCIAL SECURITIES REGULATORY AUTHORITIES

The contact information of the public official in the local jurisdiction who can answer questions about the security regulatory authority's or regulator's indirect collection of information is as follows:

#### **British Columbia Securities Commission**

P.O. Box 10142, Pacific Centre 701 West Georgia Street Vancouver, British Columbia V7Y 1L2 Inquiries: (604) 899-6854 Toll free in Canada: 1-800-373-6393 Facsimile: (604) 899-6581 Email: inquiries@bcsc.bc.ca

#### **Ontario Securities Commission**

20 Queen Street West, 22<sup>nd</sup> Floor Toronto, Ontario M5H 3S8 Telephone: (416) 593- 8314 Toll free in Canada: 1-877-785-1555 Facsimile: (416) 593-8122 Email: exemptmarketfilings@osc.gov.on.ca Public official contact regarding indirect collection of information: Inquiries Officer

- 37 -

#### Exhibit 10.2

NEITHER THIS SECURITY NOR THE SECURITIES FOR WHICH THIS SECURITY IS EXERCISABLE HAS BEEN REGISTERED WITH THE U.S. SECURITIES AND EXCHANGE COMMISSION OR THE SECURITIES COMMISSION OF ANY STATE IN RELIANCE UPON AN EXEMPTION FROM REGISTRATION UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT") AND APPLICABLE STATE SECURITIES LAWS, AND, ACCORDINGLY, MAY NOT BE OFFERED, SOLD, DELIVERED AFTER SALE, TRANSFERRED, PLEDGED OR HYPOTHECATED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT FILED BY THE COMPANY WITH THE U.S. SECURITIES AND EXCHANGE COMMISSION COVERING SUCH SECURITIES UNDER THE SECURITIES ACT OR AN OPINION OF COUNSEL THAT SUCH REGISTRATION IS NOT REQUIRED.

# UNLESS PERMITTED UNDER SECURITIES LEGISLATION, THE HOLDER OF THIS SECURITY MUST NOT TRADE THE SECURITY BEFORE [INSERT DATE THAT IS 4 MONTHS AND A DAY AFTER THE DATE OF THIS CERTIFICATE].

#### COMMON STOCK PURCHASE WARRANT

#### BULLFROG GOLD CORP.

Warrant Shares: [•] Warrant No: [•]W- [•] Initial Issuance Date: October [•], 2020

THIS COMMON STOCK PURCHASE WARRANT (the "<u>Warrant</u>") certifies that, for value received,  $[\bullet]$  or its assigns (the "<u>Holder</u>") is entitled, upon the terms and subject to the limitations on exercise and the conditions hereinafter set forth, at any time prior to the close of business on October  $[\bullet]$ , 2024 (the "<u>Termination Date</u>") but not thereafter, to subscribe for and purchase from BULLFROG GOLD CORP., a Delaware corporation (the "<u>Company</u>"), up to  $[\bullet]$  shares (as subject to adjustment hereunder, the "<u>Warrant Shares</u>") of Common Stock of the Company ("<u>Common Stock</u>"). The purchase price of one share of Common Stock under this Warrant shall be equal to the Exercise Price, as defined in Section 2(b).

Section 1. Definitions and Other Terms. Capitalized terms used and not otherwise defined herein shall have the meanings set forth in that certain [Membership Interest Purchase Agreement/Subscription Agreement], dated October [ $\bullet$ ], 2020, among the Company and the signatories thereto (the "<u>Purchase Agreement</u>"). This Warrant was issued pursuant to the terms of the Purchase Agreement and certain holders are parties to an Investor Rights Agreement dated October [ $\bullet$ ], 2020, among the Company and the signatories thereto (the "<u>Investor Rights Agreement</u>").

Section 2. Exercise.

a) Exercise of Warrant. Exercise of the purchase rights represented by this Warrant may be made, in whole or in part, at any time or times on or before the Termination Date by delivery to the Company (or such other office or agency of the Company as it may designate by notice in writing to the registered Holder at the address of the Holder appearing on the books of the Company) of a duly executed facsimile copy or PDF copy submitted by e-mail (or e-mail attachment) of the Notice of Exercise in the form annexed hereto (the "Notice of Exercise"). Within two Business Days following the date of exercise as aforesaid, the Holder shall deliver the aggregate Exercise Price for the shares specified in the applicable Notice of Exercise by wire transfer or cashier's check drawn on a Canadian or United States bank. No ink-original Notice of Exercise shall be required, nor shall any medallion guarantee (or other type of guarantee or notarization) of any Notice of Exercise be required. Notwithstanding anything herein to the contrary, the Holder shall not be required to physically surrender this Warrant to the Company until the Holder has purchased all of the Warrant Shares available hereunder and the Warrant has been exercised in full, in which case, the Holder shall surrender this Warrant to the Company for cancellation within three Business Days of the date on which the final Notice of Exercise is delivered to the Company. Partial exercises of this Warrant resulting in purchases of a portion of the total number of Warrant Shares available hereunder shall have the effect of lowering the outstanding number of Warrant Shares purchasable hereunder in an amount equal to the applicable number of Warrant Shares purchased. The Holder and the Company shall maintain records showing the number of Warrant Shares purchased and the date of such purchases. The Company shall deliver any objection to any Notice of Exercise within one Business Day of receipt of such notice. The Holder and any assignee, by acceptance of this Warrant, acknowledge and agree that, by reason of the provisions of this paragraph, following the purchase of a portion of the Warrant Shares hereunder, the number of Warrant Shares available for purchase hereunder at any given time may be less than the amount stated on the face hereof.

b) <u>Exercise Price</u>. The exercise price per share of the Common Stock under this Warrant shall be CDN\$0.30, or the prevailing equivalent price in the lawful currency of the United States of America calculated using the daily average exchange rate of the Bank of Canada on the Business Day prior to the date of exercise, subject to adjustment hereunder (the "<u>Exercise</u> <u>Price</u>").

#### c) Mechanics of Exercise.

i. Delivery of Warrant Shares Upon Exercise. The Company shall cause the Warrant Shares purchased hereunder to be transmitted by National Securities Administrators Ltd. (or any successor or replacement thereof) (the "Transfer Agent") to the Holder by (1) crediting the account of the Holder's or its designee's balance account with The Depository Trust Company through its Deposit or Withdrawal at Custodian system ("DWAC") if the Company is then a participant in such system or (2) issuing and causing to be deposited electronically with CDS Clearing and Depositary Services Inc. ("CDS") through the book-based system administered by CDS using the "non-certificated inventory" issue process that number of Warrant Shares issuable on exercise of the Warrants so exercised, if either (A) there is an effective prospectus or registration statement permitting the issuance of the Warrant Shares to or resale of the Warrant Shares by the Holder pursuant to applicable securities laws (including under the Securities Act) or (B) the Warrant Shares are eligible for resale (without volume or manner of sale limitations, and without the need for current public information) by the Holder pursuant to applicable securities laws, including the resale restrictions under Canadian securities laws, if applicable, and Rule 144 in the United States (and provided in each case the Company has received an opinion to that effect reasonably acceptable to the Company), otherwise, the Company shall cause the Warrant Shares purchased hereunder to be transmitted by the Transfer Agent to the Holder by physical delivery of a certificate, registered in the Company's share register in the name of the Holder or its designee, for the number of Warrant Shares to which the Holder is entitled pursuant to such exercise to the address specified by the Holder in the Notice of Exercise by the date that is three Business Days after the delivery to the Company of the Notice of Exercise. Upon the later of delivery of the Notice of Exercise and payment of the Exercise Price, the Holder shall be deemed for all corporate purposes to have become the holder of record of the Warrant Shares with respect to which this Warrant has been exercised, irrespective of the date of delivery of the Warrant Shares. The Company agrees to maintain a transfer agent that is a participant in the FAST program so long as this Warrant remains outstanding and exercisable.

ii. <u>Delivery of New Warrants Upon Exercise</u>. If this Warrant shall have been exercised in part, the Company shall, at the request of a Holder and upon surrender of this Warrant certificate, at the time of delivery of the Warrant Shares, deliver to the Holder a new Warrant evidencing the rights of the Holder to purchase the unpurchased Warrant Shares called for by this Warrant, which new Warrant shall in all other respects be identical with this Warrant.

iii. <u>No Fractional Shares or Scrip</u>. No fractional shares or scrip representing fractional shares shall be issued upon the exercise of this Warrant. As to any fraction of a share which the Holder would otherwise be entitled to purchase upon such exercise, the Company shall, at its election, either pay a cash adjustment in respect of such fraction in an amount equal to such fraction multiplied by the Exercise Price or round up to the next whole share.

iv. <u>Charges, Taxes and Expenses</u>. Issuance of Warrant Shares shall be made without charge to the Holder for any issue or transfer tax or other incidental expense in respect of the issuance of such Warrant Shares, all of which taxes and expenses shall be paid by the Company, and such Warrant Shares shall be issued in the name of the Holder or in a name or names of an Affiliate of the Holder as may be directed by the Holder; <u>provided</u>, <u>however</u>, that in the event that Warrant Shares are to be issued in a name other than the name of the Holder in compliance with Section 4 (a), this Warrant when surrendered for exercise shall be accompanied by the Assignment Form attached hereto duly executed by the Holder and the Company may require, as a condition thereto, the payment of a sum sufficient to reimburse it for any transfer tax incidental thereto. The Company shall pay all Transfer Agent fees required for same-day processing of any Notice of Exercise and all fees to The Depository Trust Company, CDS or another established clearing corporation performing similar functions, as applicable, required for same-day electronic delivery of the Warrant Shares.

v. <u>Closing of Books</u>. The Company will not close its stockholder books or records in any manner which prevents the timely exercise of this Warrant, pursuant to the terms hereof.

#### Section 3. Certain Adjustments.

a) <u>Stock Dividends and Splits</u>. If the Company, at any time while this Warrant is outstanding: (i) pays a stock dividend or otherwise makes a distribution or distributions on shares of its Common Stock or any other equity or equity equivalent securities payable in shares of Common Stock to all of the record holders of any class of shares of Common Stock, (ii) subdivides outstanding shares of Common Stock into a larger number of shares, (iii) combines (including by way of reverse stock split) outstanding shares of Common Stock into a smaller number of shares, or (iv) issues by reclassification of shares of the Common Stock any shares of capital stock of the Company, then in each case the Exercise Price shall be multiplied by a fraction of which the numerator shall be the number of shares of Common Stock (excluding treasury shares, if any) outstanding immediately before such event and of which the denominator shall be the number of shares of Common Stock outstanding immediately after such event, and the number of shares issuable upon exercise of this Warrant shall be proportionately adjusted such that the aggregate Exercise Price of this Warrant shall remain unchanged. Any adjustment made pursuant to this Section 3(a) shall become effective immediately after the effective date in the case of a subdivision, combination or re-classification.

b) <u>Subsequent Rights Offerings</u>. In addition to any adjustments pursuant to Section 3(a) above, if at any time the Company grants, issues or sells any equity interests or rights to purchase stock, warrants, securities or other property pro rata to all of the record holders of any class of shares of Common Stock (the "<u>Purchase Rights</u>"), then the Holder will be entitled to acquire, upon the terms applicable to such Purchase Rights, the aggregate Purchase Rights which the Holder could have acquired if the Holder had held the number of shares of Common Stock acquirable upon complete exercise of this Warrant (without regard to any limitations on exercise hereof) immediately before the date on which a record is taken for the grant, issuance or sale of such Purchase Rights, or, if no such record is taken, the date as of which the record holders of shares of Common Stock are to be determined for the grant, issue or sale of such Purchase Rights.

c) <u>Pro Rata Distributions</u>. During such time as this Warrant is outstanding, if the Company shall declare or make any dividend or other distribution of its assets (or rights to acquire its assets) to holders of shares of Common Stock, by way of return of capital or otherwise (including, without limitation, any distribution of cash, stock or other securities, property or options by way of a dividend, spin off, reclassification, corporate rearrangement, scheme of arrangement or other similar transaction) (a "<u>Distribution</u>"), at any time after the issuance of this Warrant, then, in each such case, the Holder shall be entitled to participate in such Distribution to the same extent that the Holder would have participated therein if the Holder had held the number of shares of Common Stock acquirable upon complete exercise of this Warrant (without regard to any limitations on exercise hereof) immediately before the date of which a record is taken for such Distribution, or, if no such record is taken, the date as of which the record holders of shares of Common Stock are to be determined for the participation in such Distribution.

d) Fundamental Transaction. If, at any time while this Warrant is outstanding, (i) the Company, directly or indirectly, in one or more related transactions effects any merger or consolidation of the Company with or into another Person, (ii) the Company, directly or indirectly, effects any sale, lease, license, assignment, transfer, conveyance or other disposition of all or substantially all of its assets in one or a series of related transactions, (iii) any, direct or indirect, purchase offer, tender offer or exchange offer (whether by the Company or another Person) is completed pursuant to which holders of Common Stock are permitted to sell, tender or exchange their shares for other securities, cash or property and has been accepted by the holders of 50% or more of the outstanding Common Stock, (iv) the Company, directly or indirectly, in one or more related transactions effects any reclassification, reorganization or recapitalization of the Common Stock or any compulsory share exchange pursuant to which the Common Stock is effectively converted into or exchanged for other securities, cash or property (which will not include, for the avoidance of doubt, any stock split, reverse stock split or other transaction that would result in an adjustment to the exercise price under Section 3(a)), or (v) the Company, directly or indirectly, in one or more related transactions consummates a stock or share purchase agreement or other business combination (including, without limitation, a reorganization, recapitalization, spin-off or scheme of arrangement) with another Person or group of Persons whereby such other Person or group acquires more than 50% of the outstanding shares of Common Stock (not including any shares of Common Stock held by the other Person or other Persons making or party to, or associated or affiliated with the other Persons making or party to, such stock or share purchase agreement or other business combination) (each a "Fundamental Transaction"), then, upon any subsequent exercise of this Warrant, the Holder shall have the right to receive, for each Warrant Share that would have been issuable upon such exercise immediately prior to the occurrence of such Fundamental Transaction, at the option of the Holder, the number of shares of Common Stock of the successor or acquiring corporation or of the Company, if it is the surviving corporation, and any additional consideration (the "Alternate Consideration") receivable as a result of such Fundamental Transaction by a holder of the number of shares of Common Stock for which this Warrant is exercisable immediately prior to such Fundamental Transaction. For purposes of any such exercise, the determination of the Exercise Price shall be appropriately adjusted to apply to such Alternate Consideration based on the amount of Alternate Consideration issuable in respect of one share of Common Stock in such Fundamental Transaction, and the Company shall apportion the Exercise Price among the Alternate Consideration in a reasonable manner reflecting the relative value of any different components of the Alternate Consideration. If holders of Common Stock are given any choice as to the securities, cash or property to be received in a Fundamental Transaction, then the Holder shall be given the same choice as to the Alternate Consideration it receives upon any exercise of this Warrant following such Fundamental Transaction. The Company shall cause any successor entity in a Fundamental Transaction in which the Company is not the survivor (the "Successor Entity") to assume in writing all of the obligations of the Company under this Warrant and the other Transaction Documents in accordance with the provisions of this Section 3(d) pursuant to written agreements in form and substance reasonably satisfactory to the Holder and approved by the Holder (without unreasonable delay) prior to such Fundamental Transaction and shall, at the option of the Holder, deliver to the Holder in exchange for this Warrant a security of the Successor Entity evidenced by a written instrument substantially similar in form and substance to this Warrant which is exercisable for a corresponding number of shares of capital stock of such Successor Entity (or its parent entity) equivalent to the shares of Common Stock acquirable and receivable upon exercise of this Warrant (without regard to any limitations on the exercise of this Warrant) prior to such Fundamental Transaction, and with an exercise price which applies the exercise price hereunder to such shares of capital stock (which exercise price shall protect the economic value of this Warrant immediately prior to the consummation of such Fundamental Transaction), and which is reasonably satisfactory in form and substance to the Holder. Upon the occurrence of any such Fundamental Transaction, the Successor Entity shall succeed to, and be substituted for (so that from and after the date of such Fundamental Transaction, the provisions of this Warrant and the other Transaction Documents referring to the "Company" or "Corporation" shall refer instead to the Successor Entity), and may exercise every right and power of the Company and shall assume all of the obligations of the Company under this Warrant and the other Transaction Documents with the same effect as if such Successor Entity had been named as the Company herein.

e) <u>Calculations</u>. All calculations under this Section 3 shall be made to the nearest cent or the nearest 1/100<sup>th</sup> of a share, as the case may be. For purposes of this Section 3, the number of shares of Common Stock deemed to be issued and outstanding as of a given date shall be the sum of the number of shares of Common Stock (excluding treasury shares, if any) issued and outstanding.

#### f) Notice to Holder.

i. <u>Adjustment to Exercise Price</u>. Whenever the Exercise Price is adjusted pursuant to any provision of this Section 3, the Company shall promptly provide notice to the Holder setting forth the Exercise Price after such adjustment and any resulting adjustment to the number of Warrant Shares and setting forth a brief statement of the facts requiring such adjustment.

ii. Notice to Allow Exercise by Holder. If (A) the Company shall declare a dividend (or any other distribution in whatever form) on the Common Stock, (B) the Company shall declare a special nonrecurring cash dividend on or a redemption of the Common Stock, (C) the Company shall authorize the granting to all holders of the Common Stock rights or warrants to subscribe for or purchase any shares of capital stock of any class or of any rights, (D) the approval of any stockholders of the Company shall be required in connection with any reclassification of the Common Stock or any Fundamental Transaction, or (E) the Company shall authorize the voluntary or involuntary dissolution, liquidation or winding up of the affairs of the Company, then, in each case, the Company shall provide notice to the Holder, at least 20 calendar days prior to the applicable record or effective date hereinafter specified, a notice stating (x) the date on which a record is to be taken for the purpose of such dividend, distribution, redemption, rights or warrants, or if a record is not to be taken, the date as of which the holders of the Common Stock of record to be entitled to such dividend, distributions, redemption, rights or warrants are to be determined or (y) the date on which such reclassification or Fundamental Transaction is expected to become effective or close, and the date as of which it is expected that holders of the Common Stock of record shall be entitled to exchange their shares of the Common Stock for securities, cash or other property deliverable upon such reclassification or Fundamental Transaction; provided that the failure to mail such notice or any defect therein or in the mailing thereof shall not affect the validity of the corporate action required to be specified in such notice. To the extent that any notice provided hereunder constitutes, or contains, material, non-public information regarding the Company or any of its subsidiaries, the Company shall simultaneously file such notice with the Canadian securities authorities or SEC, as applicable, pursuant to a Current Report on Form 8-K and any equivalent filing required under Canadian securities laws. The Holder shall remain entitled to exercise this Warrant during the period commencing on the date of such notice to the effective date of the event triggering such notice except as may otherwise be expressly set forth herein.

#### Section 4. Transfer of Warrant.

a) <u>Transferability</u>. This Warrant and all rights hereunder are non-assignable, non-transferable and non-negotiable and may not be exercised by or for the benefit of any person other than the Holder or an Affiliate of the Holder, except with the prior written consent of the Company.

b) <u>New Warrants</u>. This Warrant may be divided or combined with other Warrants upon presentation hereof at the aforesaid office of the Company, together with a written notice specifying the names and denominations in which new Warrants are to be issued, signed by the Holder or its agent or attorney. Subject to compliance with Section 4(a), as to any transfer which may be involved in such division or combination, the Company shall execute and deliver a new Warrant or Warrants in exchange for the Warrant or Warrants to be divided or combined in accordance with such notice. All Warrants issued on transfers or exchanges shall be dated the initial issuance date of this Warrant and shall be identical with this Warrant except as to the number of Warrant Shares issuable pursuant thereto.

c) <u>Warrant Register</u>. The Company shall register this Warrant, upon records to be maintained by the Company or its transfer agent for that purpose (the "<u>Warrant Register</u>"), in the name of the registered Holder hereof from time to time. The Company may deem and treat the registered Holder of this Warrant as the absolute owner hereof for the purpose of any exercise hereof or any distribution to the Holder, and for all other purposes, absent actual notice to the contrary.

#### Section 5. Miscellaneous.

a) <u>No Rights as Stockholder Until Exercise</u>. This Warrant does not entitle the Holder to any voting rights, dividends or other rights as a stockholder of the Company prior to the exercise hereof as set forth in Section 2(c)(i).

b) Loss, Theft, Destruction or Mutilation of Warrant. The Company covenants that upon receipt by the Company of evidence reasonably satisfactory to it of the loss, theft, destruction or mutilation of this Warrant or any stock certificate relating to the Warrant Shares, and in case of loss, theft or destruction, of indemnity or security reasonably satisfactory to it (which, in the case of the Warrant, shall not include the posting of any bond), and upon surrender and cancellation of such Warrant or stock certificate, if mutilated, the Company will make and deliver a new Warrant or stock certificate of like tenor and dated as of such cancellation, in lieu of such Warrant or stock certificate.

c) <u>Saturdays, Sundays, Holidays, etc</u>. If the last or appointed day for the taking of any action or the expiration of any right required or granted herein shall not be a Business Day, then, such action may be taken or such right may be exercised on the next succeeding Business Day.

#### d) Authorized Shares.

(i) The Company covenants that, during the period the Warrant is outstanding, it will reserve from its authorized and unissued Common Stock 100% of the maximum number of shares for the issuance of the Warrant Shares upon the exercise of any purchase rights under this Warrant. The Company further covenants that its issuance of this Warrant shall constitute full authority to its officers who are charged with the duty of executing stock certificates to execute and issue the necessary certificates for the Warrant Shares upon the exercise of the purchase rights under this Warrant. The Company will take all such reasonable action as may be necessary to assure that such Warrant Shares may be issued as provided herein without violation of any applicable law or regulation, or of any requirements of the principal market upon which the Common Stock may be listed. The Company covenants that all Warrant Shares which may be issued upon the exercise of the purchase rights represented by this Warrant will, upon exercise of the purchase rights represented by this Warrant will, upon exercise of the purchase rights under thore issued, fully paid and non-assessable and free from all taxes, liens and charges created by the Company in respect of the issue thereof (other than taxes in respect of any transfer occurring contemporaneously with such issue).

(ii) Except and to the extent as waived or consented to by the Holder, the Company shall not by any action, including, without limitation, amending its certificate of incorporation or through any reorganization, transfer of assets, consolidation, merger, dissolution, issue or sale of securities or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, but will at all times in good faith assist in the carrying out of all such terms and in the taking of all such actions as may be necessary or appropriate to protect the rights of the Holder as set forth in this Warrant against impairment. Without limiting the generality of the foregoing, the Company will (i) not increase the par value of any Warrant Shares above the amount payable therefor upon such exercise immediately prior to such increase in par value, (ii) take all such action as may be necessary or appropriate in order that the Company may validly and legally issue fully paid and non-assessable Warrant Shares upon the exercise of this Warrant, and (iii) use commercially reasonable efforts to obtain all such authorizations, exemptions or consents from any public regulatory body having jurisdiction thereof as may be necessary to enable the Company to perform its obligations under this Warrant.

(iii) Before taking any action which would result in an adjustment in the number of Warrant Shares for which this Warrant is exercisable or in the Exercise Price, the Company shall obtain all such authorizations or exemptions thereof, or consents thereto, as may be necessary from any public regulatory body or bodies having jurisdiction thereof.

e) <u>Legends</u>. The certificate or certificates representing Warrant Shares issued upon the exercise of the Warrants represented hereby, if required by law, will contain a legend substantially in the following form:

THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), OR ANY APPLICABLE STATE SECURITIES LAWS. SUCH SECURITIES HAVE BEEN ACQUIRED FOR INVESTMENT PURPOSES AND MAY NOT BE OFFERED FOR SALE, SOLD, DELIVERED AFTER SALE, TRANSFERRED, PLEDGED OR HYPOTHECATED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT FILED BY THE COMPANY WITH THE U.S. SECURITIES AND EXCHANGE COMMISSION COVERING SUCH SECURITIES UNDER THE SECURITIES ACT OR AN OPINION OF COUNSEL THAT SUCH REGISTRATION IS NOT REQUIRED.

THE PRESENCE OF THIS LEGEND MAY IMPAIR THE ABILITY OF THE HOLDER HEREOF TO EFFECT "GOOD DELIVERY" OF THE SECURITIES REPRESENTED HEREBY ON A CANADIAN STOCK EXCHANGE.

#### UNLESS PERMITTED UNDER SECURITIES LEGISLATION, THE HOLDER OF THIS SECURITY MUST NOT TRADE THE SECURITY BEFORE [INSERT DATE THAT IS 4 MONTHS AND A DAY AFTER THE DATE OF THIS CERTIFICATE].

f) <u>Governing Law</u>. This Warrant shall be governed by and construed in accordance with the laws of the Province of Ontario and the federal laws of Canada applicable therein. Each of the Company and the Holder irrevocably and unconditionally: (i) submits to the non-exclusive jurisdiction of the courts of the Province of Ontario over any action or proceeding arising out of or relating to this Warrant; (ii) waives any objection that it might otherwise be entitled to assert to the jurisdiction of such courts; and (iii) agrees not to assert that such courts are not a convenient forum for the determination of any such action or proceeding.

g) <u>Restrictions</u>. The Holder acknowledges that the Warrant Shares acquired upon the exercise of this Warrant, if not registered, will have restrictions upon resale imposed by applicable securities laws.

h) <u>Non-waiver and Expenses</u>. No course of dealing or any delay or failure to exercise any right hereunder on the part of the Holder shall operate as a waiver of such right or otherwise prejudice the Holder's rights, powers or remedies. Without limiting any other provision of this Warrant or the Purchase Agreement, if the Company intentionally and knowingly fails to comply with any provision of this Warrant, which results in any material damages to the Holder, the Company shall pay to the Holder such amounts as shall be sufficient to cover any costs and expenses including, but not limited to, reasonable attorneys' fees, including those of appellate proceedings, incurred by the Holder in collecting any amounts due pursuant hereto or in otherwise enforcing any of its rights, powers or remedies hereunder.

i) <u>Notices</u>. Any notice, request or other document required or permitted to be given or delivered to the Holder by the Company shall be delivered in accordance with the notice provisions of the Purchase Agreement.

j) <u>Limitation of Liability</u>. No provision hereof, in the absence of any affirmative action by the Holder to exercise this Warrant to purchase Warrant Shares, and no enumeration herein of the rights or privileges of the Holder, shall give rise to any liability of the Holder for the purchase price of any Common Stock or as a stockholder of the Company, whether such liability is asserted by the Company or by creditors of the Company.

k) <u>Remedies</u>. The Holder, in addition to being entitled to exercise all rights granted by law, including recovery of damages, will be entitled to specific performance of its rights under this Warrant. The Company agrees that monetary damages would not be adequate compensation for any loss incurred by reason of a breach by it of the provisions of this Warrant and hereby agrees to waive and not to assert the defense in any action for specific performance that a remedy at law would be adequate.

l) <u>Successors and Assigns</u>. Subject to applicable securities laws, this Warrant and the rights and obligations evidenced hereby shall inure to the benefit of and be binding upon the successors and permitted assigns of the Company and the successors and permitted assigns of the Holder. The provisions of this Warrant are intended to be for the benefit of any Holder from time to time of this Warrant and shall be enforceable by the Holder or holder of Warrant Shares.

m) <u>Amendment</u>. This Warrant may be modified or amended or the provisions hereof waived with the written consent of the Company and the Holder.

n) <u>Severability</u>. Wherever possible, each provision of this Warrant shall be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Warrant shall be prohibited by or invalid under applicable law, such provision shall be ineffective to the extent of such prohibition or invalidity, without invalidating the remainder of such provisions or the remaining provisions of this Warrant.

o) <u>Headings</u>. The headings used in this Warrant are for the convenience of reference only and shall not, for any purpose, be deemed a part of this Warrant.

(Signature Page Follows)

IN WITNESS WHEREOF, the Company has caused this COMMON STOCK Purchase Warrant to be executed by its officer thereunto duly authorized as of the date first above indicated.

# BULLFROG GOLD CORP.

By: Name:			
Title:			
8			

THE SECURITIES FOR WHICH THIS SECURITY IS EXERCISABLE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT") AND MAY NOT BE EXERCISED EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OR PURSUANT TO AN AVAILABLE EXEMPTION FROM, OR IN A TRANSACTION NOT SUBJECT TO, THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT AS EVIDENCED BY A LEGAL OPINION OF COUNSEL TO SUCH EFFECT, THE SUBSTANCE OF WHICH SHALL BE REASONABLY ACCEPTABLE TO THE COMPANY.

#### NOTICE OF EXERCISE

#### TO: BULLFROG GOLD CORP.

(1) The undersigned hereby elects to purchase \_\_\_\_\_ Warrant Shares of the Company pursuant to the terms of the attached Warrant (only if exercised in full), and tenders herewith payment of the exercise price in full, together with all applicable transfer taxes, if any.

(2) Payment shall take the form of (check applicable box):

[] in the lawful money of Canada; or

[] in the lawful money of the United States of America.

(3) Please issue a certificate or certificates representing said Warrant Shares in the name of the undersigned or in such other name as is specified below:<sup>(1)</sup>

The Warrant Shares shall be (i) delivered to the following DWAC Account Number, (ii) deposited electronically with CDS through the book-based system administered by CDS using the "non-certificated inventory" issue process in accordance with the following instructions, or (iii) delivered by physical delivery of a certificate to the following address:

[SIGNATURE OF HOLDER]

Name of Investing Entity:

Signature of Authorized Signatory of Investing Entity:

Name of Authorized Signatory:

Title of Authorized Signatory:

Date:

<sup>(1)</sup> If the Warrant Shares are to be assigned as permitted in Section 4(a), please complete the Assignment Form.

THIS WARRANT AND ALL RIGHTS HEREUNDER ARE NON-ASSIGNABLE, NON-TRANSFERABLE AND NON-NEGOTIABLE AND MAY NOT BE EXERCISED BY OR FOR THE BENEFIT OF ANY PERSON OTHER THAN THE HOLDER OR AN AFFILIATE OF THE HOLDER, EXCEPT WITH THE PRIOR WRITTEN CONSENT OF THE COMPANY.

#### ASSIGNMENT FORM

(Do not use this form to exercise the warrant.)

The assignment of the Warrant Shares acquired on exercise of the Warrants is permitted in Section 4(a) as follows (check applicable box):

[ ] Affiliate of Holder; or[ ] Prior Written Consent of the Company.

To complete the assignment of the Warrant Shares acquired on exercise of the Warrants, execute this form and supply required information.

#### BULLFROG GOLD CORP.

FOR VA thereby are hereby assigned		l of or [] share	s of the foregoing V	Warrant and all rights evidenced
		whose address is		
		Dated:	,_	
	Holder's Signature:			
	Holder's Address:			
		10		

#### Exhibit 99.1



BULLFROG GOLD CORP.<sup>™</sup> AMERICA'S GOLD COMPANY

#### Bullfrog Gold Corp. Signs Definitive Agreements with Barrick and Augusta

Grand Junction, Colorado, October 13, 2020 – Bullfrog Gold Corp. (BFGC:OTCQB; BFG:CSE; 11B:FSE) ("**Bullfrog**", "**BFGC**" or the "**Company**") announces it has entered into definitive agreements ("**Definitive Agreements**") with certain Barrick Gold Corporation subsidiaries ("**Barrick**") and Augusta Investments Inc. ("**Augusta**") in connection with the previously announced transaction pursuant to which Bullfrog will acquire rights to 1,500 additional acres adjoining the Company's Bullfrog Gold Project from Barrick, and Augusta and certain individuals identified by Augusta will concurrently complete a C\$22 million investment in Bullfrog (the "**Transaction**").

Bullfrog's CEO & President, David Beling commented, "We are delighted to announce the signing of the Definitive Agreements and are encouraged by the positive market reaction to the announcement of the binding term sheet in respect of the Transaction. Bullfrog will be well-funded to expedite the exploration and development of the Bullfrog project, and we are excited to welcome our new cornerstone shareholders, Barrick and Augusta."

Upon completion of the Transaction, Barrick and Augusta will become significant shareholders in Bullfrog. Additionally, the Board of Directors and Management of the Company will be reconstituted upon closing such that Maryse Bélanger will be appointed President, CEO and Director of the Company along with the appointment of Donald Taylor, Daniel Earle, and a Barrick nominee to the Company's Board of Directors. David Beling will remain on the Board.

The Definitive Agreements formalize the terms in the binding term sheet dated September 7, 2020 and detailed in the Company's press release dated September 8, 2020. The Mineral Lease and Option to Purchase Agreement between a subsidiary of BFGC and Barrick dated March 23, 2015, as amended, has been terminated and the Barrick back-in right has been eliminated. A 2% NSR royalty granted to Barrick also decreases to a minimum 0.5% NSR royalty on certain Barrick lands already subject to royalties.

Closing of the Transaction is expected to follow during the fourth quarter of 2020, subject to customary conditions, including certain government approvals and the approval of the Canadian Securities Exchange.

#### About Bullfrog Gold Corp.

Bullfrog Gold Corp. is a Delaware corporation that currently controls the commanding land and resource positions in the Bullfrog Mine area where Barrick produced 2.3 million ounces of gold by mining and conventional milling beginning in 1989 and ending in 1999 when ore reserves were depleted. Measured and indicated 43-101 compliant resources were estimated in mid-2017 by Tetra Tech Inc. at 525,000 ounces of gold, averaging 1.02 gold g/t in base case plans of the MS and Bullfrog pits. Most of these resources are in the north extension to the Bullfrog pit. Inferred resources were estimated at 110,000 ounces of gold averaging 1.2 g/t, of which most were in the Mystery Hill area adjacent to the Bullfrog pit and where 17 additional holes were recently drilled.

It is noted that the 3-year trailing average gold price is more than \$200 higher than the \$1,200 average price used in 2017. The resource estimates were also based on a heap leach gold recovery of 72% at a minus 3/8-inch feed size compared to the average 85% leach test recoveries recently achieved on four bulk samples sized at minus 1/16-inch. Much additional technical and corporate information may be sourced at <u>www.bullfroggold.com</u>.

#### About the Augusta Group

The Augusta Group is a mining sector focused management group based in Canada and the United States led by Richard Warke. Augusta has an industry leading track record of value creation totaling over C\$ 4.5 billion in exit transactions since 2011 and has strategic partnerships with the leading entrepreneurs in the mining sector. Additional information on the Augusta Group can be found at <u>www.augustacorp.com</u>.

#### Cautionary Note Regarding Forward Looking Statements

This press release contains certain "Forward-Looking Statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the United States Securities Exchange Act of 1934, as amended. All statements, other than statements of historical fact, included herein with respect to the objectives, plans and strategies of the Company and those preceded by or that include the words "believes," "expects," "given," "targets," "intends," "anticipates," "plans," "projects," "forecasts" or similar expressions, are forward-looking statements that involve various risks and uncertainties. Forward-looking information in this press release includes but is not limited to statements regarding the Transaction and changes to the management and board of the Company.

Such forward-looking information and statements are based on numerous assumptions, including among others, the Company's ability to successfully maintain its listings, the stability of industry and market costs and trends and the Company's ability to obtain all regulatory approvals required for its planned objectives. Furthermore, by their very nature, forward-looking information involves a variety of known and unknown risks, uncertainties and other factors which may cause the actual plans, intentions, events, results, performance or achievements of the Company to be materially different from those expressed or implied by such forward-looking information. Such risks, uncertainties and other factors include, without limitation, those related to: (a) closing of the Transaction (b) adverse regulatory or legislative changes (c) market conditions, volatility and global economic conditions (d) industry-wide risks (e) the Company's inability to maintain or improve its competitive position and (f) the ability to obtain financing needed to fund the continued development of the Company's business.

#### **Qualified Person**

David Beling, P.E. has 56 years of project and corporate experience in the mining industry and is a qualified person as defined by Canadian National Instrument 43-101 – Standards of Disclosure or Mineral Projects. Mr. Beling has prepared, supervised the preparation of, or approved the technical information that forms the basis of the Company's disclosures, but is not independent of Bullfrog Gold Corp, as he is the CEO & President and holds common shares and incentive stock options of the Company.

For further information, please contact David Beling, CEO & President, at (970) 628-1670.